



Eboni Anderson, DHEd, MA, MEd, MSW Allison Knight, PhD Margaret Rea, PhD

CHANGEMEDED



Foreword

Since its inception the American Medical Association's Accelerating Change in Medical Education Consortium has made innovation that improves medical education—and, ultimately, patient outcomes—a driving objective. Well-being, with a particular focus on medical students, has long been important to this work. With the onset of the COVID-19 pandemic, however, it became clear educator well-being needed to be part of the fundamental calculus.

Much as the ongoing pandemic highlighted cracks in many American societal structures and institutions, it exposed cracks in the medical education system. In real time, and in real ways, we have learned educator well-being efforts are too often under-resourced. When the pandemic emerged early in 2020, many of those inadequacies were revealed as educators had to pivot to teaching virtually, which was novel for many. Those who were also clinicians were pulled away from educational duties to care for the increasing number of patients with COVID-19. Educators without clinical duties often had to cover some of the educational and academic responsibilities of those whose clinical workload had expanded. Everything during the pandemic required a greater cognitive load, and there was more to come.

The murder of George Floyd in May 2020, and the deaths of other Black people as a result of police action, increased the focus on structural racism. This, in turn, contributed to additional educator stress as those from historically excluded groups, while still processing their own experiences to these horrific events, were asked, implicitly and explicitly, to participate in addressing these difficult challenges. Many faculty from dominant cultures also experienced the increased need to grapple with structural racism personally and in their official roles, but often the lacked training to do so effectively.

The stresses of the pandemic and increased attention to social injustice formed a perfect storm, one that placed medical education at increased risk of losing core educators. The consortium chose to lean into the moment, seeking to gather and understand the spectrum of issues and points of view needed to shape a well-informed response.

This book, *Educator Well-Being in Academic Medicine*, which is a partial extension of the consortium's early discussions and activities, presents the work of experts from across the country who have studied, planned and implemented educator well-being programs in undergraduate and graduate medical education. The authors and editors provide concrete, systems-based solutions to better support the educational mission, and by extension, educator well-being.

We are pleased to offer this book and look forward to learning about your experiences in improving educator well-being at your institution.

Sanjay V. Desai, MD Chief academic officer American Medical Association

Preface

The American Medical Association is deeply invested in supporting the well-being of physicians and endorses the objectives of the Quadruple Aim: improving the health of populations, enhancing the patient experience of care, reducing the per capita cost of health care and improving the work life of physicians and other health care professionals.¹

Many well-being initiatives, whether targeting practicing physicians or those in training, focus on the attributes and resilience of the individual. The AMA, recognizing that negative work environments cannot be overcome by individual resilience alone, continues to press vigorously for organizational and systems-level changes, including those that will have a direct and positive effects on medical education. Removing obstacles to care, such as burdensome documentation or inefficient workflows, allows physicians to concentrate their efforts on the needs of patients and communities. Programs such as the AMA's <u>STEPSforward</u> offer pragmatic strategies to refine practices and improve organizational culture to support well-being.

Health care systems and medical education institutions and programs have a duty to attend to workforce well-being. It is both the right thing to do and the fiscally responsible approach to take. Burnout can be very costly to an organization. For example, physicians who experience burnout are twice as likely to leave their organization,^{2,3} with the estimated costs to replace a single physician ranging from \$500,000 to \$1.3 million. Across the U.S. health care system, physician burnout is estimated to cost approximately \$5 billion annually.^{4,5}

This ties back to the well-being of educators in medicine—a topic that does not get enough attention. Just as the well-being of physicians is foundational to optimal clinical care, the well-being of educators is foundational to optimal learning environments. The AMA Accelerating Change in Medical Education Consortium has modeled its efforts to support well-being in the medical education setting after many of the organizational strategies the AMA has applied to physicians in the health care setting. Paralleling the framework and principles established for care providers, members of the consortium outlined the unique needs of educators and generated a preliminary guide in February 2022 (reprinted in the Appendix) for protecting the educational mission. That was an important first step. This book takes those efforts and insights to the next level by providing greater actionable detail to guide organizations dedicated to supporting the educators who cultivate our future health care workforce.

Christine Sinsky, MD
Vice president
Professional satisfaction
American Medical Association

Kimberly D. Lomis, MD
Vice president
Undergraduate medical education innovations
American Medical Association

References

- 1. Bodenheimer T, Sinsky C. From triple to quadruple aim: care of the patient requires care of the provider. *Ann Fam Med.* 2014;12(6):573–576. doi: 10.1370/afm.1713
- 2. Windover AK, Martinez K, Mercer MB, Neuendorf K, Boissy A, Rothberg MB. Correlates and outcomes of physician burnout within a large academic medical center. *JAMA Intern Med.* 2018;178(6):856–858. doi:10.1001/jamainternmed.2018.0019
- 3. Hamidi MS, Bohman B, Sandborg C. et al. Estimating institutional physician turnover attributable to self-reported burnout and associated financial burden: a case study. *BMC Health Serv Res.* 18, 851 (2018). https://doi.org/10.1186/s12913-018-3663-z.
- 4. Han S, Shanafelt TD, Sinsky CA, et al. Estimating the attributable cost of physician burnout in the United States. *Ann Intern Med.* 2019 Jun 4;170(11):784-790. doi: 10.7326/M18-1422. Epub 2019 May 28. PMID: 31132791.
- 5. Sinsky CA, Shanafelt TD, Dyrbye LN, Sabety AH, Carlasare LE, West CP. Health care expenditures attributable to primary care physician overall and burnout-related turnover: A cross-sectional analysis. *Mayo Clin Proc.* 2022 Apr;97(4):693-702. doi: 10.1016/j.mayocp.2021.09.013. Epub 2022 Feb 26. PMID: 35227508.

Letter from the editors

Dear Readers,

We are pleased to present *Educator Well-Being in Academic Medicine*. The idea for this book emerged from a thematic discussion series hosted early in 2021 by the American Medical Association Accelerating Change in Medical Education Consortium. The series, entitled "Supporting Medical Educators During Disruption and Beyond," saw more than 60 different medical institutions participate in active discussions that identified the need to better address the value of medical education and the well-being of educators. The COVID-19 pandemic, which remains an ongoing concern, highlighted the fact that educator well-being has not received the attention it deserves and has not been formally addressed.

We are deeply committed to the educational mission of academic medicine and to playing an active role in addressing the issue of educator well-being. It is clear that if we are to tackle the systemic and cultural issues that impact educator well-being, our efforts must be inclusive of both clinical and nonclinical educators from allopathic and osteopathic institutions and across the undergraduate and graduate continuum. Only then can we ensure that individual educators remain well and flourish as they make invaluable contributions to educating our future scientists and physicians.

With this in mind—and to honor our personal commitment to support a healthy, diverse and inclusive workforce where education is valued—we reached out to colleagues from across the continuum to provide expert lenses and guidance regarding the issues in academic medicine that need attention and action.

Our goal in creating this book is to provide administrators and leaders in academic medicine with a unique, solutions-focused guide for taking the needed steps at their institutions to ensure educators feel valued and the educational mission is uplifted and sustained. Grounded in evidence-based research and through personal narratives and lived experiences, the book chapters offer recommendations and guidance that are not only meant to resolve issues of well-being at the institutional level, but also to help heal U.S. medical education at the systems level, which we believe can only be achieved by supporting inclusive practices and policies.

We encourage readers to use this book as a guide to enrich educator well-being, both individually and collectively, and to further the discourse on systemic well-being in U.S. medical education. As co-editors, we take this opportunity to express our gratitude to the AMA, the authors and reviewers, and the many other individuals for their assistance in bringing this effort to life. We are proud to offer this book and feel strongly it

has the ability to be a seminal work that contributes to the body of literature focused on making academic medical institutions systemically well.

Eboni Anderson, DHEd, MA, MEd, MSW
Director, community oriented primary care
Assistant professor, Department of Public Health
A.T. Still University-School of Osteopathic Medicine in Arizona

Allison Knight, PhD Assistant vice dean for student affairs and director of student wellness Assistant professor, Department of Psychiatry and Behavioral Sciences Eastern Virginia Medical School

Margaret Rea, PhD Clinical professor, emergency medicine Director student and resident wellness UC Davis School of Medicine

Contributors

Saadia Akhtar, MD

Associate dean, trainee well-being and resilience in graduate medical education Icahn School of Medicine at Mount Sinai Chapter 1

Catherine L. Coe, MD
Assistant professor
Department of Family Medicine
University of North Carolina
Chapter 5

Mario Fahed, MD
Assistant professor, psychiatry
University of Connecticut School of Medicine
Chapter 4

Tonya Fancher, MD, MPH

Chapter 4

Associate dean, workforce innovation and education quality improvement
Professor, general internal medicine
University of California Davis, School of Medicine

Alicia Gonzalez-Flores, MD Associate professor Department of General Internal Medicine University of California Davis School of Medicine. Chapter 4

David Henderson, MD

Associate dean, multicultural and community affairs Professor and chair, Department of Family Medicine University of Connecticut School of Medicine Introduction and Chapter 4

Donna Jackson, EdD
Assistant dean for admissions
Director, Office of Student Outreach Programs
Virginia Commonwealth University School of
Medicine
Chapter 4

Julie A. Kerry, PhD
Professor, microbiology and molecular cell biology
Eastern Virginia Medical School
Chapter 5

Sara M. Lamb, MD
Associate dean for education curriculum
Clinical associate professor, internal medicine
University of Utah School of Medicine
Chapter 1

Elizabeth C. Lawrence, MD
Professor, Department of Internal Medicine
Assistant dean, professional well-being
Chief wellness officer
University of New Mexico School of Medicine
Chapter 6

Khanh-Van Le-Bucklin, MD, MEd Vice dean, medical education UCI School of Medicine Chapter 3

Rosa Lee, MD
Senior associate dean, curricular affairs
Columbia University Vagelos College of Physicians
and Surgeons
Chapter 4

Andrea N. Leep Hunderfund
Associate professor, neurology
Medical director, Office of Applied Scholarship and
Education Science
Mayo Clinic College of Medicine and Science
Chapter 6

Winston Li, MD
Clinical assistant professor, Department of
Psychiatry
University of North Carolina School of Medicine
Chapter 5

Yasmin Magdaleno, MD

Associate professor, pediatrics

Director, faculty initiatives, Office of Professional

Well-being

University of New Mexico School of Medicine

Chapter 6

Lisa Meeks, PhD, MA

Assistant professor, Department of Learning Health

Sciences and Family Medicine

University of Michigan Medical School

Chapter 4

Kathryn B. Moore, PhD

Professor, neurobiology & anatomy

University of Utah School of Medicine

Chapter 1

Senthil Kumar Rajasekaran, MD

Chief academic officer

Interim dean

Khalifa University College of Medicine and Health

Sciences

Chapter 3

Jonathan Ripp, MD, MPH

Dean, well-being and resilience

Icahn School of Medicine at Mount Sinai

Chief wellness officer

Mount Sinai Health System

Chapter 1

Annie Robinson, MS, CHWC

Senior manager, professional fulfillment and well-

being, GME

NYU Grossman School of Medicine

NYU Langone Health

Chapter 2

Stuart Slavin, MD, MEd

Senior scholar, well-being

Accreditation Council for Graduate Medical

Education.

Chapter 1

Richard Van Eck, PhD

Associate dean for teaching and learning, education

& faculty affairs

Founding Monson Endowed Chair for Medical

Education

University of North Dakota

Introduction

Abigail Ford Winkel, MD, MHPE

Professor

Vice chair, education

Department of Obstetrics & Gynecology

Assistant director, education scholarship

Institute for Innovations in Medical Education

Assistant director, master's in health professions

education

New York University Grossman School of Medicine

Chapter 2

Julie Youm, PhD

Associate dean, education compliance and quality

UCI School of Medicine

Chapter 3

Acknowledgements

The editors and authors of this book would like to thank the American Medical Association for their leadership and support of this project. We also would like to acknowledge the contributions of Kathryn Pajak, project manager in the AMA's Medical Education department who kept this book on schedule, and Victoria Stagg Elliott, technical writer in the AMA's Medical Education department, who corrected our grammar and caught our typos. We would like to express our gratitude to Amanda Moutrage, also of the AMA, for her administrative project support. For his role in the development of the content of this book and his leadership throughout, we are grateful to Kevin Heckman, the AMA's director of product development.

We would like to also thank the contributing editors for equity, diversity and belonging—David Henderson, MD; Rosa Lee, MD; and Alicia Gonzalez-Flores, MD,—for applying an equity lens to all of our work in this area. We would also like to thank the contributing editors—Kimberly Lomis, MD, the AMA's vice president for undergraduate medical education innovations, and Joaquin Baca, MPH, PhD, the AMA's director of equity, diversity and belonging—for their thorough reviews of every chapter. Their feedback was invaluable.

Table of contents

ntroduction
ichard Van Eck, PhD, and David Henderson, MD
hapter 1: Value of education to the institutional mission
hapter 2: Institutional structure
hapter 3: Responding to/functioning in the larger picture of academic medicine
hapter 4: Diversity, equity and inclusion
hapter 5: Intergenerational cultures within medical education
hapter 6: Measuring educator well-being
ppendix: Protecting the education mission during sustained disruption96

Introduction

The well-being of educators is of paramount importance if academic medicine is to fulfill its role in producing a diverse and flourishing health care workforce. This publication articulates many of the issues—whether historical, structural, cultural or financial—that impact educator well-being and, in turn, influence educator efforts to meet the needs of students, institutions and patients.

Well-being can (and should) be characterized in many different ways. It is often viewed as a state of being or an experience. Far less often is it viewed as an *action*, which in fact may be the most relevant way to represent the role and impact of well-being in academic medical education. Well-being comprises many different facets, some often used as a preface or framework: emotional, social, cultural, psychological, financial, professional and personal—all have very different meanings and implications. It is critical to understand these and other frameworks and lenses if we are to design solutions that promote well-being. While these facets share many things in common, they also have unique contributing factors and often require divergent solutions.

Suppose well-being is indeed an action as well as a state of being. If this is the case, it cannot be achieved without intentional, sustained effort and support on the part of health systems, undergraduate and graduate medical education institutions, and individual educators. As an action, well-being shapes educators' capacity to engage emotionally and intellectually. It contributes to their motivation, sense of satisfaction and, ultimately, effectiveness and ability to achieve success. So, what should the well-being efforts of the health system, the medical education continuum and individual educators look like? Does the medical education system know enough to define, let alone address, well-being and the challenges educators face?

Additionally, the body of educators is broad and diverse. It is composed of individuals from a wide variety of communities, and each educator may embody multiple identities. For example, an educator who identifies as transgender and a member of a historically excluded racial and/or ethnic community will view well-being in variable and complex ways. If well-being is viewed as the outcome of interactions between individuals and the prevailing environment mediated by an intersectional web of identities, well-being is not one, but many things. Designing well-being solutions necessitates an awareness of how well-being is negotiated by individuals, identities and environmental factors; one size will not fit all.

Medical education as a profession presents its own diverse challenges when it comes to well-being. Educators' work is often invisible or poorly understood by students, administrators and health care systems, creating unique stressors and threats to well-being. The dozens of hours of design that can precede a high-quality instructional event take place in offices and conference rooms long before anyone experiences the event. The rigor that underlies meaningful assessment design and evaluation of sessions, courses, phases and programs is often invisible to those who receive only the reports that distill hundreds of hours of effort and skill.

All of this conspires to create a culture where well-being is experienced in hundreds of different ways yet treated as if a single approach can promote well-being for all. The stressors faced by faculty of color, LGBTQ+ faculty, and women faculty are the result of a combination of factors such as structural racism, prejudice and misogyny that are not just *additive*, but *multiplicative* of the stressors experienced by all educators. Worse, it is often

faculty from groups that have been historically excluded from medicine who are routinely tasked with chairing well-being, diversity, equity and inclusion efforts. This frequently imposed "diversity tax" compounds threats to well-being. The solutions to these and other problems must therefore be enlightened and diverse.

So, given these challenges how can an organization hope to embrace this complexity in a meaningful way? This publication is part of the answer. One of the key messages we hope readers will take from this book is that it is both impractical and counterproductive to assume well-being can be universally defined or that solutions can be transferred from one context to another without modification.

The content of this publication offers several views on well-being, from theoretical and conceptual perspectives to experiential narratives. It also adopts a systems view of the topic and proposes steps medical education organizations can take to define and begin to address well-being's cultivation. Chapter 1 provides the framework for the book by examining the value of education to the mission of academic medicine and the impact it may have on educator well-being. The authors discuss the challenges facing both basic science and clinical educators.

Chapter 2 considers the impact on well-being of the conflicting structural priorities present in academic health centers. For example, new faculty may be expected to model well-being and create a nurturing environment for students while experiencing personal burnout from the stressors of learning to be an effective teacher and establishing a research and funding portfolio. Chapter 3 focuses on the additional burdens placed on educators to simultaneously fulfill multiple roles across clinical care, research and administration in response to the bigger picture of academic medicine.

Chapter 4 shares a series of narratives that depict the challenges faced by educators from historically excluded groups and the impact those challenges have had on their well-being. These unique threats to well-being intersect and interact with those of being an educator. Chapter 5 discusses the challenges present in the intergenerational culture of medical education environments with specific emphasis on issues such as diminished autonomy, lack of faculty development and isolation.

Lastly, Chapter 6 explores the challenges inherent in the measurement of well-being. It provides an overview of the strengths and weaknesses of several conceptual frameworks, as well as the challenges inherent in data analysis and application of results. These frameworks will be more effective for some organizations than others and while not comprehensive, they highlight the need to operationally define well-being in a way that embraces the unique cultural climate of an institution. Each chapter offers thoughtful solutions to address the issues that are explored.

We recognize there are many audiences for this publication and that each will view well-being from a different perspective. This book attempts to provide as many perspectives as possible, including undergraduate medical education, graduate medical education, health care systems, administration, faculty and staff. It addresses issues faced by allopathic and osteopathic educators as well as those teaching basic science and clinical science.

It is not possible to fully address every context for medical education and well-being. However, we hope the approach the authors have used will provide readers with the tools and insights to conduct their own analyses and definitions of well-being at their institutions and to use the results to develop customized solutions well aligned with the culture and climate of their organizations.

Richard Van Eck, PhD
Associate dean for teaching and learning, education & faculty affairs
Founding Monson Endowed Chair for Medical Education
University of North Dakota

David Henderson, MD Associate dean, multicultural and community affairs Professor and chair, Department of Family Medicine University of Connecticut School of Medicine

Chapter 1: Value of education to the institutional mission

Sara M. Lamb, MD; Kathryn B. Moore, PhD; Saadia Akhtar, MD; Stuart Slavin, MD, MEd; and Jonathan Ripp, MD, MPH

Chapter Summary

We can imagine a future where our most talented and inspiring medical educators have decided it is no longer worth it to teach. A future where, because the well-being of educators has not been adequately addressed, the health system risks losing the individuals who are unequivocally needed to educate the future workforce. We believe explicit changes to how institutions value educators, and the work done to advance the education mission, is paramount to the future sustainability and success of the national health care system. In this chapter we highlight the misalignment of current measures of educational success with the culture that we aspire to work within. We outline major challenges facing foundational science and clinical educators using an organizational framework proposed by Bolman and Deal. We conclude by delineating specific changes we believe can help demonstrate the value of educators to institutions. Implementing these adaptations can aid in preserving educator well-being and help to positively transform the culture of medicine.

This introductory chapter helps set the stage for the book, highlighting numerous key issues that are central to the conversation about medical educator well-being. Through the lens of how educators and education are valued at institutions, we touch on components that are further explored in subsequent chapters, including: institutional structure (Chapter 2), functions that are characteristic of academic medicine (Chapter 3), concerns around diversity, equity and inclusion (Chapter 4), challenges relating to the intergenerational culture of medical education (Chapter 5), and the importance of measuring educator well-being (Chapter 6).

Introduction

The issue of physician well-being is increasingly in the public eye. That said, not all who educate medical students, residents and practicing physicians are physicians themselves, although much can be gleaned about educator well-being from data measuring the well-being of physicians. Equally as important, and intertwined with educator well-being, is the issue of how medical education and medical educators are valued.

The direct cost of physician burnout is estimated at \$4.6 billion due to physician turnover and reduced clinical hours.² Competition for extramural funding to support the research mission of academic institutions is only increasing, and teaching continues to be undervalued at many institutions.^{3,4} Pervasive practices of unfunded expectations for educators to teach students and graduate trainees, on top of full clinical and research workloads, are contributing to the challenges of educator well-being.⁵ Further, the indirect costs to institutions may be an even greater burden on the health system. How can our medical scientists afford the time to invest in uncovering the next scientific discoveries, yet still devote time and effort to developing the educational skills needed to teach and mentor medical and graduate students? With physician burnout rates approaching 50% or higher,⁶ the negative implications of burned-out educators as role models for medical students, graduate students, residents and fellows may have a far greater impact on the profession's future. We can imagine a future where the most talented and inspiring medical educators suddenly decide it is no longer worth it to teach because of the pressures to generate clinical revenue, balance time commitments for other administrative demands, acquire grants and disseminate research outcomes. The risk of not addressing the tension between

how faculty contributions to education are valued versus clinical productivity and research is high. As such, we propose a broad approach to thinking about the value and the well-being of educators as necessary components for the future sustainability and success of the medical education enterprise.

This chapter describes commonly used measures of medical educator value that are currently in place. We propose more meaningful metrics that will help achieve the desired culture in medical education. We further and intentionally focus on how the value of medical education, and valuing of educators, impacts educator well-being in ways that are relevant to health system leaders (e.g., CEOs, CFOs), academic undergraduate medical education (UME), graduate medical education (GME) leaders (e.g., deans, department chairs, program directors), and educators. The ability for all key stakeholders to understand the downstream consequences of insufficiently valuing medical educators, or the education mission, is paramount to ensuring academic medicine's future success.

Defining and Measuring Value

In his seminal paper, "What is Value in Health Care?" Michael Porter defines value as the desired patient outcomes divided by the cost to achieve those outcomes. Similarly, in medical education, value can be defined as the sum of the quality of the education plus the experience of learners and educators, divided by the cost to deliver medical education (Fig. 1.1). The costs, however, are not just that of the buildings, supplies, and staff and faculty time. Importantly, they also include the cost to educators and their well-being as they invest their efforts to advance the education mission, which is not consistently recognized, rewarded or supported across institutions.



Fig. 1.1: Value Equation for Medical Education

W. Edwards Deming is commonly quoted: "Every system is perfectly designed to get the results it gets," which highlights the importance of focusing on and defining the results that are the most crucial to achieving the educational and medical culture we hope to work in. Academic medical centers and health systems highly prioritize clinical productivity, quality rankings, extramural grant funding and growing research space. National rankings of medical schools, in part, hinge on higher applicant-to-acceptance ratios, applicant grade point averages and medical college admissions test (MCAT®) scores, 9,10 which run counter to institutional efforts to improve diversity, equity, belonging and inclusion. United States Medical Licensing Exam® (USMLE®) and Comprehensive Osteopathic Medical Licensing Examination of the United States (COMLEX-USA) performance is also commonly used to compare the strength of students, and in aggregate, the programs, yet work against those favoring less emphasis on standardized test taking skills and efforts to achieve a diverse workforce. Altogether, the aforementioned measures do not necessarily contribute to high quality medical education programs. In this chapter, we outline the importance of measuring the scholarly approach to medical education

that fuels the continuous quality improvement of the enterprise. Quantifying the financial support of educators, and making it transparent and equitable, not only signals the importance of educator expertise to the education mission of academic institutions, but also signals the value of the education mission to the broader community. Resourcing and creating high regard for educator communities is vitally important. It encourages faculty engagement in educator training initiatives. Further, it creates an inclusive culture where faculty are more apt to feel safe in exploring educational innovations, which is critical to improving education practices. Perhaps most importantly, we emphasize the need to cultivate environments that value educator job satisfaction (balanced with learner satisfaction) and ensure faculty advancement and recognition for their impact as scholarly teachers, curriculum developers, assessors, leaders, mentors and coaches.¹¹⁻¹⁶

Current State, Key Challenges

Funding issues in medical education—Costs, revenue and value

The current financial state of medical education poses many challenges for medical schools, but academic centers and medical schools can work in partnership to advance medical care. Many academic medical centers face significant financial pressures that have been exacerbated by the COVID-19 pandemic, and the pressure on educator productivity has increased.¹⁷ In UME, where educators have been historically underfunded for teaching and course leadership positions, increasing or even maintaining appropriate levels of support may be challenging. In GME, Accreditation Council for Graduate Medical Education (ACGME) requirements mandate protected time for residency program directors. As a result, GME directors perhaps face fewer challenges in being funded and have more protected time to carry out their education and administrative duties than their UME colleagues. However, GME allows residents and fellows who earn salaries at a lower price point to provide expensive patient care and revenue for academic centers. Administration in these centers may view physiciansin-training as an expense with little revenue potential and may be hesitant to make significant investments that support resident education beyond what is required to meet accreditation standards. Zinoviev et al. 18 found that GME funding may not increase the financial stability of a hospital. This research also found that hospitals receiving more GME funding had lower financial standing than those that did not receive such funding or received less of it. Further, the loss of clinical revenue as a result of educational responsibilities will likely be reflected in further loss of support for the education mission beyond GME, including teaching and research support. These tensions between the health care system and medical education need to be considered as a part of any potential solutions.

Inadequate integration of education into the health care system and lack of educators' voices in decision-making

Physicians and scientists may perhaps view the health care system as a black box¹⁹ because they have limited insights into the drivers that influence how health systems operate and how system leaders make decisions. Though medical education programs are central to academic medical centers, medical educators tend not to be well-integrated into the teams that influence decision-making at the systems level, which is another example of how the voice of the educator is not heard or valued. Consequently, educators often hold little sway in the decisions that can impact education, resulting in potentially profound negative impacts on education goals, clinical and non-clinical educators and, ultimately, learners.

Insufficient leadership training for clinical and foundational science educators

Patient outcomes are improved with effective physician leadership.²⁰ Physicians-in-training need to be prepared for informal or formal leadership roles once their training is complete. Sadowski et al.²¹ conducted a systematic review on existing literature of GME curricula related to leadership development. They concluded that GME leadership curricula are heterogeneous and limited in its effectiveness. Similarly, many foundational science educators are thrust into leadership roles in medical education without significant leadership training.

Lack of time for scholarship

Many educators struggle to find adequate time in their busy schedules to participate in scholarly activities. Scholarly activities may include, but are not limited to, developing grant proposals, submitting peer-reviewed journal articles (or other academic publications), and/or presenting at professional conferences.²² Zibrowski et al.⁴ explored the insufficient time barrier in more detail and concluded that many educators are only able to devote negligible amounts of time to academic scholarship. The time-related factors involved three themes: fragmentation (i.e., Opportunities to work on education projects are sporadic.); prioritization (i.e., Work responsibilities compete for time, and there is difficulty in securing financially compensated time.); and motivation (i.e., This is directly impacted by the limited degree of recognition, support and funding for education work by department and institutional leadership.). Additionally, the lack of institutional funding is a major barrier to scholarly engagement since some educators need additional support for such things as data analysis, an area of expertise that many educators lack. Education grants are few and difficult to find, and the payment is significantly lower than for National Institutes of Health (NIH) grants. The education grants that exist are highly competitive, and there is often a lack of formal grant-writing training.²³ Further, many institutions deprioritize education grant submissions since there are typically lower indirect costs that benefit the institution.²⁴

Insufficient training and funding to carry out education research and scholarship

Many educators with leadership roles have had little to no formal training in academic research, which may hinder their productivity and academic advancement. Government and private financial support for education development remains inadequate and is dwarfed by grant support for biomedical research. If the U.S. health system is to further advance medical education, increased grant support for education research is of vital importance.

Lack of diversity, equity, inclusion and belonging in medicine

By 2060, more than 50% of Americans will identify as a race other than white; however, in 2018, fewer than 12% of physicians were from groups that have been historically excluded from medicine.²⁵ Within health care leadership, the numbers are worse with 98% of health care organization leaders being white. According to the Association of American Medical Colleges (AAMC),²⁶ as of 2018 fewer than 4% of full-time faculty in medical schools identified as Black or African American, and fewer than 6% identified as Hispanic, Latinx or of Spanish Origin. Further, these faculty face different barriers and challenges to success and well-being.²⁷ As discussed further in Chapter 4, these low numbers and unique challenges can have a negative impact on the presence of a diversified health care workforce, which includes medical students, physicians-in-training and faculty. The well-being of educators from groups that have been historically excluded and marginalized represents an area that needs specific solutions.

Opportunities and Solutions

Shanafelt's work has highlighted the importance of enhancing meaning in work as a mechanism to increase physician satisfaction and reduce burnout.²⁸ Other researchers have identified the best predictors of educators feeling energized and vital at work due to feeling a sense of belonging, having strong relationships with colleagues and having institutional support for their career development. ²⁹⁻³¹ In this section we propose ideas organized and informed by the four frames outlined by Bolman and Deal¹ (Fig. 1.2): structural, human resource, symbolic and political. We emphasize the values of transparency, equity, support, community, training and recognition as fundamental to ensuring educators feel valued, sustained and, ultimately, well while working in medical education. We deliberately present a host of ideas and suggestions hoping to appeal to a broad spectrum of institutions and leaders. Some may be perceived as quite lofty and others ostensibly simple to implement. Others may be feasible with some modifications, and several are discussed in more detail in subsequent chapters. Altogether, we feel these represent a comprehensive set of evidence-based actions that can be taken to advance educator well-being.



Fig. 1.2: Four frames of organizational excellence informed by Bolman and Deal¹ structural, human resource, symbolic and political.

Structural

Funding medical education

Understanding the flow of funds in academic medical centers is anything but simple. Arguably, it is often only well understood by those at the higher levels of the organization. This may not be well-communicated or made transparent to educators and is certainly not consistent across institutions. For educators to feel valued, and for the education mission to be perceived as valuable (rather than a money-losing endeavor), institutions might consider the following to improve how educators feel about the value they bring to the institution:

Protect time and provide transparency

For faculty who are involved in educational activities, their time should be supported (both in teaching and in training), and the level of support, including how it is derived, needs to be transparent. Much like the ACGME delineates required levels of protected time for residency program directors,³² medical schools can identify a broad array of core roles and expected effort for educators to ensure the optimal design and delivery of the UME program. While the Liaison Committee on Medical Education (LCME) does not currently dictate requirements for support of teaching faculty, this could be an area for future advocacy, whereby accreditation requirements could expect transparency and clear financial support of educators to have protected time to teach and direct major components of UME programs.

Track and communicate

For those who receive significant support, education leaders should generate regular reports of education productivity and communicate the quality and value of educators' efforts to division leads, department chairs and supervisors. The actual time commitment and performance outcomes influenced by educators should be made clear to highlight the value of the educators and their contributions to achieving the goals of the education mission.³³

• Prioritize equity for educators

The support for faculty time required to engage in the education mission should, at minimum, be consistent across faculty devoting significant effort to education. Consideration should be given to how to ensure equity in remuneration for contributions to other mission areas (i.e., research, administration, clinical). Faculty often feel disincentivized to teach (i.e., are expected to spend their evenings and weekends preparing for education activities to preserve clinical time and revenue generation).⁴ Gender-based pay inequities remain an important challenge for female educators. Clinician-educators may opt for greater schedule flexibility offered by having a mixture of clinical and education roles in their professional lives. Reducing salary disparities between male and female clinician-educators will help institutions appoint and retain quality female clinical faculty who choose careers in academic medicine.

Similarly, institutions must also ensure an equitable pathway to appointing and retaining non-clinician educators. Leaders should consider compensation plans that reward non-clinician educators just as those with active labs and grant funding. At the same time, leaders should recognize the incredible challenge of maintaining active labs and writing grants in the face of increased work responsibilities and expectations. This should also include equitable support for non-clinician educators dedicated to full-time teaching. After all, education is the key to the success of translational science.

Provide education about funding

We recommend that institutions consider how they can openly share how expenses are allocated and revenues are generated by each mission area to create a culture of transparency, accountability and understanding.³⁴ Most educators have limited knowledge of how funds flow within an organization. Educating everyone, even at a rudimentary level, about how education is supported can go a long way toward engendering trust. Free, online resources such as those published by the AAMC³⁵ can be utilized to accomplish this goal.

Develop education incentives

In the same way that incentives for clinical and research productivity are baked into the culture of an institution and remuneration practices, there should be comparable education incentives. Models for

recognizing, incentivizing and tracking outcomes that meet institutional education goals are best when they are consistent across departments, developed in partnership with educators, communicated to educators and incentivized to advance desirable behaviors (i.e., as opposed to merely accounting for time spent teaching at the cost of education quality).³⁶

Promotion process reform

"Not everything that can be counted counts and not everything that counts can be counted." (attributed to Albert Einstein).

While manuscripts and grants are easily countable, the positive impact teaching faculty have on learners can be more difficult to quantify. Satisfaction with promotion progress is lower for clinical faculty who find themselves less likely to be at higher academic ranks and contributes to the increased likelihood these faculty will leave academic medicine.³⁷⁻³⁹ Academic medical centers struggle to achieve parity in advancement and promotions between educators and discovery-oriented researchers in part because of narrow definitions of scholarship, lack of clear criteria for measuring excellence and barriers to making education contributions available for peer review.⁴⁰ As of 2019, fewer than 10% of U.S. medical schools have thoroughly embraced published recommendations for documenting and evaluating educational excellence.⁴¹ To address the gap in how education is valued in the promotion and tenure process we recommend that institutions implement the following:

Develop clear criteria

Institutions can partner with education leaders and faculty educators to develop clear criteria for promotion based on excellence in education. The specific measures of educational excellence should go beyond papers published and consider the value of taking a scholarly approach to education design, education delivery and the impact on learning outcomes. ^{11,12} One approach may be to create a standards-based, peer-reviewed teaching award committee to evaluate the impact of education faculty in order to inform promotion and tenure committee decisions. ¹⁴ Another is the creation of academies of medical educators. These academies are now highly prevalent in academic medicine and allow those gaining membership a means of demonstrating formal institutional recognition for excellence in education. ¹³

• Use consistent guidelines for evaluation

The AAMC proposes five core educator domains (teaching, curriculum development, advising and mentoring, education leadership, and learner assessment) and we suggest that each be recognized in the review of faculty for promotion.¹⁵ Furthermore, consistent guidelines for evaluating educator excellence should be utilized by retention, promotion and tenure decision-making committees, such as those proposed by Gusic et al.¹⁶

Utilize appropriate tools

Educator portfolios (EPs) or annotated curriculum vitae are effective tools for educators to communicate both qualitatively and quantitatively their accomplishments within education. However, the integration of EPs, or the ability to provide sufficient annotation, may require institutions to develop, adapt or acquire new platforms. Further, educators must be prepared to effectively utilize these tools, and reviewers must be trained to evaluate the content using appropriate criteria. 40,42 In spite of such

barriers, these tools have great potential to help re-shape how educators demonstrate, and institutions assess, excellence in education.

Representation

Education faculty should be represented in all departmental and institutional committees that render decisions for the retention, promotion and tenure of faculty. This ensures adequate interpretation of educator impact and evaluation of excellence which may improve consistency of the application of guidelines across departments and the institution.

Invest in infrastructure, practices and personnel

Central to the origins of burnout in physicians are decreased control over the workload and inefficient use of time due to administrative requirements.⁴³ To address these common issues, institutions might consider the following opportunities to make meaningful change.

Mitigate clerical burden

Hiring and assigning sufficient staff to support educators to eliminate non-essential tasks (e.g., organizing learning materials in learning management systems, messaging learners about course changes) is an opportunity to allow educators to focus on teaching and assessment roles that will likely improve job satisfaction.

Invest in technology

Investing and training educators in the use of easy access/user-friendly education technology and tools that allow the quick completion of required tasks (e.g., applications that can be easily loaded and completed on portable personal devices). Personnel focused on maintaining up-to-date knowledge of new technology that helps programs keep pace with emerging tools making educators' lives easier (and teaching and learning better) further demonstrates institutional commitment to promoting educator quality of life.

• Eliminate inefficient workflows

Opportunities to improve educator efficiency can come in multiple forms and some of the most impactful fall in the following areas.

- o Electronic health records (EHRs): Providing robust training and practice for learners to utilize and navigate EHRs in advance of arriving in clinics and hospitals will eliminate the need for educators to guide and train new learners in the use of this critical tool.
- o Restructure learner assignments to allow longitudinal relationships to develop with educators, staff and sites (and the technology), as is the case with longitudinal integrated clerkships to substantially reduce the load on educators in this domain.
- O Documentation: Learning how to document patient encounters is a critical skill for early medical school learners. Integrating training and time for learners to take on documentation roles (e.g., scribes) in patient care settings from the outset of the training experience can decrease the burden on faculty and serves an important purpose in the education program for junior trainees.
- o Create value-added roles for learners: Capitalize on ways for learners to contribute to the system by 1) integrating students into quality improvement (QI) initiatives that improve clinical workflows and patient outcomes 2) allowing learners to assist with standardization of EHRs for QI initiatives and projects, and 3) training and deploying students as near-peer educators for more junior

students assigned to the same clinical sites. These strategies can increase learner site capacity while maintaining appropriate balance with educator effort and time for teaching.⁴⁴

• Invest in infrastructure to support education research

Clinical and foundational science faculty are typically well versed in the execution of clinical and bench research, and by comparison, funding opportunities in these domains greatly overshadow that available for education research. As a consequence, educators are often ill supported or trained to design, execute and successfully publish education research and are further hampered by the comparatively fewer avenues for disseminating their scholarship. Institutions should consider adopting the following to ensure the success of faculty in producing scholarly products:

- Support the design and implementation of education research labs or centers to provide a framework to support the careers of educators in carrying out high quality education research.⁴⁵
 These centers can provide statistical and grant writing support.
- Create academies of medical educators that are well designed and resourced. This can be a
 powerful central institutional resource for development of faculty skills in research, providing crossdisciplinary mentorship and creating spaces for collaborative education projects.⁵¹
- o Advocate for a greater number of and more significant funding structures in support of education research (e.g., create awards comparable to the NIH RO1 grants, which largely favor bio-scientists, such as EO1s that would elevate education research to a similar stature as that of clinical and foundational science research). Possibilities include training grants aiming to produce future scientist and clinician educators similar to the NIH sponsored T35s or grants like the NIH R21s to fuel innovative teaching approaches and initiatives. Together these could greatly enhance institutional hunger to support education research as well as faculty and trainee development in education.
- o Invest in personnel who are experts in grant-writing and who will lead teams in being aware of funding opportunities, maintaining timelines and supporting education investigators. This effort can fuel the acquisition of extramural funding and grow the education research enterprise.

Human Resource

Institutions that prioritize understanding and working to meet educators' needs, while ensuring learners' needs are also met, will stand a higher chance of sustaining educator well-being. Affording educators the time and space to build community, have autonomy, feel empowered to do their roles well and grow professionally will further the cause. We suggest that institutions:

- Offer and guide educators toward comprehensive offerings of high quality, easily digestible
 professional development/training to meet their needs with consideration for differences across UME
 and GME. Sources both within and external to the institution are valuable. External programs for
 professional educator development such as those available through the Macy Foundation, International
 Association of Medical Science Educators and the AAMC can enhance the ability of educators to
 develop professional networks. Internally developed programming can meet the specific needs of
 educators tailored to institutional priorities and can be tightly aligned with program quality
 improvement efforts.
- Create communities of practice that can guide the development of interventions to make medical education more effective. These can help educators and learners better cope with medical education's

- complexity and simultaneously create a welcoming community with an emphasis on explicitly addressing role modeling, mentoring and reflection.⁴⁹
- Develop peer coaching programs that can further meet the needs of educators and may enhance teaching effectiveness. Embedding programs in the workplace setting and into routine workflows further increases access and feasibility for educator participation.⁵⁰ Incorporating peer feedback instead of primarily learner feedback into overall evaluations of educator effectiveness helps mitigate educator hesitancy to explore contentious or unpopular topics and can alleviate concern around trying out new or innovative approaches to teaching and assessment.^{47,48}
- Commit to measuring the well-being of educators. Partnering with educators to devise plans and strategies to improve well-being when it is low demonstrates the commitment of the institution to valuing the importance of educator job satisfaction. Establishing leadership and resources to facilitate action and improvement further emphasizes the importance of educator well-being to the school and institution. Considerations for how to measure educator well-being are discussed in Chapter 6.
- Professional resources to support educators and their well-being, such as mental-health counseling and
 wellness programs should be available and easily accessible. These are commonly available to learners
 across the education continuum and there should be parity in offerings for educators. Communities of
 practice focused on educator well-being and peer support may also be particularly effective and can
 provide a supportive environment for educators to work through the challenges they face in education.

Symbolic

The symbolic domain addresses the importance of recognition, meaning, purpose, feelings of inspiration and a sense of community. Institutions, education leaders and educators should ensure the following are well integrated into systems as they bear significant weight in promoting the value of educators.

Recognition

Recognizing educator contributions is symbolically important as it affirms an individual's choice to pursue a path that is meaningful to society and the institution. Highly publicized and celebrated educator awards should be offered and recognize individuals who have made significant impact at the UME and GME level. They should originate from students and trainees as well as from administration and peers to fully emphasize the value of educators to various stakeholder groups. Aligning awards with institutional values further cultivates the behaviors desired among the education community and provides concrete ways for educators to demonstrate excellence for promotion and tenure decisions.

Reward

The sense of reward that emerges from sharing feelings of impact and inspiration among a community of educators—and with learners—further promotes educator identity development and well-being. Opportunities for educators to reflect and share challenges and successes, as in communities of practice or periodic educator retreats, ⁴⁹ are potential mechanisms to foster this. Establishing forums and communications to celebrate accomplishments in education, which are on par with scientific accomplishments, can help elevate the importance and recognition for educators at an institutional level.

Sense of community

Academies of educators can be effective investments that serve to raise the prominence of the education mission and create a community and home for educators.⁵¹ Creating a place of belonging

for educators within a community that shares goals, ideals and a common sense of meaning and purpose can help sustain educators especially in the face of stress and adversity.

Political

The pressures to generate clinical revenue or foundational/translational science research grant dollars are at odds with investing in educators to develop their teaching skills. Financial downturns further strain leaders to match expenditures and revenues to attain the best value for their investments. With value considered in terms of cost, choices are often made for financial reasons rather than institutional goals.⁴⁶ To capitalize on opportunities for win-wins in education and the health care system, we suggest the following.

Representation

Assure routine and broad representation of education leadership and educators on system-level committees to provide a voice in the decision-making process. This is particularly important with regard to decisions about what is financially supported and prioritized (e.g., faculty effort to teach, personnel to support educators, technological investments to mitigate administrative burden on educators).

Advocacy

- Advocate for the creation of larger, more numerous and higher profile education grants that will bring institutional recognition to education innovation efforts and the faculty that lead and contribute to them.
- o Create guidelines for educator retention, promotion and tenure committees that identify contributions toward the education mission.
- o Invest in people to develop relationships with philanthropic organizations or individuals who may garner financial support for medical education efforts in both public and private medical education institutions. For publicly funded institutions, identify government relations personnel to partner with education leaders to advocate for expanded state appropriations for education efforts.

The well-being of educators is of fundamental importance as it impacts the quality of the education experience for learners across the medical education continuum. Feeling valued and celebrated for education accomplishments are some of the key factors that contribute to educator well-being. We have summarized a host of opportunities to promote educator well-being, utilizing an organizational framework anchored in structural, human resource, symbolic and political domains. Education and institutional leaders can utilize these recommendations to take actions that we believe are evidence-based and comprehensive. The following chapters will further explore the factors impacting well-being as well as the strategies to improve it.



Case Study 1

Situation

A bench scientist has been studying neuronal development for 19 years and is currently a research associate professor working with a well-established principal investigator (PI). They have done well in the research track, with significant publications and consistent grant funding. They have always been interested in teaching, have formally taught graduate courses and mentored numerous undergraduate and graduate students as well as shared their research in various community outreach sessions. Over the years, their interest in education has

been growing, and they have recognized that they derive the most satisfaction when teaching. At their annual faculty review, this scientist broaches the topic of formalizing their teaching interests by getting involved in teaching medical students with the goal of becoming a full-time medical educator. They ask their department chair for guidance and support in making this transition. Their major concerns are that they do not have formal training in teaching, that they will not have time to devote to teaching and that their current principal investigator will not support them financially for teaching, since they were hired to do research. In addition, they have major concerns about retention, since they are on a year-to-year contract, and about promotion, since the criteria for the research track require evidence for "excellence in investigation."

Approach

The chair suggests that this scientist train up to teach a school of medicine laboratory course under the guidance of the current director, with the goal of taking over his didactic teaching and lab directing duties when he retires in 2-3 years. She also recommends that the scientist become a case-based learning facilitator to gain more experience and to become more involved in medical education. Finally, she recommends a formal track change to address issues of retention and promotion. To help ensure the scientist's success, the chair provides the following support.

- Initiates a conversation with scientist's principal investigator to ensure that they will have protected time for teaching and appropriate financial compensation
- Provides the scientist with a faculty mentoring committee with colleagues who have been heavily involved in medical education to provide guidance on teaching and career
- Sends the scientist to courses and national conferences for formal training in education (lesson and curriculum design, effective instruction methods, assessment, etc.)
- Advocates for the scientist when teaching opportunities arise so that their teaching eventually will be supported at a 1.0 FTE level
- Recommends that the scientist join the institution's Academy of Health Science Educators
- Consults with the office of academic affairs and faculty development to determine requirements for a track change and how to showcase hybrid achievements

Lessons learned

- Provide and fund opportunities for informal and formal training in education
- Support faculty financially during training and teaching
- Connect faculty with mentors to guide career development
- Connect faculty to local and national resources
- Advocate for faculty at departmental and institutional levels

Case Study 2

Situation

An emergency medicine physician graduated from residency four years ago and subsequently completed a two-year fellowship in simulation and medical education. At the completion of the fellowship, she obtained a master's degree in medical education. She is currently a core faculty member in the department of emergency medicine at an urban academic institution. Upon joining as faculty two years ago, this emergency medicine physician became involved in residency education and developed a simulation curriculum for the residency program. She is very interested in medical education and advancing her career in simulation. At the annual faculty review, she expressed her interest in becoming the director of simulation for the emergency medicine

residency program. She then asked the chairperson for guidance and support in obtaining this new leadership position. Her major concerns are lack of protected time to dedicate to the role and lack of funds to develop the simulation program for the department. In addition, she has concerns about her limited research experience as a junior faculty member.

Approach

The chairperson suggests that a formal interview process be set up for the position of director of simulation. The chair has reviewed the responsibilities for the role and states that a percentage of time will be allocated as protected time for the director of simulation role. In addition, the chair recommends creating a specific budget for the simulation program. To gain further research experience, the chair suggests that this emergency medicine physician participate in the departmental monthly research meetings and set up a one-on-one meeting with the department's research director. To help ensure her success, the chair provides the following support.

- Initiates conversation with the emergency medicine residency program director to ensure that she will have an opportunity to apply for the director of simulation position
- Communicates with the director of finance of the department to confirm the protected time allocation for the director of simulation position
- Connects her with colleagues who have been heavily involved in medical education research
- Provides access to a formal institution-wide mentorship program allowing opportunity to connect with mentors who can provide guidance on teaching and career planning
- Recommends that she join local, regional and national organizations focusing on simulation

Lessons learned

- Support faculty with protected time to pursue educational endeavors and leadership roles
- Allocate funds to specific budgets focusing on educational advancement
- Provide and fund opportunities for education research
- Connect faculty with mentors to guide career development
- Connect faculty to local and national resources
- Advocate for faculty at departmental and institutional levels

Take-Home Points

- The low value placed on educators' contributions to the education mission is a problem for the sustainability of the future workforce.
- Educator well-being is directly impacted by how frequently foundational science and clinician educators encounter barriers and disincentives to engaging in education activities.
- We propose numerous actionable ways that institutions can implement to overcome these
 challenges and ensure the sustainability of medical educators and the education mission
 and, ultimately, educator well-being.

Solutions

Structural

- Protect time for educators to be engaged in education roles.
- Be transparent about how the education enterprise and education effort is funded.
- Create equity for educators dedicating time to education, and as possible, create equity for faculty time across mission areas (education, research, administration, clinical).
- Offer incentives to educators for quality education contributions.
- Promote faculty educators using clear and consistent criteria recognizing all areas of education excellence.
- Reduce clerical burden and inefficiencies in carrying out education tasks.
- Invest in education research infrastructure and support for educators to carry out high quality education scholarship.

Human resource

- Support professional development in education.
- Create opportunities for educators to form communities of practice and coaching programs to enhance skill building.
- Measure educator well-being and deploy resources to improve it if it is faltering.

Symbolic

- Foster a culture that recognizes and rewards outstanding educators.
- Create a sense of community amidst education-oriented faculty.

Political

- Ensure representation of education leadership and faculty educators on system-level committees to provide an education voice in institutional decisions.
- Advocate for financial support for education efforts and faculty.
- Advocate for more numerous, substantial and high-profile education grants.

References

- 1. Zibrowski EM, Weston WW, Goldszmidt MA. 'I don't have time': Issues of fragmentation, prioritisation and motivation for education scholarship among medical faculty. *Med Ed.* 2008;42(9):872-878. doi:10.1111/j.1365-2923.2008.03145.x.
- 2. Hu WC, Thistlethwaite JE, Weller J, Gallego G, Monteith J, McColl GJ. 'it was serendipity': A qualitative study of academic careers in medical education. *Med Ed.* 2015;49(11):1124-1136. doi:10.1111/medu.12822.
- 3. Carbajal E. 29 physician specialties ranked by 2021 burnout rates. *Becker's Hospital Review*. Published February 21, 2022. https://www.beckershospitalreview.com/hospital-physician-relationships/29-physician-specialties-ranked-by-2021-burnout-rates.html. Accessed December 7, 2022.
- 4. Porter ME. What is value in health care? NEJM. 2010;363(26):2477-2481. doi:10.1056/nejmp1011024.
- 5. Lamb S; TT. Analyzing the cost of medical education as a component to understanding education value. *JAHSE*. Published June 23, 2021. https://jahse.med.utah.edu/analyzing-the-cost-of-medical-education-as-a-component-to-understanding-education-value/. Accessed December 7, 2022.
- 6. Morse RM, Brooks E, Hines K, Lara-Agudelo. Methodology: 2023 best medical schools rankings US news & world report. Published March 28,2022. https://www.usnews.com/education/best-graduate-schools/articles/medical-schools-methodology. Accessed December 7, 2022.
- 7. Moody J. 10 medical schools with the most applicants. *US News & World Report*. Published June 16, 2022. https://www.usnews.com/education/best-graduate-schools/the-short-list-grad-school/articles/10-medical-schools-with-the-most-applicants. Accessed December 7, 2022.
- 8. Boyer EL. *Scholarship Reconsidered Priorities of the Professoriate*. Princeton, NJ: The Carnegie Foundation for the Advancement of Teaching; 1997.
- 9. Glassick CE, Huber MT, Maeroff GI. *Scholarship Assessed: Evaluation of the Professoriate*. San Francisco, CA: Jossey-Bass; 1997.
- 10. Searle NS, Thompson BM, Friedland JA, et al. The prevalence and practice of academies of medical educators: A survey of U.S. medical schools. *Acad Med*. 2010;85(1):48-56. doi:10.1097/acm.0b013e3181c4846b.
- 11. Searle NS, Teal CR, Richards BF, et al. A standards-based, peer-reviewed teaching award to enhance a medical school's teaching environment and inform the promotions process. *Acad Med.* 2012;87(7):870-876. doi:10.1097/ACM.0b013e3182584130.
- 12. Simpson D, Fincher RM, Hafler JP, et al. Advancing educators and education by defining the components and evidence associated with educational scholarship. *Med Educ*. 2007;41(10):1002-1009. doi:10.1111/j.1365-2923.2007.02844.x.
- 13. Gusic ME, Amiel J, Baldwin CD, et al. Using the AAMC Toolbox for Evaluating Educators: You be the judge! *MedEdPORTAL*. 2013. doi:10.15766/mep_2374-8265.9313.
- 14. Ozuah PO. Undergraduate medical education: thoughts on future challenges. *BMC Med Educ*. 2002;2:8. Published 2002 Jul 30. doi:10.1186/1472-6920-2-8.
- 15. Zinoviev R, Krumholz HM, Pirruccio K, Forman H. Association of graduate medical education with hospital performance and patient outcomes [published correction appears in JAMA Netw Open. 2021 Feb 1;4(2):e211827]. *JAMA Netw Open.* 2021;4(1):e2034196. Published 2021 Jan 4. doi:10.1001/jamanetworkopen.2020.34196.
- 16. Gunderman R, Kanter SL. Perspective: Educating physicians to lead hospitals. *Acad Med.* 2009;84(10):1348-1351. doi:10.1097/ACM.0b013e3181b6eb42.

- 17. Goodall AH. Physician-leaders and hospital performance: is there an association? *Soc Sci Med.* 2011;73(4):535-539. doi:10.1016/i.socscimed.2011.06.025.
- 18. Sadowski B, Cantrell S, Barelski A, O'Malley PG, Hartzell JD. Leadership training in graduate medical education: A systematic review. *J Grad Med Educ*. 2018;10(2):134-148. doi:10.4300/JGME-D-17-00194.1.
- 19. Winthrop University. Scholarly Activity. https://www.winthrop.edu/uploadedFiles/artscience/CAS-RolesRewards-ScholarlyActivity.pdf. Accessed December 7, 2022.
- 20. Gruppen LD, Durning SJ. Needles and haystacks: Finding funding for medical education research. *Acad Med.* 2016;91(4):480-484. doi:10.1097/ACM.0000000000000983.
- 21. Reed DA, Kern DE, Levine RB, Wright SM. Costs and funding for published medical education research. *JAMA*. 2005;294(9):1052. doi:10.1001/jama.294.9.1052.
- 22. Saizan AL, Douglas A, Elbuluk N, Taylor S. A diverse nation calls for a diverse healthcare force. *EClinical Medicine*. 2021;34:100846. Published 2021 Apr 14. doi:10.1016/j.eclinm.2021.100846.
- 23. Diversity in medicine: Facts and figures 2019. AAMC. https://www.aamc.org/data-reports/workforce/report/diversity-medicine-facts-and-figures-2019. Accessed December 7, 2022.
- 24. Konuthula D, de Abril Cameron F, Jonassaint N, et al. Perspectives on anti-Black racism and mitigation strategies among faculty experts at academic medical centers [published correction appears in JAMA Netw Open. 2022 Jun 1;5(6):e2218888]. *JAMA Netw Open.* 2022;5(4):e228534. Published April 1, 2022. doi:10.1001/jamanetworkopen.2022.8534.
- 25. Shanafelt TD. Enhancing meaning in work: a prescription for preventing physician burnout and promoting patient-centered care. *JAMA*. 2009;302(12):1338-1340. doi:10.1001/jama.2009.1385.
- 26. Pololi LH, Evans AT, Civian JT, et al. Faculty vitality-Surviving the challenges facing academic health centers: A national survey of medical faculty. *Acad Med.* 2015;90(7):930-936. doi:10.1097/ACM.00000000000000674.
- 27. Brenner AM, Beresin EV, Coverdale JH, et al. Time to teach: Addressing the pressure on faculty time for Education. *Acad Psychiatry*. 2017;42(1):5-10. doi:10.1007/s40596-017-0851-9.
- 28. Federico Martinez G, Giblin CR, Willis BC. Physician-faculty perceptions towards teaching incentives: A case study at a children's hospital. *Med Educ*. 2021;55(5):604-613. doi:10.1111/medu.14418.
- 29. Common program requirements. Accreditation Council for Graduate Medical Education. https://www.acgme.org/globalassets/PFAssets/ProgramRequirements/CPRResidency2021.pdf. Accessed December 7, 2022.
- 30. Regan L, Jung J, Kelen GD. Educational value units: A mission-based approach to assigning and monitoring faculty teaching activities in an academic medical department. *Acad Med.* 2016;91(12):1642-1646. doi:10.1097/ACM.0000000000001110.
- 31. Miller JC, Andersson GE, Cohen M, et al. Perspective: follow the money: the implications of medical schools' funds flow models. *Acad Med.* 2012;87(12):1746-1751. doi:10.1097/ACM.0b013e3182713b77.
- 32. The basic principles of medical school funding. Association of American Medical Colleges. Webinar. https://aamc.elevate.com/products/the-basic-principles-of-medical-school-funding#tab-product_tab_overview. Accessed September 25, 2022.
- 33. Clyburn EB, Wood C, Moran W, Feussner JR. Valuing the education mission: implementing an educational value units system. *Am J Med*. 2011;124(6):567-572. doi:10.1016/j.amjmed.2011.01.014.

- 35. Stull MJ, Foster KW, Dredla B, Gruppen L, Santen S. The education portfolio. *Acad Med.* 2021;96(9):1368. doi:10.1097/ACM.000000000003596.
- 36. Lowenstein SR, Fernandez G, Crane LA. Medical school faculty discontent: Prevalence and predictors of intent to leave academic careers. *BMC Medical Education*. 2007;7(1). doi:10.1186/1472-6920-7-37.
- 37. Hoffman L, Lufler R, Brown K, et al. A review of U.S. medical schools promotion standards for excellence in education. *The FASEB Journal*. 2019;33(S1). doi:10.1096/fasebj.2019.33.1_supplement.440.3.
- 38. Shanafelt TD, Sloan JA, Habermann TM. The well-being of physicians. *Am J Med*. 2003;114(6):513-519. doi:10.1016/s0002-9343(03)00117-7.
- 39. Gonzalo JD, Dekhtyar M, Hawkins RE, Wolpaw DR. How can medical students add value? Identifying roles, barriers, and strategies to advance the value of undergraduate medical education to patient care and the health system. *Acad Med.* 2017;92(9):1294-1301. doi:10.1097/ACM.000000000001662.
- 40. Thomas PA, Diener-West M, Canto MI, Martin DR, Post WS, Streiff MB. Results of an academic promotion and career path survey of faculty at the Johns Hopkins University School of Medicine. *Acad Med*. 2004;79(3):258-264. doi:10.1097/00001888-200403000-00013.
- 41. Gisondi MA, Michael S, Li-Sauerwine S, et al. The purpose, design, and promise of medical education research Labs. *Acad Med.* 2022;97(9):1281-1288. doi:10.1097/ACM.0000000000004746.
- 42. Mullangi S, Blutt MJ, Ibrahim S. Is it time to reimagine academic promotion and tenure? *JAMA Health Forum*. 2020;1(2). doi:10.1001/jamahealthforum.2020.0164.
- 43. Seifert WE, Strobel HW. Values, RVUs and teaching. Med Sci Educ. 2010 20(2):62-66.
- 44. Pierce JR Jr, Rendón P, Rao D. Peer observation of rounds leads to collegial discussion of teaching. *Teach Learn Med.* 2018;30(2):233-238. doi:10.1080/10401334.2017.1360185.
- 45. Campbell N, Wozniak H, Philip RL, Damarell RA. Peer-supported faculty development and workplace teaching: An integrative review. *Med Ed.* 2019;53(10):978-988. doi:10.1111/medu.13896.
- 46. Cruess RL, Cruess SR, Steinert Y. Medicine as a community of practice: Implications for medical education. *Acad Med.* 2018;93(2):185-191. doi:10.1097/ACM.000000000001826.
- 47. Carlson K, Ashford A, Hegagi M, Vokoun C. Peer coaching as a faculty development tool: A mixed methods evaluation. *J Grad Med Educ*. 2020;12(2):168-175. doi:10.4300/JGME-D-19-00250.1.
- 48. Corral J, Guiton G, Aagaard E. The impact of an academy of medical educators on the culture of an American health sciences campus. *Acad Med.* 2017;92(8):1145-1150. doi:10.1097/ACM.00000000001508.

Chapter 2: Institutional structure

Abigail Ford Winkel, MD, MHPE, and Annie Robinson, MS, CHWC

Chapter Summary

Balancing health care, education and finance means that academic medical centers must balance conflicting incentives. To support the thriving of academic medicine, institutions and educators who work with medical students in undergraduate medical education (UME) and residents and fellows in graduate medical education (GME) need alignment between the academic missions of education, research, service and clinical practice. This requires prioritizing faculty development, delineating clear job structures, providing adequate support for job functions, making necessary changes in culture and mentorship, and identifying promotion pathways in financially viable ways. In this chapter, we identify the unique challenges educators face in the systems within which they work, illustrate how paradoxical expectations can compromise educator well-being, and consider institutional interventions that could help foster well-being. Collaboration between educational and clinical leadership and research missions is essential to ensure educator engagement. Educators may feel isolated in addressing their unique challenges. However, these are common problems that can be addressed by common approaches focused on supporting and celebrating the contributions of these highly valuable members of the professional community.

Educators and Institutions: Competing Priorities and Values

Health care institutions focus on educating the future health care workforce and expanding medical science, and educators occupy a wide range of roles. To meet the realities of health care in the 21st century, business models for medical institutions face many challenges. Clinical care accounts for the majority of revenue, with significant contributions from research grants and smaller proportions from tuition, endowments and other sources. Medical educators work in these real-world settings. A substantial number who are responsible for training medical students and residents also typically bear responsibility for clinical productivity. In addition, some educators face pressure to either maintain research funding or effectively maintain their responsibilities as an administrator. Thus, educators may feel pulled between the competing priorities and values of these various roles without adequate support. As discussed throughout this book, health care professionals are at high risk of burnout. Protecting the well-being of clinical and non-clinical educators and preventing the exhaustion, cynicism and low sense of accomplishment associated with burnout is essential. Humans thrive when we feel a sense of self-efficacy, the belief in our own capacity to be successful in our pursuits. Institutional structures must support the self-efficacy and well-being of educators as they juggle their professional commitments. Health care organizations that both value and give structural support to professional well-being will create cultures that reinforce a sense of competency, accomplishment and satisfaction, which results in a thriving workforce.

The prevalence of burnout, its root causes and its extensive consequences are well-documented.⁵ The Mayo Foundation for Medical Education and Research identifies seven key driver dimensions that fuel burnout:

- 1. Excessive workload and job demands
- 2. Lack of control and flexibility
- 3. Challenges of work-life integration
- 4. Inadequate social support and community at work
- 5. Misalignment with organizational culture and values

- 6. Inefficiency and low resources
- 7. Disconnection from meaning in work⁶

These drivers lead to "exhaustion, cynicism, inefficiency." Among the most significant drivers of burnout that educators experience are workload-related demands due to workplace inefficiencies, excessive work and hours, adverse organizational culture, and barriers to work-life integration. In efforts to control the care costs, academic health centers may become overly lean, providing insufficient administrative support to educators. When educators are pulled in so many directions without adequate support, they feel powerless. Since institutions still expect them to fulfill multifaceted job responsibilities, their sense of self-efficacy naturally wavers or even dissolves. Furthermore, employees in competitive institutions may fear retaliation and experience intimidation. When these feelings are potentially combined with ineffective systems of accountability, educators' feelings of helplessness and burnout can be exacerbated.

Further, educators often hold roles related to their clinical or research work that are situated in other mission areas. In some organizations, an office or department of medical education provides oversight for educators, but not all institutions have this structure. Organizational structures in which reporting lines are blurred for educators can result in poor boundaries for job responsibilities and work expectations, as well as challenges negotiating salaries or other benefits. Further, blurred reporting lines and conflicting communication among leaders across different mission areas can potentially leave educators feeling vulnerable and without a psychologically safe space to help resolve problems.

Of particular concern is the potential for educators to experience moral injury. Moral injury is "a particular type of trauma characterized by guilt, existential crisis and loss of trust that may develop following a perceived moral violation." The guilt, shame, existential conflict and loss of trust associated with moral injury may increase the risk of depression, anxiety, substance use disorders and other problems. As researchers work to understand workplace burnout among physicians and other health care professionals, they recognize elements of moral injury as the duties of their professional lives challenge their own values and norms. However, the term "burnout" may not encapsulate the full scope of the emotional distress suffered by health care professionals or point toward the most effective solutions. Continuing to explore the full range of mental health disorders and treatment needed for health care professionals is important to the welfare of the populations they serve.

Importantly, the above-mentioned drivers of burnout can also be considered in the reverse direction, as targets for organizational interventions to improve engagement. When institutions prioritize and value these facets of professional life, educators feel "vigor, dedication, absorption." Supporting well-being not only benefits the individual, but it also improves the performance of the academic health care system. Many institutions have adopted a position of improving access to mental health services for members of the workforce, which both addresses the problem and sends a powerful message about the value of addressing psychological and emotional needs. When institutions address structural issues, it communicates that they acknowledge and care about the needs of the individuals working in that system, which leads to cultural change. A strong leadership commitment to well-being, putting infrastructure in place and measuring the impact of changes provides feedback about interventions. Recognizing and appreciating those who champion necessary changes and being sure to create fair and inclusive programs all contribute to improvements in culture.

A Structure and Culture Supporting Well-Being: A Paradigm Shift

The implementation of well-being interventions requires institutions to shift their priorities from the immediate focus on financial and reputational success to learner and educator well-being. Ironically, burnout and moral distress in the workforce lead to what institutions want to avoid: compromised performance and quality. In studies of physicians and nurses, burnout predicted medical errors, health care-associated infections, patient mortality ratios and poor interpersonal teamwork. The degree of a physician's depersonalization impacts patient satisfaction with the care they receive and the physician's self-reported adherence to medical advice and job satisfaction. In addition, educators dissatisfied with their jobs are more likely to leave those jobs, which results in high turnover and loss of productivity. When institutions do not make necessary systemic changes to mitigate burnout, burnout increases expenditures via higher turnover rates, medical errors, malpractice claims and lower productivity. Thus, a shift is vital for both educators/researchers and institutions for both quality and cost.

The epidemic of burnout in health care indicates that cultural transformation is imperative. However, how can we replace the presiding culture of burnout with a culture of well-being? While researchers have provided valuable insight into systemic interventions, the efficacy of these proposed solutions is contingent upon the whole community enacting a culture of well-being. Cultural norms need not be passed on; institutions can interrupt their transmission, reassess and augment them. Evolving to meet the needs of modern realities will require a re-examination of how leaders and systems transform and what values guide those transformations.

Finally, the COVID-19 pandemic has disrupted the status quo in the lives of educators. While educators have experienced unforeseen challenges to their well-being, these disruptions have provided insights into addressing

the existing challenges in medical education. As medical institutions work to manage the complex financial realities of the current era, lean management systems have proven that redundancies exist. Amid faculty and staff attrition, greater work burdens have fallen on those educators who continue to teach in medical education.

Organizational Strategies to Improve Professional Well-Being

Several prominent U.S. organizations have developed conceptual models of professional wellbeing as roadmaps for individuals and institutions seeking to mitigate, heal and create alternatives to burnout. Along with the Mayo Clinic, institutions like Stanford University and the National Academy of Medicine have embarked on a mission to better understand and develop conceptual models for navigating this national crisis. The Stanford Model



Fig. 2.1: The Stanford WellMD Model https://wellmd.stanford.edu/about/model-external.html

of Professional Fulfillment¹² illustrates one way that institutions can assess and address well-being issues. (Fig. 2.1). It identifies three dimensions—efficiency of practice, a culture of wellness, and personal resilience—as essential to mitigate burnout and promote fulfillment. The first two are factors that organizations have more control over than individuals do. While many current approaches to addressing well-being focus on the domain of personal resilience through interventions to cope with stressors, addressing structural and cultural issues is more likely to have a lasting impact on well-being.

Similarly, the National Academy of Medicine model of factors affecting clinician well-being and resilience identifies personal factors, skills and abilities that the clinician can control and recognizes the significant contribution of factors outside of the individual's control (Fig. 2.2).¹³ These include organizational factors, the learning and practice environment, and other external factors that influence clinician well-being, the clinician-patient relationship, and ultimately, patient well-being. In both models, the balance between what individuals can do and what institutions can do clearly shows a heavier weight of responsibility falling on institutions to create solutions.



Fig. 2.2: Factors affecting clinician well-being and resilience

This conceptual model depicts the factors associated with clinician well-being and resilience; applies these factors across all health care professions, specialties, settings and career stages; and emphasizes the link between clinician well-being and outcomes for clinicians, patients and the health system. The model should be used to understand well-being, rather than as a diagnostic or assessment tool. Used with permission from the National Academy of Medicine. Learn more at NAM.edu/clinicianwellbeing.

• Relationships and social support

 Sense of meaning Work-life integration

Fig. 2.2: (continued) **EXTERNAL FACTORS SOCIETY & CULTURE RULES & REGULATIONS ORGANIZATIONAL FACTORS** Alignment of societal expectations Accreditation, high-stakes Bureaucracy and clinician's role assessments and publicized quality • Congruent organizational mission and Culture of safety and transparency ratings values • Discrimination and overt and Documentation and reporting • Culture, leadership and staff unconscious bias requirements engagement Media portrayal Human resource policies and • Data collection requirements Patient behaviors and expectations compensation issues Diversity and inclusion Initial licensure and certification Political and economic climates • Harassment and discrimination Social determinants of health Insurance company policies • Level of support for all health care Stigmatization of mental illness Litigation risk team members Maintenance of licensure and Power dynamics certification • Professional development • National and state policies and opportunities practices Scope of practice Reimbursement structure Workload, performance, Shifting systems of care and compensation, and value attributed to administrative requirements work elements LEARNING/PRACTICE **HEALTH CARE RESPONSIBILITIES ENVIRONMENT** Administrative responsibilities · Alignment of responsibility and Autonomy authority • Collaborative vs. competitive Clinical responsibilities environment Learning/career stage Curriculum Patient population Health IT interoperability and Specialty related issues usability/electronic health records Student/trainee responsibilities Learning and practice setting Teaching and research • Mentorship program • Physical learning and practice conditions Professional relationships • Student affairs policies • Student-centered and patientcentered focus Team structures and functionality **INDIVIDUAL FACTORS PERSONAL FACTORS SKILLS & ABILITIES** • Access to a personal mentor Clinical competency Inclusion and connectivity level/experience Family dynamics Communication skills Financial stressors/economic vitality Coping skills Flexibility and ability to respond Delegation to change Empathy Level of engagement/connection Management and leadership to meaning and purpose in work Mastering new technologies or · Personality traits proficient use of technology · Personal values, ethics and Optimizing workflow Organizational skills • Physical, mental, and spiritual well- Resilience skills/practices being Teamwork skills

Institutional Structural Solutions

In 2019, the National Academy of Medicine concluded in a consensus study that institutions should enact a "systems approach to professional well-being" to address systemic and policy changes necessary to foster positive work and learning environments.¹³ Research indicates an array of approaches are needed to mitigate burnout and promote engagement, from micro-interventions that provide coping skills and opportunities to process to macro-interventions addressing structural and cultural barriers to well-being (Fig 2.3). Recommended large-scale interventions to improve educational environments include:

- Cutting back the administrative burden from compliance, regulatory and policy mandates that deplete
 educators by adding time-consuming paperwork that reduces the time and energy they should be
 devoting to their work and their learners.
- Optimizing technological tools for efficiency and effectiveness to eliminate stress and result in higherquality work.
- Eliminating barriers that prevent educators and learners from accessing mental health services due to both stigma and access issues.

Supporting educators at medical institutions starts with 1) clear job descriptions, 2) developing available pathways for development and 3) advancement and support for navigating conflicting responsibilities. Job dissatisfaction and attrition result when these needs are not met, and may be higher among educators in clinical roles who bear greater pressure to generate clinical revenue.9 Further, identifying advocates for faculty who can help clarify boundaries and job responsibilities, negotiate salary, and manage work expectations across activities is essential to mitigate burnout. Effective communication between leadership across different mission areas can inform how best to support educators working within these different capacities. Educators need a psychologically safe space to help resolve problems and to report incidents of mistreatment. It is also important to clearly communicate information about resources available to faculty through human resources or the office of an ombudsperson.



Fig. 2.3: National Academy of Medicine Resource Compendium for Healthcare Workforce Well-Being. https://nam.edu/compendium-of-key-resources-for-improving-clinician-well-being/. Reprinted with permission.

Paths to professional advancement should reflect the importance of these educators to the health of the medical institution. These paths should not narrowly focus on the contributions to research and teaching, as they may not align with the other job duties of educators. As discussed in Chapter 1, promotion pathways play a key role in signaling value to roles occupied by faculty. While research careers remain critical to advancing science and supporting the reputation of academic institutions, expanding pathways to include contributions to

clinical and educational work and defining paths to leadership for those individuals is critical for a flourishing education mission.

Finally, additional steps an organization might take include identifying well-being champions, forming an interdepartmental well-being committee, facilitating narrative medicine sessions, hosting listening sessions with educators and learners to elicit feedback and developing concrete and transparent action plans to address challenges.

Mentoring

Mentorship has played an important role in the development of careers in medicine and has been shown to influence research productivity, promotion, retention and satisfaction across the lifetime of careers. ¹⁴ Mentoring can advance the quality of work across research, patient care and leadership; however, mentorship requires time and skill. While mentoring can be a mutually rewarding activity for both the mentor and mentee, a structure to inform and reward mentoring activities is important to ensure that a diverse group of participants get the support they need in their pathway to medicine. Without formal structure, certain groups may identify mentors more easily and receive higher-quality mentorship than others. Specifically, the value of sponsorship (i.e., advocating for concrete opportunities for sponsees) compared to the type of mentorship focused on relationship development and emotional support has been known to influence career success. In addition, women and faculty from historically excluded and marginalized groups may not have equitable access to mentors. Making mentoring a core strategic priority and providing a scaffold and support for this process will result in a better distribution of mentoring throughout faculty than assuming mentoring will occur based on serendipitous connections.

Research into mentoring programs in academic medicine suggests that mentoring programs should include mentor preparation, planning committees, mentor-mentee contracts, mentor-mentee pairing, mentoring activities, formal curricula and program funding.¹⁵ Supporting mentoring programs with adequate resources, setting aside protected time for mentoring activities, and engaging mentees in selecting mentors were critical to successful uptake of mentoring programs. In addition, a well-being plan for educators and trainees alike should be viewed as essential for creative scholarly work by clinical educators and researchers and should be considered in mentoring discussions alongside other goals.

Accounting for Educational Contributions

In addition to the advancement and recognition of the value of contributions across an institution's mission areas, material rewards or incentives for the work of educators hold promise. Financial rewards for educational contributions, sometimes termed "educational value units" (EVUs), have been implemented in some settings to mirror the rewards given for clinical productivity. These systems identify time-based activities that can be counted and rewarded for educational work. Medical institutions that use this type of system report that educators are willing to engage with this type of system, and as a result, a more balanced distribution of educational responsibilities increases across the faculty body. Research regarding the implementation of EVUs across different institutional settings is needed to identify the most effective means of using these systems; however, EVUs lend promise to ensuring that educators are rewarded and not penalized for integrating innovative and impactful educational work into their job duties.

Scholarly Growth and Research Support

Educators in academic centers aspire to contribute to scholarship in addition to their contributions to clinical work and teaching. When a scarcity of resources prevents educators from investing as much as they would like in the academic part of their job, stress and burnout can result. Thus, the success of academic clinicians is critically dependent on the availability of sufficient resources to support their academic responsibilities. Success in academic medicine requires scholarly output, whether it centers on educational or research scholarship. Workshops for the development of grantsmanship skills that focus on National Institutes of Health research and career development awards are typically highly effective, specifically if they include a strong peer and mentor-driven feedback system. Core facilities providing services such as statistical consultation and experimental design are also excellent in promoting rigorous research.

Supporting Mental Health

While progress has been made in recognizing the importance of health care workers attending to their own mental health, high rates of depression, suicide and other mental health problems remain. With barriers such as privacy concerns, reputation risks, licensing and hospital privileges in place, many individuals do not get the care they need for manageable problems. Institutional culture can celebrate attending to mental health care needs as part of professionalism, rather than evidence of weakness. Internal policies and health care plans must support access to mental health services. Efforts must be made to reduce stigma as well as to address licensing issues that can dissuade individuals from seeking necessary mental health care.¹⁷

Equity and Opportunity

It is critical to enhance workforce diversity to meet the population's needs and address the needs of women and other historically excluded and marginalized groups. Unfortunately, significant gender gaps persist in the representation of women and people of color in the context of medical institution leadership. Multiple studies exploring the influence of gender on careers in medicine suggest that implicit cultural norms of success and leadership are biased towards white, male, cisgender individuals. Women in particular are at higher risk of burnout navigating their professional lives. Barriers impede the academic medical careers of Black, Indigenous and people of color (BIPOC). These issues include the "diversity tax," which is the expectation that BIPOC educators must bear the responsibility of mentoring others; serving as the sole representatives of institutional-based diversity, equity and inclusion issues; seeing more historically excluded and minoritized patients; and ultimately shouldering heavier clinical loads.

Additionally, the distance traveled through education and training might result in greater need to support skill development and address imposter syndrome. Representation of BIPOC in senior leadership and on committees is important to minimize power differentials and empower BIPOC educators.²¹ These factors intersect and complicate professional pathways for women and BIPOC, resulting in greater threats to educator well-being. Increasing transparency around professional roles and expectations could be one way to improve these problems. Other solutions include developing concrete and explicit structures for career development and structured mentorship and addressing issues of implicit bias. Chapter 4 includes a further discussion of these issues.

As discussed in Chapter 3, adequate professional development is critical to support educators in keeping up with transformations in teaching modalities and curriculum design. Being able to effectively adapt to changes in content, format and goals of educational programs requires educators to adopt a growth mindset toward learning themselves. In addition, educators' roles may change as the curriculum evolves, and clear reporting systems within the educational mission can provide scaffolding to allow faculty to develop new skills and integrate feedback in a constructive way.

During this time, recognizing the humanity and the limits of educators has emerged as critically important. Adjusting timelines and work schedules, accommodating leaves of absence and advocating for flexibility have allowed many educators to remain engaged and satisfied in their work despite the additional burdens. With increased reliance on distance learning and remote work, educators and students' comfort with these technologies has increased. While the isolation of the COVID-19 pandemic revealed how important social connections within the academic medical center community are to educators, this flexibility had benefits for work-life integration. Learning how best to use these tools outside of the pandemic is an opportunity for an improved experience for educators.



Case Study

Situation

A pediatrician was offered a role as a medical student clerkship director. The role came with 0.2 FTE dedicated administrative time to support the medical student curriculum, evaluation and mentoring and pursue scholarship in education. Her clinical director has ensured that one day per week, she is not assigned clinical duties; however, she has the same relative value unit (RVU) target and does the same number of weeks attending on the inpatient units each year as her partners. While this arrangement was intended to ensure fairness within the group, with required clerkship activities taking place on the medical student schedule, the pediatrician has more inpatient service weeks than medical school vacation weeks. She winds up covering more of the holiday and summer weeks, rather than being able to take vacation with her family. In addition, her regular panels are frequently overbooked to account for the day per week that she is not seeing patients because she is teaching medical students. Bonuses within the clinical practice are calculated based on RVU targets. She feels that she is penalized for the educational leadership role and suffering from burnout.

Approach

She discussed the situation with her mentor and the vice chair for education who was able to advocate to the department chair. In discussions with the clinical director, her RVU target for the practice was adjusted to reflect the 80% clinical time. Clear communication with the practice manager ensured that the clinic schedule reflected the time set aside for education work, and service weeks were adjusted to 80% of the total.

Lessons learned

 Medical educators are at risk of being overwhelmed by conflicting responsibilities in educational and clinical roles.

- Clear communication around job duties and expectations for educational roles is essential to prevent burnout.
- Mentoring plays an important role in helping identify when issues need to be addressed at a structural level through advocacy to leadership of mission areas.

Take-Home Points

- Educators working in health care risk being pulled between competing missions of clinical, research and educational work.
- Organizations bear responsibility for removing drivers of burnout and creating efficient systems that support a culture of well-being.
- Clear reporting systems, job descriptions and adequate administrative support for educator roles are essential.
- Mentoring programs and faculty development that include research training and promotion tracks valorize the efforts of educators and may improve well-being.
- Removing barriers and ensuring adequate access to mental health care is an institutional imperative to support a flourishing educator workforce.

Solutions

Prioritize well-being at an institutional level

- Undertake an institution-wide survey of well-being and aim to implement a certain number of efficient solutions to identified systematic sources of stress and burnout.
- Adopt a model of well-being and use it as a measure to identify and address issues and assess the impact of interventions.

Reduce unnecessary workload

- Consult with a specialist in how to utilize technology tools to ease the burdens educators face in managing workload.
- Ensure enough clinical and administrative staff have been hired in proportion to the workload.
- Reduce compliance, regulatory and policy mandates for educators to allow their time to be devoted to their work and learners.

Improve professional development and support

- Address implicit bias and develop structures that support women and BIPOC career development and mentorship.
- Develop professional development programs to build skills in educational and administrative roles to support educators.
- Create a sophisticated mentoring program.

Improve mental health

- Create accessible mental health services covered by the institution's insurance for educators that are on par with the services that exist for trainees.
- Address mental health stigma by normalizing mental health experiences (e.g., by facilitating opportunities to share experiences with burnout/well-being in panel discussions or support groups).

Improve work-life integration

- Experiment with strategies that support work-life integration, such as remote flexibility and hybrid opportunities.
- Improve transparency around job duties and compensation to ensure full scope of work is reflected.

References

- 1. Gonzalo JD, Dekhtyar M, Caverzagie KJ, et al. The triple helix of clinical, research, and education missions in academic health centers: A qualitative study of diverse stakeholder perspectives. *Learn Health Syst.* 2020;5(4):e10250.
- 2. Wartman SA, Zhou Y, Knettel AJ. Health reform and academic health centers: Commentary on an evolving paradigm. *Acad Med.* 2015;90(12):1587-1590. doi:10.1097/ACM.0000000000000948.
- 3. Singh R, Volner K, Marlowe D. Provider burnout. *StatPearls*. 2022. https://www.ncbi.nlm.nih.gov/books/NBK538330/. Accessed December 8, 2022.
- 4. Bandura A, Freeman WH, Lightsey R. Self-efficacy: The exercise of control. *J Cogn Psychother*. 1999;13:158-166. doi:10.1891/0889-8391.13.2.158.
- 5. Dyrbye LN, Shanafelt TD, Sinsky CA, et al. Burnout among health care professionals: A call to explore and address this underrecognized threat to safe, high-quality care. *National Academy of Medicine*. 2017. https://nam.edu/burnout-among-health-care-professionals-a-call-to-explore-and-address-this-underrecognized-threat-to-safe-high-quality-care. Accessed December 8, 2022.
- 6. Shanafelt TD, Noseworthy JH. Executive leadership and physician well-being: Nine organizational strategies to promote engagement and reduce burnout. *Mayo Clin Proc.* 2017:92(1):129-146.
- 7. Jinkerson JD. Defining and assessing moral injury: A syndrome perspective. *Traumatology*. 2016 Jun;22(2):122.
- 8. Kopacz MS, Ames D, Koenig HG. It's time to talk about physician burnout and moral injury. *The Lancet Psychiatry*. 2019 Nov 1;6(11):e28.
- 9. Wallace JE, Lemaire JB, Ghali WA. Physician wellness: a missing quality indicator. *Lancet*. 2009;374(9702):1714-21. doi:10.1016/S0140-6736(09)61424-0.
- 10. Girod SC, Fassiotto M, Menorca R, Etzkowitz H, Wren SM. Reasons for faculty departures from an academic medical center: a survey and comparison across faculty lines. *BMC Med Educ*. 2017;17(1):8. doi:10.1186/s12909-016-0830-y.
- 11. Kashima Y, Bain PG, Perfors A. The psychology of cultural dynamics: What is it, what do we know, and what is yet to be known? *Ann Rev Psychol*. 2019;70:1,499-529.
- 12. The Stanford Model of Professional FulfillmentTM. Stanford Medicine. https://wellmd.stanford.edu/about/model-external.html. Accessed December 8, 2022.
- 13. Factors affecting clinician well-being and resilience conceptual model. National Academy of Medicine. Published 2018. https://nam.edu/clinicianwellbeing/resources/factors-affecting-clinician-well-being-and-resilience-conceptual-model/. Accessed December 8, 2022.
- 14. Choi AMK, Moon JE, Steinecke A, Prescott JE. Developing a culture of mentorship to strengthen academic medical centers. *Acad Med.* 2019;94(5):630-633. doi:10.1097/ACM.0000000000002498.
- 15. Kashiwagi DT, Varkey P, Cook DA. Mentoring programs for physicians in academic medicine: A systematic review. *Acad Med.* 2013;88:7, 1029-1037.
- 16. Regan L, Jung J, Kelen GD. Educational value units: A mission-based approach to assigning and monitoring faculty teaching activities in an academic medical department. *Acad Med.* 2016;91(12):1642-1646. doi:10.1097/ACM.000000000001110.
- 17. Moutier C. Physician mental health: an evidence-based approach to change. *J Med Regul.* 2018 Jul;104(2):7-13.

- 18. Winkel AF, Telzak B, Shaw J, et al. The role of gender in careers in medicine: A systematic review and thematic synthesis of qualitative literature. *J Gen Intern Med.* 2021;36(8):2392-2399. doi:10.1007/s11606-021-06836-z.
- 19. Hoff T, Lee DR. Burnout and physician gender: What do we know? *Med Care*. 2021;59(8):711-720. doi:10.1097/MLR.00000000001584.
- 20. Garcia LC, Shanafelt TD, West CP, et al. Burnout, depression, career satisfaction, and work-life integration by physician race/ethnicity. *JAMA Netw Open*. 2020;3(8):e2012762. doi:10.1001/jamanetworkopen.2020.12762.
- 21. Campbell KM, Hudson BD, Tumin D. Releasing the net to promote minority faculty success in academic medicine. *J Racial Ethn Health Disparities*. 2020;7(2):202-206. doi:10.1007/s40615-020-00703-z.

Acknowledgements

The authors thank Luis Fernando Santana, PhD, for his contributions to earlier drafts of the section of this chapter on scholarly growth and research support.

Chapter 3: Responding to/functioning in the larger picture of academic medicine

Julie Youm, PhD; Senthil Kumar Rajasekaran, MD; and Khanh-Van Le-Bucklin, MD, MEd

Chapter Summary

"Help me remember why I chose to heed this call.

Condemned to hold the skies on shoulders not of gods, the weight of heavy flesh bears mightily on my soul..."

An excerpt from *On Burnout: An Ode to Atlas*by Sherine Salib, MD¹
Reprinted with permission

Educators are a principal and irreplaceable constituent in the development of the physician workforce. Preclinical educators provide the knowledge and skills that learners use to engage their clinical training following the long-standing adage of "see one, do one, teach one" (SODOTO), with the "see one" aspect of this tripartite approach being predicated on the need for an experienced educator to model clinical skills and behaviors. The advances in evidence-based learning principles, educational technologies and instructional pedagogies challenge the applicability of the SODOTO approach to modern medical education,² yet the paramount role of the educator remains consistent. What also remains consistent are the inherent challenges of educators within the larger picture of academic medicine including competing demands of patient care, research and administrative responsibilities. In this chapter, we focus on the curricular and accreditation stressors educators face in the specialized context of academic medicine and propose ways in which these stressors can be addressed to preserve educator well-being.

Curricular Stressors

The education mission of academic medicine is the reason a medical school can call itself a school.³ This mission focuses on the preparation and training of medical students and residents for careers as practicing physicians. Allopathic and osteopathic medical education programs establish curricula that aligns their educational program objectives with the standards set forth by the Liaison Committee on Medical Education (LCME) and the Committee on Osteopathic College Accreditation (COCA). Medical school and residency curricula are then continually changing to meet accreditation standards, as well as the ongoing evolution of the field reflecting scientific advancements, medical breakthroughs, systems-related upgrades and societal issues.⁴ This pervasive state of change creates stressors on educators, particularly those related to curriculum and teaching.

Redesigns of educational programs and curricular reform

Educators experience a direct impact and notable stressors when schools undertake curricular reform. LCME and COCA standards require a regular review of the curriculum by allopathic and osteopathic medical schools and related, corresponding action by educators to address areas that are not satisfactory.^{5,6} While curricular

reform commonly occurs from local institutional pressures, such as meeting accreditation standards and curriculum review processes, national movements in medical education also prompt medical schools to respond with shifts in their curriculum. Curricular response to a national movement is not always an explicit accreditation requirement yet may feel like an imperative for schools to maintain a competitive status when compared with peer institutions. For example, emerging trends identified in health professions education include interprofessional education, education on the social determinants of health, competency-based medical education (CBME), artificial intelligence and the integration of new educational and information technologies.⁷ Attempts to implement one or all of these trends will be met by conflicts, including the limited number of curricular hours, faculty availability and content expertise, and the structural constraints of an existing curriculum.

Further, curricular reform often represents a dramatic transformation focused on contemporary areas like health systems science and diversity, equity and inclusion (DEI) that current educators may not have been formally taught during their own academic experiences.⁸ Developing entirely new courses integrating novel content along with learning principles and pedagogies that are foreign to educators increases the burden on them. Educators may feel ill-equipped to support the implementation of curricular reform without appropriate and relevant professional development. They may develop concerns that their content expertise and teaching opportunities could be eliminated. Additionally, associated curricular stressors arise from administrative and scheduling challenges due to the need to balance existing teaching, clinical and research schedules with potential travel to distributed academic sites.

Beyond content and structural changes, assessments related to curricular reform also contribute greatly to educator burdens. For example, educators in graduate medical education (GME) experience pressures while assessing specialty specific milestones through clinical competency committees (CCC). CCCs vary widely in membership, structure and process, likely due to a variety of factors including the limited description of CCCs in the Accreditation Council for Graduate Medical Education (ACGME) common program requirements, the range in program sizes and availability of institutional resources. This creates barriers to establishing generalized best practices and professional development around CCC work. More importantly, educators participating in CCCs often engage in secondary roles beyond their primary one to determine trainee achievement of milestones with added work related to providing feedback, developing remediation strategies and participating in continuous quality improvement activities for the residency program.⁹ When assessment methods around milestones change due to curriculum reform, there is often little guidance to educators on how to approach such changes and limited bandwidth for them to react.

Educators in undergraduate medical education (UME) encounter similar pressures due to assessment needs resulting from curricular reform. For example, CBME often requires a significantly greater number of assessments of students, both formative and summative, and ideally in a workplace-based setting. The impact and dependence on educators to reliably assess students within CBME becomes a notable barrier to implementing such programs. Related administrative responsibilities and requirements to learn new systems (e.g., systems for evaluation, scheduling and curriculum mapping to support CBME) create more barriers and lead to change fatigue.

A broader issue related to assessment revolves around collecting, analyzing and reporting learner performance data. The lack of sufficient assessment data, missing evaluations or poor-quality data places educators in a challenging decision-making position that requires additional effort to seek supporting data and evaluations or drawing conclusions that may not be evidence-based. Implementing effective solutions to this data problem is, however, time-consuming, costly and labor intensive, leaving the onus on educators to derive their own manually driven, inefficient options.

Addressing curricular stressors is imperative for educator well-being and their probable impact on faculty educator attrition. A report by the Association of American Medical Colleges (AAMC) found an average overall faculty attrition rate of 38% and a higher rate of 43% of first-time assistant professors leaving academic medicine within a 10-year period. Faculty who identified as women and non-white demonstrated disproportionately higher rates of attrition leading to implications for institutional DEI efforts as well. Educators and trainees from backgrounds historically excluded from medicine are often asked or expected to take on the bulk of the efforts around curricular development and teaching on DEI content, typically without adequate compensation, adding to the diversity tax and contributing to burnout. Though diversity officers are being hired at schools to fill these gaps around DEI curriculum and support, there are limited standards for teaching and training educators for this role. In turn, potential stressors related to the marginalization of the diversity officer role in the hierarchy of academic medicine compound the marginalization that was likely already present based on the race of the educator carrying forth the DEI work.

Teaching to the test

The United States Medical Licensing Examination® (USMLE®) Step 1 is the first part of a three-step examination for medical licensure in the U.S. Preparing students to perform successfully on USMLE Step 1 alongside meeting a medical school's graduation requirements is another considerable curricular stressor on educators. Educators are assigned the explicit expectation to teach to the learning objectives of a course and program while also addressing the expectation to ensure student readiness for USMLE Step 1. The latter is arguably the greater priority from the student perspective¹² and an implied priority for schools. Accordingly, the pressures of "teaching to the test" for student satisfaction and program reputation can be repressive to the creativity and innovation of educators, and a decreased focus on teaching around their content of expertise evokes concerns about their ability to hold onto their teaching responsibilities. Contributing to this stressful situation is the availability of an increasing number of third-party study aids that focus on USMLE Step 1 preparation. When the intrinsic value of an educator's work is diminished, curricular teaching hours on top of competing demands affect the motivation to teach.¹³

The transition of USMLE Step 1 to pass/fail scoring in January 2022¹⁴ and a national reaction to find alternative markers to assess students' readiness for residency has the potential to renew primacy for new curricular content and curriculum design. The Comprehensive Osteopathic Medical Licensing Examination of the United States (COMLEX-USA) Level 1 from the National Board of Osteopathic Medical Examiners (NBOME) is akin to the USMLE Step 1 for medical licensure of osteopathic physicians. In a move aimed at student wellness, the COMLEX-USA Level 1 also went to pass/fail scoring in May 2022.¹⁵ The implications of this change to the Step 1 and Level 1 exams are broad in scope and could bring relevance and value back to the role of the educator if they result in a movement beyond "teaching to the test."

Adapting to innovations and the use of educational technology

In March 2020, the world at-large found itself forced into a pandemic response due to the surge of COVID-19. Educational institutions at all levels were impacted as COVID-19 guidelines required physical distancing and strict limits on mass gatherings. These restrictions affected the ability to conduct direct, face-to-face instruction for students and residents. Educators suddenly found themselves pivoting to emergency remote teaching with little or no preparation. At times, students were removed from their clinical rotations and residents experienced disruption in their in-person learning. Virtual and hybrid instruction during COVID-19 demanded that faculty build a teaching skill set that included synchronous instruction on video-conferencing platforms, creating instructional videos for teaching asynchronously and supporting remote active learning (e.g., through small-group activities and student assessment). The toll on educator well-being during this time was immeasurable as the uncertainty and lack of knowledge around the disease made it difficult to plan for curricular alternatives and pressures to provide sufficient required clinical experiences for students and residents grew. Clinical responsibilities for educators simultaneously increased as it became paramount to address the increased patient load. Subsequently, research agendas were put on hold contributing to concerns about promotion and advancement, and inordinate personal demands became arduous to balance.

The motivations for educators to teach appear to be primarily altruistic. Educators have a desire to help learners become good doctors. ¹⁶ Despite such admirable motivations, educators largely come into their teaching roles without formal training in pedagogy. Reflections on the response to the COVID-19 pandemic from an academic medicine perspective highlight the need to train faculty to be flexible, adaptive and evolving teachers. In a study that aimed to assess medical educator knowledge around pedagogy, content and technology knowledge as well as the intersections of these three types of knowledge, faculty reported having a high level of content knowledge, while rating their pedagogical and technological knowledge lower. ¹⁷ Knowledge related to technology at any intersection of knowledge type (e.g., technological and pedagogical knowledge) was also significantly lower. Implications of such findings emphasize the significance of educator development around teaching and especially around the use of technology for teaching.

A critical lesson learned here is to recognize the need to prepare educators now for what uncertainties may follow with the ultimate goal of decreasing their burdens and burnout. Two recommendations¹⁸ that resulted from the collaborative work among members of the American Medical Association Accelerating Change in Medical Education Consortium and the AMA Council on Medical Education in response to the COVID-19 pandemic are to 1) "leverage shared online or commercial resources as appropriate rather than generating new institutionally based materials or solutions" and 2) "provide training for educators in tools for remote education, simulation and for engaging learners in remote care settings." Such recommendations underscore the value of reducing educator burden and increasing support through training. Chapter 5 further explores insights gleaned from the COVID-19 pandemic to inform the future age of well-being in academic medicine.

Accreditation Stressors

Medical education programs leading to the MD degree in the United States are accredited by the LCME. Programs leading to the DO degree are accredited by the COCA. The ACGME accredits medical residency and fellowship programs.

Medical education accreditation has become more complex, data focused and resource intensive over the years. For example, in UME, the LCME reformatted its accreditation standards in 2002 to clearly link team findings and data to specific standards. This change led to greater clarity and precision, but also an associated increase in severe action decisions. ¹⁹ Similarly, COCA regularly reviews and revises its accreditation standards, requiring schools to timely adjust to these changes to maintain accreditation. In 2012, the ACGME added a second type of survey visit in its adoption of the Clinical Learning Environment Review. These visits occur every 18 to 24 months in addition to institutional and program accreditation visits. Maintenance of medical education accreditation is a high-stakes endeavor for institutional sponsors. Accreditation status may be tied to federal and state funding, as well as trainee eligibility for licensure and board certification. For example, in 2014, the Joan C. Edwards School of Medicine at Marshall University cited the high cost of an adverse accreditation action in terms of direct expenses, opportunity costs, reputation, recruitment, grant procurement and morale.²⁰ The downstream effects of a severe accreditation decision contribute to the stress experienced by medical education leaders and educators as it relates to their roles in the accreditation process.

In UME, the LCME requires schools to practice continuous quality improvement, with a systematic and ongoing process for monitoring school-selected accreditation standards.²¹ COCA requires schools to be compliant with all standards and meet all elements to achieve a full 10-year accreditation cycle.⁶ When schools only focus on accreditation requirements right before a survey visit, they risk falling below these standards between visits. It is not unusual for a school to appoint a faculty accreditation lead for a survey visit who does not have responsibility for the ongoing management of accreditation compliance outside of survey visit preparation. The rush to move a school into compliance in short order creates undue stress for educators and others involved in the accreditation process, both before and after the accreditation review.

Sometimes schools assign the task of accreditation oversight to the same people who are already burdened with other major responsibilities. These educators are placed in the unenviable position of managing multiple competing priorities. In GME, the ACGME requires the appointment of a designated institutional official (DIO) to focus on institutional accreditation oversight. In hospital accreditation, the concept of being in compliance at all times is an accepted expectation, and resources are allocated to the ongoing accreditation readiness of a hospital. When this culture of continuous quality improvement for UME is not supported, it creates undue stress and burnout for educators who are tasked with this work in addition to their primary duties.

In GME, program directors often carry heavy clinical responsibilities on top of their accreditation oversight duties. While the ACGME has made efforts to ensure minimum time requirements for DIOs and program directors to focus on their administrative responsibilities, these requirements are not uniformly applied and may come at a cost to the sponsoring department. Faculty educators who carry these roles may suffer a loss of clinical incentive revenues or may take on additional clinical duties to avoid a reduction in pay. Institutions can respond to this disparity by developing a transparent funds flow model to ensure protected time and prevent educator burnout.

Finally, other accreditation-related factors that contribute to educator stress include the use of student and resident satisfaction surveys as quality measures. When triangulated with other performance metrics, learner surveys provide valuable information for further investigation. The fear of survey retaliation by learners when

educators' decisions do not align with learner expectations, however, can contribute to low educator morale and burnout. At its worst, educators may fail to move forward with positive initiatives to avoid learner dissatisfaction, especially if they lack the time or emotional reserve to effectively communicate and manage change with learners.

Possible Solutions

In the sections above, we outline common curricular and accreditation stressors that educators face in the context of academic medicine. Here we propose some possible solutions to address these stressors.

Professional development

A common thread among the curricular stressors described in this chapter is a lack of professional development for medical educators. Many receive little to no formal training on how to be effective teachers, even when they acquire education leadership roles.²² Because medical education has traditionally been an apprenticeship, physicians are often assumed to be able to teach without focused training or competency assessment on the complex teaching skills necessary to succeed as a medical educator.²² Some instructors may teach with the mindset "I was taught this way" or "When I went to medical school." These beliefs can lead to suboptimal learning environments that compromise learner performance and satisfaction. These factors may be exacerbated at institutions where educators, as well as course and clerkship directors, do not have a direct reporting relationship to medical education leaders. Such stressors can be potentially mitigated through the establishment of a department of medical education, assigning partial medical education department full-time equivalents (FTEs) to core educators and developing memorandums of understanding that provide clear reporting lines and identified mentors for these faculty. Investing in teaching academies can also improve faculty participation in professional development activities and equip them with requisite skills to effectively perform their educator functions.²³

In planning professional development activities, it is important to adopt accessible and adaptable approaches as teaching often falls on a full plate of responsibilities for educators including research, patient care, maintenance of time-consuming electronic health records and pressures to meet relative value unit targets.⁸ Professional development around curricular change is essential and since clinical and basic science faculty may be affected by curricular changes in different ways due to competing priorities of patient care and research, a unified approach to educator development may not always apply. Van Schaik⁸ presents eight guiding principles for faculty development that the University of California, San Francisco followed in preparing faculty for curriculum reform. These include the following:

- Build on existing resources, networks and communities to avoid redundancy of effort; leverage existing expertise and identify gaps.
- Promote collaboration between content experts and faculty developers to mitigate turf wars, resistance to change and create buy-in.
- Develop curriculum leaders and faculty developers to ensure sustainability and capacity for future reform.

Professional development around technology should also be a priority. Technology plays a critical role in implementing and catalyzing innovation in medical education. For example, advances in virtual reality,

augmented reality, augmented intelligence, mixed reality and artificial intelligence present opportunities for innovation in learning clinical procedures and skills. The integration of tablet devices²⁴ and mobile medical devices such as handheld ultrasound offer other examples of technology used to motivate educational engagement for both educators and learners in medical education. Integration of such technologies enables opportunities for more engaging and active remote and hybrid education, delivery modalities that were emphasized during the COVID-19 pandemic. Providing educators with training on the use of technology and exposing them to the advances in the field will give them the building blocks to envision and lead cutting-edge curriculum and complement their altruistic motivations to teach.

Beyond professional development sessions on topics related to medical education and teaching, sessions designed to address leadership and self-development among educators can address their burnout and disengagement. Training sessions specifically addressing topics such as negotiation strategies and conflict management can serve as a mitigation approach to preserve educator vitality.

Curriculum review process

To achieve a balance between a proactive versus a reactive response to external influences on curriculum, medical education leadership need to pay particular attention to accreditation elements focused on curricular design, review and revision/content monitoring. LCME Element 8.3 describes the need to have a process for formal review of the phases of the curriculum including "horizontal and vertical integration of curriculum content and whether sufficient content is included and appropriately placed in the curriculum related to each of the medical education program objectives." This curricular review process should provide educators the oversight necessary to determine how, when and if national movements and emerging trends are addressed and help mitigate stressors to educators as a result. Similarly, COCA Standard 6 requires the periodic and regular review and revision of an osteopathic curriculum by faculty. When a well-defined process for curriculum review and monitoring is implemented as required by accreditation standards and educators are given a primary voice in this process, their needs and concerns can be made explicit and educator buy-in for curricular initiatives has a chance to thrive over educator burnout.

Institutional culture

Medical school senior leadership (e.g., chairs and dean) must create a culture of recognizing and rewarding excellence in education. While there are accreditation elements that address institutional support for educator development, promotion and tenure, specific accreditation mandates requiring a formal mentorship for junior and mid-career educators can promote institutional commitment to creating a culture of educational excellence among educators. Chapter 2 further explores the impact of institutional structure in supporting educator well-being.

Accreditation support

Institutional investment and commitment to the accreditation process are critical for the health of educators, as well as the vitality of the school. Institutions can support educators involved in accreditation oversight through the provision of time, staff and resources to carry out the duties of the position.

For example, the LCME recommends the appointment of a senior level administrator who is knowledgeable in LCME accreditation and has appropriate time allocated for the responsibility of monitoring and managing a school's continuous quality improvement process.²¹ This administrator would ideally have access to experts in program evaluation, data analysis and information technology as well as sufficient time and resources to carry out an effective compliance system. COCA accreditation leaders similarly require protected time and resources to ensure ongoing compliance with accreditation standards. Many medical schools have formed units focused on education accreditation, compliance and quality to support educators overseeing these critical responsibilities.

In GME, ACGME institutional and program requirements specify protected time for directors and support staff. Institutions can support faculty directors by ensuring that clinical schedules align with the ACGME-required protected time and that forfeited clinical incentives can be earned through education-based achievements. Program coordinators with other department administrative responsibilities should be monitored for workload to avoid conflicts in duties. In addition, given the evolving and growing accreditation responsibilities, institutions should reevaluate designated protected time on a regular basis to ensure it aligns with required effort.

Institutions can further promote educator well-being and education innovation by adopting a growth mindset toward accreditation, celebrating overall success rather than reprimanding individual negative accreditation findings. A recommendation from the AMA in response to the COVID-19 pandemic suggested that organizations "advocate for appropriate adjustments in accreditation processes and assist leaders of educational programs to follow evolving recommendations and requirements." Nationally, a collaborative between accrediting bodies and professional organizations, the Coalition for Physician Accountability, was successful in responding to challenges faced by educators during the COVID-19 pandemic. Such a collaborative could continue to convene on matters of mutual interest that promote educator well-being in accreditation, such as institutional support, professional training and survey instrument development.



Case Study

Situation

University X is embarking on a curriculum redesign, moving from a lecture heavy, discipline-based curriculum to organ-based modules with an emphasis on active learning. At the same time, the school is preparing for an upcoming accreditation visit. The accreditation self-study uncovered several areas of noncompliance, and the medical education team has not been able to fully focus on curriculum reform while working with key stakeholders to address these accreditation concerns. Meanwhile, students openly voice discontent to the dean over the curriculum's lack of focus on USMLE Step 1 preparation at a town hall meeting. The medical education team feels overwhelmed and worries their reputation is on the line if the school suffers an adverse accreditation outcome.

Approach

The vice dean met with the dean to discuss these concerns. The school formed an education compliance and quality office with an associate dean and support staff to oversee the school's accreditation activities for the upcoming survey visit. This team was also charged with developing a system for continuous quality

improvement to maintain accreditation compliance on an ongoing basis. New faculty and staff were hired to develop and implement the redesigned curriculum. The redesigned curriculum included the addition of USMLE Step 1 practice questions as formative assessments. One of the most important actions taken by the dean was to emphasize to all stakeholders the value of curriculum reform and accreditation preparation. Importantly, a great deal of stress was removed from the medical education team when the dean reassured them that no matter the outcome of the accreditation visit, any identified concerns, including professional development opportunities focused on learning approaches and pedagogies, would be faced together as a team and resources would be allocated to address them. These efforts resulted in an acceleration of curricular innovations which were positively received by students and implementation of accreditation-related compliance systems that led to a successful accreditation survey visit.

Lessons learned

- Organizations need to conduct a regular inventory of administrative duties and ensure alignment of protected time for educators to successfully carry out these tasks.
- Institutions that invest in prospective accreditation oversight can avoid the direct and opportunity costs associated with an adverse outcome.
- Educators and leaders need affirmation for their successes and reassurance that they will not be reprimanded for institution-based concerns leading to accreditation findings, such as inadequate or inappropriate staffing.
- Professional development opportunities should be offered to help create a foundation for accreditation and curricular redesign efforts as well as personal and professional growth.

Take-Home Points

- In addition to the stressors that arise from the intersection of clinical, research and teaching responsibilities, educators in academic medicine are consistently confronted with stressors related to external forces, including curricular reform, emerging trends and accreditation requirements.
- Common curricular stressors arise from curriculum redesign and reform efforts that call upon
 educators to develop content, use learning approaches and embrace pedagogies in which they
 have little training or expertise. Implicit and explicit requirements to teach to USMLE and COMLEXUSA hinder educators' creativity and innovation.
- Additional stressors arise from accreditation requirements that require continuous quality improvement of a medical education program and tax educators who must respond to existing responsibilities with added accreditation tasks.
- Supporting educators through professional development in areas including teaching, accreditation, leadership skills and technology as well as enabling their voice in the curriculum review process are some steps that can be taken to address educator well-being.
- Institutional support through protected time for curricular and accreditation efforts and rewarding excellence in teaching offer another mitigation approach to preserve educator vitality.

Solutions

Addressing curricular stressors

- Move beyond "teaching to the test" to help bring relevance and value back to the role of the educator by finding alternative markers to assess students' readiness for residency and renewing primacy for new curricular content and curriculum design.
- Review the eight guiding principles for faculty development around curricular reform.⁸
- Establish a teaching academy or other formal faculty development opportunities to promote interdepartmental education, support the teaching faculty and encourage educational innovation and scholarship.
- Engage educator oversight in the curricular review process to determine how/when/if national movements and emerging trends are addressed and help mitigate stressors to educators as a result.
- Establish a department of medical education, assigning partial FTEs to core educators and developing memorandums of understanding that provide clear reporting lines and identified mentors for these faculty.
- Create a culture of recognizing and rewarding excellence in education led by the medical school senior leadership (e.g., the chair and dean).
- Design professional development around addressing leadership and self-development among educators (e.g., negotiation strategies, in addition to topics related to medical education).

Addressing accreditation stressors

 Appoint a senior administrator who is knowledgeable in LCME accreditation and has time allocated for the responsibility of monitoring and managing a school's continuous quality improvement process.

- Develop a transparent funds flow model to ensure protected time for accreditation efforts and reevaluation on a regular basis to ensure the time aligns with required effort.
- Support a national collaborative between accrediting bodies and professional organizations to respond to challenges faced by educators such as those during the COVID-19 pandemic.

References

- 1. Salib S. On Burnout: An ode to Atlas. Acad Med. 2021;96(5):680.
- 2. Kotsis SV, Chung KC. Application of the "see one, do one, teach one" concept in surgical training. *Plast Reconstr Surg.* 2013;131(5):1194-1201.
- 3. Kanter SL. Advancing the education mission. *Acad Med.* 2012;87(8):991-992.
- 4. Policy priorities to improve our nation's health: How medical education is changing. AAMC. https://www.aamc.org/media/19151/download?attachment. Accessed December 8, 2022.
- 5. Data collection instrument for full accreditation surveys. LCME. October 2021. https://lcme.org/publications/#DCI. Accessed December 8, 2022.
- 6. Commission on Osteopathic College Accreditation Standards. COCA. August 2021. https://osteopathic.org/accreditation/standards/. Accessed December 8, 2022.
- 7. Thibault GE. The future of health professions education: Emerging trends in the United States. *FASEB Bioadv.* 2020;2(12):685-694.
- 8. van Schaik SM. Accessible and adaptable faculty development to support curriculum reform in medical education. *Acad Med.* 2021;96(4):495-500.
- 9. Ekpenyong A, Edgar L, Wilkerson L, Holmboe ES. A multispecialty ethnographic study of clinical competency committees (CCCs). *Med Teach*. 2022;1-9.
- 10. Mejicano GC, Bumsted TN. Describing the journey and lessons learned implementing a competency-based, time-variable undergraduate medical education curriculum. *Acad Med.* 2018;93(3S Competency-Based, Time-Variable Education in the Health Professions):S42-S48.
- 11. Alexander H, Lang J. The Long-term retention and attrition of U.S. medical school faculty. https://www.aamc.org/media/9976/download?attachment. Accessed December 8, 2022.
- 12. Carmody JB, Green LM, Kiger PG, et al. Medical student attitudes toward USMLE Step 1 and health systems science a multi-institutional survey. *Teach Learn Med.* 2021;33(2):139-153.
- 13. Wisener KM, Eva KW. Incentivizing medical teachers: exploring the role of incentives in influencing motivations. *Acad Med.* 2018;93(11S Association of American Medical Colleges Learn Serve Lead: Proceedings of the 57th Annual Research in Medical Education Sessions):S52-S59.
- 14. USMLE Step 1 transition to pass/fail only score reporting. USMLE. September 2021. https://www.usmle.org/usmle-step-1-transition-passfail-only-score-reporting. Accessed December 8, 2022.
- 15. COMLEX-USA Level 1 to eliminate numeric scores. NBOME. https://www.nbome.org/news/comlex-usa-level-1-to-eliminate-numeric-scores/. Accessed December 8, 2022.
- 16. Dahlstrom J, Dorai-Raj A, McGill D, Owen C, Tymms K, Watson DA. What motivates senior clinicians to teach medical students? *BMC Med Educ.* 2005;5:27.
- 17. Youm J, Corral J. Technological pedagogical content knowledge among medical educators: what is our readiness to teach with technology? *Acad Med.* 2019;94(11S Association of American Medical Colleges Learn Serve Lead: Proceedings of the 58th Annual Research in Medical Education Sessions):S69-S72.
- 18. Protecting the education mission during sustained disruption: Organizational steps to support medical educators. American Medical Association. https://www.ama-assn.org/system/files/protecting-education-mission.pdf. Accessed December 8, 2022.
- 19. Hunt D, Migdal M, Eaglen R, Barzansky B, Sabalis R. The unintended consequences of clarity: reviewing the actions of the Liaison Committee on Medical Education before and after the reformatting of accreditation standards. *Acad Med.* 2012 May;87(5):560-6.

- 20. Miller B, Dzwonek B, McGuffin A, Shapiro JI. From LCME probation to compliance: the Marshall University Joan C Edwards School of Medicine experience. *Adv Med Educ Pract*. 2014;5:377-382.
- 21. Implementing a system for monitoring performance in LCME Accreditation Standards. LCME. https://lcme.org/publications/. Accessed December 8, 2022.
- 22. Srinivasan M, Li ST, Meyers FJ, et al. "Teaching as a competency": competencies for medical educators. *Acad Med.* 2011;86(10):1211-1220.
- 23. Cooke M, Irby DM, Debas HT. The UCSF Academy of Medical Educators. Acad Med. 2003;78(7):666-672.
- 24. Youm J, Wiechmann W. Medical student use of the iPad in the clerkship curriculum. *Clin Teach*. 2015;12(6):378-383.

Chapter 4: Diversity, equity and inclusion

Mario Fahed, MD; Alicia Gonzalez-Flores, MD; Donna Jackson, EdD; Rosa Lee, MD; Lisa Meeks, PhD, MA; Tonya Fancher, MD, MPH; and David Henderson, MD

"No one is free until we are all free."—Rev. Dr. Martin Luther King, Jr.

Chapter Summary

Among educators from historically excluded and marginalized groups, the experience of well-being is often mediated through the lens of identity. Given the intersectional nature of identity, well-being becomes vulnerable along many axes of marginalization and exclusion. These axes include gender, race/ethnicity, sexual orientation, language, income, and physical ability and body habitus, to name a few. Narratives of lived experience provide a means of making explicit the structural and personally-mediated inequities that threaten the well-being of historically excluded and marginalized educators. Academic medicine environments will continue to threaten the well-being of these educators until institutions make conscious efforts to dismantle oppressive systems and more individuals are engaged in intentional anti-ableist, anti-sexist, anti-homophobic and anti-racist work. Ameliorating structural and personally-mediated mechanisms of oppression and alienation are essential to improving the well-being of historically excluded and marginalized educators. In this chapter, we comprehensively explain diversity, equity and inclusion (DEI) in the context of well-being in medical education. Additionally, we have included our personal narratives to provide insight into our lived experiences in the medical education realm. Last but certainly not least, we address the importance of intentionally incorporating DEI efforts into academic medical centers to promote systemic and sustainable well-being for all educators. The proposed competencies are the lessons that can be drawn from this chapter to help increase the well-being of historically excluded and marginalized educators.

Defining Diversity

Culturally, people define themselves by characteristics that set them apart from some people and connect them to others, producing narrowly defined norms that are purposely highly exclusive and hierarchical. Being a member of a historically excluded and marginalized community in academic environments that reflect the dominant white-centric social culture sets one outside the norm. The path to engagement, well-being and success is ripe with exclusion. This path reflects longstanding prejudices characterized as "isms" (e.g., racism, sexism, ableism and heterosexism). These "isms" represent structural discrimination that is deeply embedded in medical education through policies, practices, opportunities and assessments. Societally, such discrimination is bolstered by laws, written or unwritten policies and belief systems that perpetuate widespread unfair treatment of individuals based solely on their identity within a group.

Experiences that negatively impact the well-being of marginalized and minoritized educators are as ubiquitous as air and often equally invisible to everyone except these educators themselves. Lack of inclusion in the workplace is associated with poor mental well-being, physical health and retention.¹ As a result, historically excluded and marginalized educators are at risk of poorer well-being than the general population, with higher rates of suicide and depression and, gravely, less help-seeking behavior.^{2,3}

What is Well-Being?

Of the elements in the triad of disease, illness and well-being, the latter is often the most difficult to delineate. The health system has specific classifications for pathologic processes identified by constellations of history, physical findings, diagnostic imaging and laboratory results to define disease. Illness then is the individual experience of disease. Well-being, however, presents a unique challenge. It is defined by social and cultural norms. Depending on an individual's relationship with the dominant culture, the norm might not always be normal in one's cultural frame of reference. As an example, aging is a normal process; however, American culture highly prizes youth. At a cultural and social level, the norm of youth does not align with the normal fact of aging. As people age, they may feel stigmatized and develop a negative image (and experience) of themselves and their state of well-being related to the passage of youth. This is analogous to the experience of historically excluded marginalized educators in medical education environments that reflect the dominant social culture.

Workplace mistreatment and organizational culture are linked to much of the distress experienced by historically excluded and marginalized educators. Unfortunately, workplace mistreatment is more common for non-majority groups, including racial and ethnic discrimination, sexual and gender harassment, and mistreatment due to disability status.²⁻⁸ Given these associations, it is clear that educators and physicians who identify as members of the non-majority face stress and discriminatory experiences that threaten their overall well-being.

The Critical Nature of Language and Agency

Assigning language or terminology to historically excluded and marginalized communities is one way in which the majority population attempts to course correct and encourage respectful communication. However, in doing so, agency over a person's identity is often taken away, again, at the hands of the dominant group. Examples include the use of the term *disabled* as a way of defining oneself or one's community. This term is excluded in academic medical journals in favor of *person with a disability*, even though many individuals with disabilities identify as—and proudly refer to themselves as—disabled. Additionally, some language is used to describe an entire community, such as minoritized, but it also speaks to the experiences of the individuals it references. Some may hold one or more diverse identities, but not identify with specific experiences and, thus, not identify with the terms used to describe their community or position in society. For these reasons, authors of the following narratives have been given freedom in their use of language. In doing so, each contributing author can express both their identity as they desire and describe their experiences using language that feels authentic. We encourage the reader to ask when unsure how to refer to a group or population in everyday communications, advocacy or academic work.



"When they approach me, they see only my surroundings, themselves or figments of their imagination, indeed, everything and anything except me."

Ralph Ellison, *Invisible Man*, 1952

A paradoxical experience confronts historically underrepresented educators whose identity does not conform to the dominant identity of an institution: one of simultaneous invisibility and hypervisibility. Settles et al. describe

issues of invisibility for historically underrepresented educators as manifesting predominantly through social and professional exclusion as well as lack of recognition for their scholarship and achievement by colleagues, whereas historically underrepresented educators also experience hypervisibility through tokenism and feelings of being used by one's institution to represent diversity. As such, a historically underrepresented educator's sense of visibility in the workplace is often experienced bi-directionally. They may feel invisible to others in the professional world because of the feelings of being "unseen and unheard" by their peers and feel invisible because they do not see their identity reflected in academic medicine. For educators who do not witness the representation of their experiences and identities in the dominant narratives of their institutional culture, the experience of invisibility can be alienating and professionally stunting. Invisibility can foment feelings of imposter syndrome. When confronted with the lack of diversity within leadership positions in academic medicine, historically underrepresented educators may feel they have limited opportunities for professional growth. Additionally, the impact of invisibility on the well-being of underrepresented faculty members can be profound, especially when pain and trauma caused by experiences and events that attack their identities go unrecognized and unaddressed by their institution.

As an Asian American and Pacific Islander (AAPI)-identified educator, I was devastated by the targeted murders of eight people, six of whom were women of Asian descent, at Atlanta area spas in 2021. These murders shook me on a deeply personal level and left me feeling fragile and vulnerable during an already fraught time in our country. While the media covered the event and I experienced vocal outpourings of grief and outrage, particularly from the AAPI community, I recall initial silence within my own workplace. The lack of acknowledgement of the event by my colleagues and my institution left me deeply disappointed. I found myself wondering if my institution did not recognize the trauma that this event could elicit among AAPI students, faculty and staff. I feared we would be left to grieve in silence and continue with our work without acknowledging our trauma.

Under the gaze of hypervisibility, the historically underrepresented educator is seen and recognized by one's institution but only to serve the purpose of institutional diversity. Ironically, this diminishes the focus on the individual faculty member. Like invisibility, hypervisibility may be perpetuated by a bi-directional gaze; our institution acknowledges us precisely for our status as being underrepresented and subjects us to tokenism by which we are expected to represent the entirety of our community. Nevertheless, we, as historically underrepresented educators, may place burdens of infallibility upon ourselves as representatives of our community who bear the "minority tax" in efforts to advance the standing, not of ourselves, but as members of our community.

Invisibility and hypervisibility can be simultaneously experienced, yet they do not cancel each other out. Instead, both can co-exist as heavy burdens felt by historically underrepresented educators under which well-being suffers. What connects invisibility and hypervisibility is this: It is the experience of being denied the opportunity to be known and valued for our individual attributes, talents and potential. Seen or unseen, we are not truly known.



"... when markers of race, gender, gender fluidity, sexual orientation, ethnicity, religion and other factors are the only criteria considered in hiring or admissions, students are cheated, as are those chosen to meet diversity measures on the basis of identity alone. Nothing is more essentialist or constraining than diversity understood strictly in terms of identity."

Michael Rectenwald, Springtime for Snowflakes: Social Justice and Its Postmodern Parentage, 2018

Walking into the room, I am honored to have an opportunity to serve on this committee, yet I see no one else like me. Questions begin to swirl in my mind. Was I the right person? Will my voice be heard? Am I the only one? As an African American, female, non-physician assistant dean, my intersectionalities allow me to bring diversity to the table; however, in the room, I am often overshadowed. My experience is not unique. It is the experience of numerous others who find themselves being the lone representative of their gender, race, ethnicity and/or sexual orientation in a meeting or on a committee.

In a recent essay, Camille Garrison describes herself as the "Lonely Only," because of how often she finds herself in situations where she is the only person of a particular identity group in a room. Academician Amado Padilla refers to this phenomenon as the "cultural tax"—an obligation to show good citizenship toward the institution by serving its needs for ethnic representation on committees or to demonstrate knowledge and commitment to a cultural group. Such service may even bring accolades to the institution, but the institution does not usually reward it. This "taxation" is not new in academia. In today's academy, you can insert—gender, race, ethnicity, religious affiliation, sexual orientation—as groups affected by such "taxation."

One may ask, "How can this be detrimental to one's well-being?" Carrying a burden has always been detrimental. The burden is exacerbated when it requires a single person to carry the "weight" of an entire group by speaking as the sole representative of that group. This causes undue stress for the individual who, in particular environments, must choose comments more carefully than others. There is often uncertainty that, when one does speak, the comments will be taken seriously. In some cases, the sentiment is that one has been given a "seat" at the table but no voice in the conversation.

What is the solution? A possible solution is acknowledging that, even as members of specific communities, I come to the table with individual thoughts and perspectives. Ask me to take a "seat" for the value I bring without expecting to speak for an entire group. I don't expect others to speak with one voice. Faculty, who are not members of the majority culture, should be afforded that same consideration.



"How can you thank a man for giving you what's already yours? How then can you thank him for giving you only part of what's already yours? You haven't even made progress, if what's being given to you, you should have had already. That's not progress."

Malcolm X, The Ballot or the Bullet, 1964

When one grows up amid scarcity, being grateful to others for what is given to you is key to survival. When I finished residency and was offered a job, I never questioned it. I was grateful. When a myriad of opportunities

arrived shortly thereafter, no matter how big or small they were, particularly if they involved diversity work, I was grateful. Not knowing which ones to turn down or pursue, I took all of them, never thinking I had earned my place or could say no. When I was overstretched to the limit, doing it all, and despite my efforts, my advancement was stagnant, I did not question the system, the process and the structures that led to that stagnation. Instead, I questioned myself, my abilities and my belonging. My perceived failures were isolating and my overcommitment exhausting. Sadly, I was not the only one experiencing this. How does one travel the path of academic medicine without a guide? How does one picture oneself at the top of the mountain when the people at the top of the mountain look nothing like you? The answer is simple. One does not.

Mentoring has been well documented to positively impact career advancement, productivity, career satisfaction and retention. ¹² It is also well known that minority educators remain underrepresented at all levels of academic medicine and are less likely to have mentors as trainees and faculty. ¹³ This leads to getting lost and feeling overwhelmed, overstretched and ultimately, burned out. More attention has been recently given to the emotional protection that mentoring offers, particularly for historically underrepresented and women educators. All of whom are more likely to experience racism, direct discrimination and "ambient discrimination." ¹⁴ Having a mentor can validate one's identity and experiences and provide a safe space for reflection. However, the assumption that historically underrepresented educators and learners can only be mentored by other historically underrepresented educators further contributes to this disparity. It also further increases the minority tax and the pressure we put on ourselves to take on all the mentoring efforts of everyone who looks like us. There are many benefits to race and gender concordant mentoring and these benefits, particularly the emotional ones, cannot be minimized. However, while our numbers remain low, we cannot afford to be the only ones engaged in this effort.

Why don't we see ourselves at the top of the mountain? Or is it that those at the top do not see us as having the skills to belong there? As such, sponsorship of underrepresented in medicine faculty becomes even more important, recognizing talent, increasing our visibility and providing opportunities for leadership positions. All of these can have a positive impact, not only on career satisfaction and retention, but also on our sense of belonging and well-being.

8

"I believe that no one should ever have to choose between a career we love and living our lives with authenticity and integrity."

Selisse Berry, Out & Equal at Work: From Closet to Corner Office, 2013

There, in the corner of my desk, sat the pin—mocking me, judging me for being a hypocrite. The decision to wear the rainbow pin on my white coat signified more than allyship. I remember my hands shaking as I pressed the clutch down on the back of the pin, securing it tightly. While I was proficient in "talking the talk," I now prepared to "walk the walk." I can't authentically wear the pin just "hinting" at my sexual orientation. I can't tell my students to feel safe if I don't contribute to the safety of the space.

It wasn't a secret. My colleagues knew I was gay, a surface understanding of my identity. They didn't know how to place me in a box and whether I matched the prevailing trope. Was I a doctor merely wearing a pride pin, or did I go out every night to gay bars and still make it to rounds on time? Was I able to keep my emotions in check, or would I have exaggerated responses to events? Would I provide fashion advice every morning for my peers? (no.) Would I give them a "you go, girl." (no) I didn't want to carry the weight of their assumptions about what it meant to be gay, or worse, be tasked with correcting their assumptions. But I needed to do this because I had privilege and some power, and I needed to yield it correctly.

Today being gay feels like being born with a coat fused to my skin—a beautiful, colorful coat. It took me a long time to like its feel on my body. Now more than ever, I wouldn't take it off, even if I could.

This coat flows behind me as I walk down the halls on rounds. It signals to my students that I am safe; they can confide in me. Despite prevailing assumptions that accompany the coat, I've learned to gently remind people that every coat is different. These coats come in different shapes, in different colors, and with different personalities—no two coats are the same.

Loving the coat also meant that I must reckon with my own internalized homophobia, judging my reactions to make sure they're not "too gay," worrying about doors closing because I'd be deemed too gay to be professional. You see, in certain rooms, the coat feels too warm. In these rooms, I am asked to be a token, to fluff up the coat, show it off, show off how diverse we are here. In other rooms, I'm asked to remove the coat, for the coat is too colorful and too loud. I can't get too worked up or emotional. People will think I'm too much.

I observe that leaders don't wear these gorgeous coats. Their coats don't flow behind them on their way into their office. So, I stay up nights wondering if I could have handled something differently. Will I risk my opportunities for promotion? But the coat stays. It's gorgeous. And I'm done hiding.



"I really want to live in a world where disability is not the exception, but the norm." Stella Young

In part, people with disabilities are drawn to medicine because of their lived experiences as patients. This experience often results in greater empathy for their patients and a more holistic understanding of patients' daily lives, countering prevailing stereotypes. Unfortunately, physicians with disabilities enter a system where they will face increased mistreatment and deeply embedded ableist attitudes in the workplace.^{15,16}

For those who enter medicine with multiple marginalized identities, that burden is greater. Educators with disabilities face challenges related to their identity, such as the burden of infallibility and feeling as if they have to perform better than their non-disabled peers just to train and work in the same space. Given the low percentage of faculty with disabilities, this group is also less likely to have access to mentors or affiliate groups. Invisibility, hypervisibility and tokenism are frequently experienced. For example, it is not uncommon for someone with a visible disability to be asked to represent the entire spectrum of disability or for the inclusion of

a person with a physical disability to serve as "proof" of inclusion. Those with non-apparent disabilities are often excluded from inclusion efforts and events. Disability is highly heterogeneous. Even within the same category of disability, people can have highly varied experiences that are impacted by their history of disability and the support and privilege, or lack thereof. Being asked to represent an entire community, therefore, can place a significant burden on the faculty member with a disability, contributing to poorer well-being.

The deeply held ableist views that position educators with disabilities as less qualified or "lucky" to be part of the academy may also have long-term and damaging effects on the well-being of these educators. Indeed, research shows that physicians with disabilities are up to 17 times more likely to be mistreated than their non-disabled peers. These educators are constantly battling ableism while simultaneously working to showcase their abilities and value in a system that continuously positions disabled people as "less qualified." For these physicians, every day is a battle against skewed perceptions and assumptions that threaten well-being and keep educators with disabilities from thriving.

8

"There is a loneliness that can be rocked. Arms crossed, knees drawn up, holding, holding on, this motion, unlike a ship's, smooths and contains the rocker. It's an inside kind—wrapped tight like skin. Then there is the loneliness that roams. No rocking can hold it down. It is alive, on its own. A dry and spreading thing that makes the sound of one's own feet going seem to come from a far-off place."

Toni Morrison, Beloved, 1987

I was so grateful, intensely and endlessly so very grateful, to be here. To be chosen to join an American health care system is a privilege; to join a university that sponsored a visa is a loan awaiting repayment. My employer put up money to petition the United States government to sponsor me so I could leave circumstances that weren't so kind to me. I already feel the pressure to get them their "money's worth." I also feel the pressure to prove that my education lives up to the American standards I have looked up to my entire life. I am also the de facto representative of everyone who will follow me from my home country. I am the one who smiles and nods my head to every "I love your food!" response that follows questions about my origins. What was once a personal accomplishment in service of the greater good becomes a personal accomplishment in the name of everyone like me who will follow me down this path. This responsibility to not only keep the door open, but also to write the manual of how to navigate and succeed here is exhausting. My identity becomes fused with the success story everyone has decided I will have. Any stumble along the way is not only damaging to me but could also be damaging to the overall narrative about folks like me, those who escape intolerant conditions to find success in the land of the free.

Hand in hand with representation is often *the burden of infallibility*. I cannot fail; I cannot make mistakes. Aside from my duty to represent and live up to expectations, I also have a duty to myself not to squander this opportunity. You see, many like me, who identify with subcultures not welcome in the countries we are from, find academia to be a haven away from persecution. "If I can just make it out of here, I would surround myself with others who are educated and, as a result, accepting. I would be safe." Success then becomes intertwined with survival. Visas, sponsorships and funding for "aliens" and "non-residents" all hinge on job performance and

the continued success we bring to institutions. Every aspect of my existence could suffer if I'm not successful at work.

The problem is that failure can take many shapes and forms and sometimes be completely out of my control. Nevertheless, I make every effort to control every variable, however irrational that may be. I start to associate any deviation from a stellar perception of my work with catastrophe. In order to keep the status quo, to not rock the boat, I err on the side of accepting abuse if I have to because I have to keep my eyes on the prize. A biased patient telling me they want an American doctor? "No problem, just please don't raise your voice and make a scene." Someone from the administration showing clear signs of bias against me? Too bad, you can't go anywhere else when your sponsor is your employer. Green cards and visas also require an immaculate record. I won't risk a parking ticket, let alone a lawsuit from a patient who should be discharged from my practice.

The burden of infallibility takes over every decision I make every day. It's a constant effort to remind myself that I do not have to be flawless every time, to be just a little kinder to myself.



"I am an expression of the divine, just like a peach is, just like a fish is. I have a right to be this way...I can't apologize for that, nor can I change it, nor do I want to...We will never have to be other than who we are in order to be successful...We realize that we are as ourselves unlimited and our experiences valid. It is for the rest of the world to recognize this, if they choose."

Alice Walker, The Color Purple, 1982

Proposed Solutions

The narratives above are shared to provide insight into the daily, pedestrian experience of educators from historically excluded and marginalized groups, the daily struggles and traumas that impact not only job satisfaction and performance, but also fundamentally well-being. Suggestions for changes to improve the experience of historically excluded and marginalized educators often focus on individual and interpersonal interactions. It is true that such changes can be helpful, but more is needed, as they do not address pervasive structural issues.

In *A Gardener's Tale* Dr. Camara Jones presents a tripartite construct of racism that divides it into three domains: structural, personally mediated and internalized.¹⁹ This framework is also relevant to other forms of discrimination and inequity (e.g., sexism, ableism, homophobia and ageism). When we discuss issues related to equity, diversity and belonging, individual experience is often the focus, which results in dialogue centered on Dr. Jones' personally mediated and internalized domains. As an example, issues such as micro/macroaggressions and imposter syndrome are interrogated through the lens of personal and interpersonal experience. These perspectives are crucial to understanding the impact and consequences of inequitable environments, but such discussions leave an important part of the garden unexplored.

Imposter syndrome is most commonly viewed as a maladaptive response to identity and professional development challenges; however, such a perspective may stigmatize the very individuals needing support. It

has been rather convincingly argued, specifically with respect to women in the workplace, that the focus on individual experience stifles interrogation of the adverse environments that trigger imposter syndrome. An educator (or learner) experiencing imposter syndrome needs support, but that does not remedy the fundamental problem. The problem in need of a solution is the environment that engenders the reaction. Imposter syndrome is not a problem for women or individuals from historically excluded and marginalized communities to overcome, but indicative of deeply embedded structural inequities in work and learning environments that reflect the dominant social culture. Therefore, meaningful change to support the well-being of historically excluded and marginalized educators must occur at a structural level.

Take-Home Points

Structural competencies for medical schools and academic health centers

To assist institutions in becoming more equity-minded and making changes that foster the well-being and professional development of everyone in the work and learning environment, presented here are a set of structural competencies for medical schools and academic health centers. The four institutional competencies are structural awareness, equity-based practices, structurally-informed engagement and continuous structural reform. Specifically, institutions should be able to demonstrate the capacity for:

- Structural awareness—Develop strategies to identify, articulate and examine the ways in which
 socially-derived institutional structure, as manifested in policies, procedures and practices, informs
 climate and environment in ways that may disproportionately negatively impact the well-being of
 historically excluded and marginalized educators.
- Equity-based practices—Develop strategies to dismantle adverse institutional structures to support
 the well-being and professional development of historically excluded and marginalized educators
 in academic medicine.
- Structurally-informed engagement—Engage communities and other external stakeholders in a manner that deconstructs hierarchies of power and promotes community-driven collaboration in pursuit of addressing the structural roots of inequity.
- Continuous structural reform—Develop strategies for ongoing continuous quality improvement to monitor climate and culture and pursue equity at all levels of internal and external engagement.

The four competencies were developed from the educational rubric of structural competency.²¹ These four competencies are offered as a broad framework for action at an institutional level and action to dismantle elements of structural inequities and discrimination that are embedded in academic medical environments. They are offered as a starting point for discussions that should lead to the development of sub-competencies specific to individual institutional settings and result in direct positive action that is sustainable over time.

At the individual level, allyship offers a framework for collaborators to take responsibility intentionally and actively for change. Anne Bishop, *On Becoming an Ally*, ²² describes allies as "people who recognize the unearned privilege they receive from society's patterns of injustice and take responsibility for changing these patterns." An ally walks hand in hand with (and not for) historically excluded and marginalized educators and actively fights injustice to promote equity in the workplace through supportive personal relationships, sponsorship and advocacy. Before joining in that fight, an ally must prepare for it by self-reflecting on their privilege, educating themselves, and not expecting historically excluded and marginalized educators to carry the burden of educating them. In *Unlocking the Power of Allyship*, ²³ the NEJM Catalyst presents an allyship toolbox, and it includes the following tools:

- 1. Read, research and educate yourself
- 2. Self-reflect
- 3. Confront your own biases
- 4. Speak up for others
- 5. Lift up others by advocating

- 6. Hold others accountable
- 7. Respond with empathy
- 8. Have open dialogues

The first three tools require self-work and preparation. This is not to say that allies should feel like they must master these before being able to join in the fight. This misconception can lead to allies getting stuck in this preparation phase and thus never moving forward with the active steps of change-making. As described in *Bound Together*,²⁴ by Ellis, the practice of medicine is much like the practice of allyship. It requires a lifetime of learning, humility, the expectation of imperfection and persistence in the pursuit of healing, and most importantly, being comfortable with being uncomfortable.

Isolation is a common theme in the narratives above. The 2019 Association of American Medical Colleges data reveals that 5.5% of medical educators identified as Hispanic, 3.6% as African American, and 0.2% as American Indian or Alaska Native.²⁵ Numeric assessments do not tell the entirety of any story, which is why narratives are included in this chapter. However, such low levels of representation do provide a quantitative perspective that aligns with the personal experience of isolation. Therefore, one major imperative for all institutions should be to increase the level of diversity among their educators, an imperative that should be pursued with all due haste.

References

- 1. Kearney MD, Barg FK, Alexis D, Higginbotham E, Aysola J. Employee health and wellness outcomes associated with perceived discrimination in academic medicine: A qualitative analysis. *JAMA Netw Open.* 2022;5(1):e2145243. doi:10.1001/jamanetworkopen.2021.45243.
- 2. Shanafelt TD, Balch CM, Dyrbye L, et al. Special report: Suicidal ideation among American surgeons. *Arch Surg.* 2011;146(1):54–62. doi:10.1001/archsurg.2010.292.
- 3. Filut A, Alvarez M, Carnes M. Discrimination toward physicians of color: A systematic review. *J Natl Med Assoc.* 2020;112(2): 10.1016/j.jnma.2020.02.008.
- 4. Boyd KM. Disease, illness, sickness, health, healing and wholeness: exploring some elusive concepts *Med Humanit*. 2000;26:9-17.
- 5. Coombs AAT, King RK. Workplace discrimination: experiences of practicing physicians. *J Natl Med Assoc.* 2005;97(4): e15868767.
- 6. Duba A, Messiaen M, Boulangeat C, et al. Sexual-orientation based discrimination is associated with anxiety and depression in young physicians. A national study. *J Affect Disord*. 2020 Sep;274:964–8.
- 7. Jackson TN, Pearcy CP, Khorgami Z, Agrawal V, Taubman KE, Truitt MS. The physician attrition crisis: A cross-sectional survey of the risk factors for reduced job satisfaction among US surgeons. *World J Surg*. 2018;42(5): 10.1007/s00268-017-4286-y.
- 8. Meeks LM, Conrad S, Nouri Z, Dill M, Moreland C. Experiences of mistreatment among physicians with disabilities. *Health affairs*. In Press.
- Settles IH, Buchanan NT, Dotson K. Scrutinized but not recognized: (In)visibility and hypervisibility experiences of faculty of color. *J Vocat Behav*. Volume 113, August 2019, Pages 62-74. https://doi.org/10.1016/j.jvb.2018.06.003.
- 10. Garrison C. "The lonely only: Physician reflection on race, bias, and residency program leadership." In *Character and Caring: A Pandemic Year in Medical Education*, edited by Kalet A, Campbell B, Peltier W, Fletcher, K, Schmitt J, Weileder E, Keane R. Waukesha, WI: Ten16 Press; 2021.
- 11. Padilla AM. Ethnic minority scholars, research, and mentoring: Current and future issues. *Educ Res.* 1994. 23(4), 24–27. https://doi.org/10.2307/1176259.
- 12. Choi AMK, Moon JE, Steinecke A, Prescott JE. Developing a culture of mentorship to strengthen academic medical centers. *Acad Med.* 2019;94(5):630-633. doi:10.1097/ACM.0000000000002498.
- 13. Ramanan RA, Taylor WC, Davis RB, Phillips RS. Mentoring matters. Mentoring and career preparation in internal medicine residency training. *J Gen Intern Med.* 2006;21(4):340-345. Doi:10.1111/j.1525-1497.2006.00346.x.
- 14. South-Paul JE, Campbell KM, Poll-Hunter N, Murrell AJ. Mentoring as a buffer for the syndemic impact of racism and COVID-19 among diverse faculty within academic medicine. *Int J Environ Res Public Health*. 2021;18(9):4921. Published 2021 May 5. doi:10.3390/ijerph18094921.
- 15. Church PT. A personal perspective on disability: Between the words. *JAMA Pediatr.* 2017;171(10):939. doi:10.1001/jamapediatrics.2017.2242.
- 16. Herzer KR. Moving from disability to possibility. JAMA. 2016;316(17):1767–1768. doi:10.1001/jama.2016.9956.
- 17. Nouri Z, Dill MJ, Conrad SS, Moreland CJ, Meeks LM. Estimated prevalence of US physicians with disabilities. *JAMA network open*. 2021 Mar 1;4(3):e211254.

- 18. Meeks LM, Conrad SS, Nouri Z, Moreland CJ, Hu X, Dill MJ. Patient and coworker mistreatment of physicians with disabilities: Study examines mistreatment of physicians with disabilities. *Health Affairs*. 2022 Oct 1;41(10):1396-402.
- 19. Jones CP. Levels of racism: a theoretic framework and a gardener's tale. *Am J Public Health*. 2000;90(8):1212-1215. doi:10.2105/ajph.90.8.1212.
- 20. Tulshyan R, Burey J. Stop telling women they have imposter syndrome. *Harv Bus Rev.* February 21, 2021. https://hbr.org/2021/02/stop-telling-women-they-have-imposter-syndrome. Accessed December 9, 2022.
- 21. Metzl JM, Hansen H. Structural competency: theorizing a new medical engagement with stigma and inequality. *Soc Sci Med.* 2014;103:126-133. doi:10.1016/j.socscimed.2013.06.032.
- 22. Bishop A. *Becoming an Ally: Breaking the Cycle of Oppression in People.* Third Edition. Black Point NS: Fernwood; 2015.
- 23. Noone D. Unlocking the power of allyship: Giving health care workers the tools to take action against inequities and racism. *NEJM Catal Innov Care Deliv*. May 2022.
- 24. Ellis D. Bound together: Allyship in the art of medicine. *Ann Surg*. August 2021 Volume 274 Issue 2 p e187-e188. doi: 10.1097/SLA.00000000000004888.
- 25. Diversity in medicine: facts and figures 2019. Association of American Medical Colleges: Washington, DC, 2019. https://www.aamc.org/data-reports/workforce/report/diversity-medicine-facts-and-figures-2019. Accessed December 9, 2022.

Chapter 5: Intergenerational Cultures within Medical Education

Catherine L. Coe, MD; Winston Li, MD; and Julie A. Kerry, PhD

Chapter Summary

Educator well-being is a concern across the entire spectrum of medical education.¹ Studies show that levels of educator burnout are highest among clinicians at the early stages of their careers and in women and individuals from groups historically excluded from medicine. While at slightly lower levels, burnout is also a significant factor for non-clinician educators. Increasingly rapid changes within the health care and education systems have led to increased educator workload and are key drivers of burnout. Differences in successive generations and the changing demographics of graduating learners lead to differing expectations from trainees, which can also contribute to the increase in educator workload.² In this chapter, we describe some of the key interpersonal differences that impact the well-being of medical educators. These factors include differences in the mental models between educators and learners, with particular focus on the differing perspectives regarding the influence educators have over the learning environment; the decline in professional autonomy with an increase in academic isolation; and insufficient resources for educators to manage changing expectations in the learning environment. We also provide potential solutions, including rewarding educational efforts, systemic transparency, mentorship and development of educator communities. Such efforts can provide educators with an enhanced sense of meaning in their roles which contribute positively to well-being.³

Differing Mental Models Between Educators and Learners

Throughout the medical education continuum—undergraduate (UME) and graduate medical education (GME)—trainees and educators naturally have different perspectives that can, at times, lead to frustration and misaligned expectations. Possibly fueled by intergenerational differences between the two groups and an incomplete understanding of the pressures impacting both, this misalignment, or lack of a shared mental model, contributes to a negative impact on well-being for all.⁴

First, educators and trainees differ in the way they view and understand the complicated, interwoven hierarchy of the academic medical centers in which they exist. To the entering medical student or resident physician, a course, clerkship or program director might appear to possess unlimited authority or influence over the curriculum and system pressures. In reality, educators with these roles report to higher leadership positions within the institution, such as department chairs, curriculum committees and medical school deans. Chairs and deans subsequently answer to governmental mandates, system rules, hospital policies and university boards of governors. Trainees who are unaware of the larger system of governance can easily place idealistic expectations (and develop subsequent disappointment) regarding how much change an individual educator can affect.

For example, program, clerkship and course directors are tasked with conveying prescribed educational material to medical students in an approachable, learner-centered way. Trainees often assume that medical educators have autonomous control of content, delivery and assessments, but educators in these roles may have limits on their ability to innovate within the curricular structure. There may be restrictions on the use of novel learning modalities such as gamification⁵ and virtual reality⁶ or exploring alternate forms of assessment like open-ended questions that can better measure conceptual understanding.⁶ This can result in frustration among both medical educators and trainees.

Additionally, curricular requirements driven by national board examinations and credentialing bodies often feel at odds with an educator's goal to facilitate the development of well-rounded, patient-centered physicians. For instance, the major focus for students in the UME preclerkship phase is the United States Medical Licensing Examination® (USMLE®) and the Comprehensive Osteopathic Medical Licensing Examination of the United States (COMLEX-USA) benchmark exams, perceived as critical gateways to residency programs. As discussed in Chapter 3, this exam-centric focus of students leads to pressure for educators to "teach to the test," and some trainees perceive material not tested on such assessments as superfluous to their learning, even though it might be critical knowledge in the clinic and for future physicians. The focus on these benchmark assessments also increases the financial burden on trainees as they turn increasingly to costly external resources such as test prep books and software. This debt burden is carried into residency and often beyond, which can lead to them to leave academic medicine for private practice, putting more pressure on those educators who remain. This inherent tension can be exacerbated if trainees subsequently evaluate courses or clerkships lower because they perceive that educators include too much non-test content. Such negative evaluations may impact educators' promotion and tenure within an institution and exacerbate risk factors associated with leaving academic medicine.

On the GME side, residents and program directors are responsible for logging specific clinical experiences and professional development milestones to meet the standards for accrediting bodies.^{8,12,13} Residents work long hours and care for critically ill patients while balancing their personal lives. They become frustrated by administrative tasks such as logging duty hours and documenting patient cases and may resist completing them, creating friction between program and residents. The root of frustration among both educators and trainees is feeling powerless in the prescribed system to effect change.

Secondly, educators and learners are usually at different points in their professional identity development. The development and maintenance of a professional identity occurs on a continuum and is constantly molded throughout a career, training and practice. Learners grapple with the integration of personal life into their developing professional identity while experiencing the pressure of examinations, training applications and new financial burdens. Simultaneously, educators juggle curricular redesigns, clinical responsibilities, administrative responsibilities and their lives outside of work. 14,15 Educators also find themselves in the role of helping guide students as they manage stress and the integration of work and personal life, often without training about how to do so effectively without contributing to their own personal stress. At times, this comes at a cost to educators' well-being—educators with increased workload expectations and insufficient support are caught between generational differences on how personal lives intertwine with professional lives and the ever-changing expectations between the two. 16 Educators of an older generation, for instance, may see a sharper boundary between "work" and "personal" lives and maintain these as two separate spheres. In contrast, trainees of a younger generation may treat this distinction less rigidly, such as texting a professor after business hours or bringing up personal matters in an advising meeting. Improved alignment of learner and educator expectations regarding professional identity formation (i.e., developing a shared mental model through either mentoring experience or faculty development) may enhance educator well-being and decrease turnover within academic medicine.17

Finally, differing attitudes and skills regarding various educational modalities and the use of technology can heighten the disconnect between educators and learners and put an increased burden on educators accustomed to traditional educational modalities. Trainees entering medical school are increasingly of the Generation Z cohort, usually defined as those born between 1996 and 2010, while those entering residency and moving into educator positions are typically millennials, which is usually defined as those born between 1981 and 1996. While any generational differences are subject to broad generalizations, conventionally, millennials are defined as having high expectations for the use of technology, are accustomed to receiving feedback and are more empathetic than previous generations.¹⁸ They tend to value more structured learning environments, so they may need more encouragement to develop critical thinking skills. ¹⁵ Generation Z individuals are also technologically savvy and seek out blended learning experiences while at the same time exhibiting heightened levels of psychological stress.¹⁹ Educators may struggle to integrate more technology into their teaching, while creating a safe environment to facilitate learner development. How can educators adapt to this younger generation of learners? We propose that educators approach teaching as lifelong learners, constantly learning about new technologies and generational trends, and that leadership and the systems in which educators work provide explicit support for these efforts. Our case study in this chapter highlights the art of adapting to honor the learners' needs (and the evolving circumstances of a pandemic).

Decreased Professional Autonomy

The roles, responsibilities and expectations of educators have undergone enormous changes in recent decades. Many of these changes have led to a sense of loss of control and diminished professional autonomy for both clinical and non-clinical educators, despite research demonstrating that an enhanced sense of professional autonomy is one factor that can effectively ameliorate educator burnout.³ Improving educators' sense of professional autonomy is increasingly difficult to achieve. Within the context of the current health care and medical education systems, there are limitations as to direct methods by which individual educators can enhance their sense of autonomy. In Table 5.1, we identify several external forces that impinge on the professional autonomy of educators.

Table 5.1: External forces that impinge the autonomy of educators

Systemic Pressures	Educational Pressures
Health care system consolidation	Duty hour restrictions
Revenue-based reimbursement models	Importance of USMLE® and COMLEX-USA
	assessments
Insurance demands	Pedagogy changes (active learning, online learning)
Electronic health records	
Accreditation requirements	Remediation
Increasing enrollment at medical schools	Formalized feedback for trainees
Increasing debt burden of trainees	Move to pass-fail assessments
Increased competitiveness for research funding	Competency-based education and assessment
24-7 culture of email and rapid response	Curricular reform

The combined outcome of these changes is increased workloads, with educators struggling to balance multiple roles including clinical activities, research, education and administration, in addition to their personal lives. The particular burden of balancing traditional gender roles with professional expectations may be one contributor to the higher levels of burnout reported in women. In addition, the increased workload and decline in well-being may increase attrition of individuals who have been historically excluded from medicine. ¹⁰ As referenced in Chapter 1, too often education is deprioritized by institutions in favor of revenue producing activities on the part of departments, deans, medical groups and hospital systems. Electronic health records continue to be timeconsuming, reducing the decision-making capacity of physicians and negatively impacting the quality of patient care. 12 Duty hour restrictions, initially established to promote the well-being of residents, have not achieved the desired results for residents and have correspondingly increased the workload for attending physicians.⁸ Research funding continues to be difficult to obtain, leading to near equivalent burnout rates for non-clinician educators. We feel that this factor is likely to be exacerbated in the near future due to changes in research priorities related to the pandemic, the increasing costs of research due to inflation and supply chain issues. Lastly, educators now receive emails and requests from leadership and students at all times of day and on weekends. While some institutions/leaders explicitly state that they do not expect a response in the "off-hours," there remains a temptation to constantly check and respond to work email.²⁰

With regard to trainees, they generally expect a learner-centered education (as opposed to educator-centered learning) with active learning in the classroom and increased reliance on electronic curriculum delivery for didactic materials. While these changes provide an increase in student autonomy for learning, they conversely further impinge on educators' workload due to the need to acquire new teaching skills and master the software necessary to create high quality online educational materials and experiences. Specific training in how to teach and design curriculum and assessment is lacking for most clinical and non-clinical educators, exacerbating the workload impacts due to the time and cognitive load needed for development in these areas. The increasing pace of advancement in medical knowledge has led to the development of new models for medical education, including the Master Adaptive Learner model, which requires new approaches to teaching such as the Adaptive Educator.²¹ One way to develop the master adaptive learner is to implement coaching programs, which require specific training and certification for educators. While coaching can lead to higher job satisfaction levels, educators also report higher levels of burnout, possibly due to the increased emotional investment in the trainees.²² Finally, the move to competency-based medical education (CBME) requires modifications in the ways that the medical education system assesses individual trainees.²³ While this is certainly an opportunity for medical educators to innovate, it also requires additional time.

Insufficient Opportunities for Collaboration/Academic Isolation

An additional source of burnout for educators includes insufficient opportunities for collaboration and academic isolation. The current research funding climate for both clinical and non-clinical educators exacerbates burnout. Decreased funding paired with continued linkage of promotion decisions to traditional metrics such as peer-reviewed scholarship and acquisition of grant monies can add stress, particularly with continued expectations for outstanding clinical care and teaching responsibilities. This climate promotes isolation instead of collaboration. Discontent in work-life balance, teaching recognition, clinical demands, lack of educator development and lack of support from governance can be used to predict an educator's intent to leave academic medicine. Educator turnover not only impacts the delivery of medical education, but is also costly to institutions, with estimates of

the cost to recruit and retain a replacement ranging from 32% to 114% of a medical educator's annual salary.⁹ Attrition risk is highest among women, those from communities historically excluded from medicine and those with higher debt burden.¹⁰

Non-clinician educators have a unique role within medical education, and increasingly schools are recruiting specialist medical educators into basic science departments to replace roles previously occupied by those who were both researchers and educators. As medical education and health care delivery rapidly evolve, the role of non-clinician educators has shifted.²⁴ Basic scientists as educators have an increasing responsibility to underscore the clinical relevance of their educational materials, guide students in the critical analysis of scientific evidence and work with clinical educators to improve health care quality. This transition undoubtedly impacts the professional identity of basic science educators and their role in research and medical education. Despite the important roles that basic scientists play in medical education, however, there is a distinct lack of current research as to how these changes impact the well-being of these educators. Moreover, the fragmented culture of academia, which still emphasizes discipline-based development of scientific expertise and the divide between the basic and clinical sciences, further contributes to burnout and a reduced capacity to implement change to address academic isolation.

The SARS-CoV 2 (COVID-19) pandemic exacerbated not only academic isolation, but also isolation from other activities that typically promote collaboration. National conferences and meetings, where happenstance collaboration occurred, transitioned to virtual platforms. Although forward progress and collaboration still happened, the organic nature and development of new collaborative projects decreased. Simultaneously, educators responsible for the educational curriculum were called to action at the beginning of the pandemic. In a short period of time, they were asked to convert the entire curriculum to a virtual platform and think creatively in the delivery of the material, with some clinical educators also being called to service in the clinical space.²⁵ This double-edged sword both invigorated educators, working together against a common enemy, and contributed to fatigue, stress and burnout.

Insufficient Educator Training

As noted previously, educators have witnessed a transformation in the teaching environment, available technology and student expectations over the years. For veteran educators, the spread of high-speed internet access and online learning resources dramatically increased the ways in which learners could acquire information. Whereas learners of prior generations gained knowledge primarily in the lecture hall or by making in-person visits to the library, today's trainees have access to a wide body of knowledge at the click of a mouse or the tap of a phone screen. Both veteran and more junior educators have seen the trend of more active learning, technology and multimedia in the classroom.²⁶ There has been a shift toward more autonomous and independent learning for trainees, including the use of flipped-classroom models, team-based learning activities and discussion groups.²⁷

With this transformation, educators are asked to utilize novel methods to teach that they themselves may not have experienced as learners. Some educators have honed their craft for years and decades in the traditional lecture format, only to find new directives from their medical schools to incorporate more active learning or to notice lecture attendance steadily dropping as students choose more self-directed methods of study. As

educators work to adapt to this new learning climate, they may find limited resources and formal programming to aid in this transition. While educators receive extensive training in fields of clinical care and research endeavors, there is relatively little requirement for formal training in education prior to becoming an educator.

Workable Solutions

In the previous sections, we outlined how interpersonal and intergenerational differences can impact educator-learner dynamics and the learning environment and our perception of professional autonomy leading to a decline in well-being. Concurrently, the increase in educators' sense of isolation and perceived lack of availability of training can exacerbate this issue. In this section, we propose some solutions to these concerns. Ultimately, for educators who are dedicated to teaching, finding value and meaning in that role can have a positive impact on well-being.¹³

Alignment

A possible approach to aligning the mental models of educators and learners within the learning environment is to foster transparency between the two groups. An in-depth understanding of what each is wrestling with, in other words a shared mental model, can promote understanding and identify common goals. When appropriate, educators can share with trainees the department mandates, system pressures and institutional policies that guide, and at times, restrict, their behavior. The ability to explain the rationale behind decisions and rules (e.g., promoting patient care, helping the billing and operations of the hospital, respecting laws and legal customs, and honoring health care system priorities) can be meaningful in promoting understanding and buy-in. Some decisions and rules make sense at face value and do not need extended explanations; however, for more opaque policies, a word of explanation can go far in promoting adherence and garnering buy-in to processes. As we outline in our case study, clear communication through town halls and emails contributed to an alignment of expectations for both educators and learners through the challenge of adapting to the pandemic. A shared understanding of these "rules of the game" can diminish distress about why things are the way they are. When trainees gain insight into this larger picture, they have a greater opportunity to participate in effective problem solving. Intentional student engagement in curriculum development further promotes mutual respect and collaboration within the learning environment.²⁸

Within the clinical setting, connections between learners and educators can be enhanced through strategies like huddling.²⁹ Additionally, continued pursuit of CBME will promote transparency, clarify expectations and may serve to align expectations between educators and trainees.³⁰ The presence of objective markers of accomplishment and goal behavior (i.e., the competencies) could help learners understand the intended outcome of their training and their progression in their development. For educators, the presence of competencies could add a more objective and consistent method for grading and evaluating learners and reduce the risk of bias in grading. Conversely, however, full integration of CBME does require additional educator and staff time to ensure the additional elements of directly observed encounters and increased feedback. Implementation also requires educator time for development around CBME and for students regarding expectations to ensure alignment between the two groups.³¹ When implementing CBME, it is imperative to continue to be mindful of the possibility of incorporating bias into evaluation and to mitigate it.³²

Promoting Autonomy

Addressing the lack of professional autonomy in medical education and its role in educators' well-being requires a multi-pronged approach. Educators must be integrated into key decision-making bodies within both the academic and health care systems. Administrative leadership can develop creative ways to reward innovative educational efforts such as awards, notes of appreciation, and recognition on department or school websites and newsletters. Educational excellence should be an integral component of promotion criteria for all educators, and institutions should utilize the advances in medical education pedagogy to design novel ways for educators to participate in development activities to enhance their skills in instruction and assessment. As noted in Chapter 1, a broader cultural change at the institutional level will be required to provide educators with sufficient time for their educational activities in the face of these barriers. For example, education value units (EVUs) can be used as a framework for measuring the value of educators' contributions toward education.³³ EVUs can be readily adapted to account for the time required for modern pedagogical approaches such as small group, teambased and other active learning strategies, mentoring, coaching, and competency assessment approaches and can be applied across basic and clinical departments.

Collaboration

To reduce academic isolation and promote collaboration, institutions should pay close attention to promotion and tenure policies to reward educators. Many institutions have updated promotion and tenure guidelines to support educators; however, these changes vary by institution. Further, while medical schools may have adopted the changes, at times the broader institution remains set in traditional promotion schema.³⁴ Institutions should continue to look for ways to harmonize promotion and tenure policies and ensure equity. Research into the impact of ongoing changes within medical education on the well-being and professional identity formation for basic science and other non-clinical educators is also a critical need. For example, such research will help to define the place of the basic science researcher and educator within the future of medical education, will help facilitate better integration between the basic and clinical sciences, and provide clear pathways for promotion and tenure for full-time educators.

A mentor can also help to abate academic isolation. Although many mentorship programs exist within research, few programs are available for educators, specifically medical education researchers.³⁵ Access to a mentor can provide guidance navigating a career in academic medicine and avoiding unintentional pitfalls that delay promotion and tenure.³⁶ Mentors can also help identify areas for administrative support, how to obtain dedicated time for teaching and how to convert educational projects into scholarly products. Encouraging faculty to participate in mentoring through recognition by awards, formal programs and as part of promotion and tenure policies would likely lead to an enhanced perception of the value of medical educators as noted above. Lastly, building intentional communities of collaboration can mitigate isolation and burnout.³⁷ Formal organization through avenues such as an academy of educators and the office of medical education can bring educators together. Informal communities also promote collaboration and resilience in medical education.

Targeted Training

We present several methods and resources toward addressing insufficient training to develop one's craft as an educator, including concrete skill learning and connecting with a larger educator community.

First, educators need not figure everything out on their own. There are often resources, programs and formal training opportunities already in existence at most medical schools. Many medical schools have an academy of educators, or some similarly named organization, that serves to honor excellence in teaching, build a sense of community among educators and host development lectures and workshops.³⁸ These sessions can be great forums to learn about the latest educational trends and developments as well as discover novel teaching methods and curricula. Naturally, these academies attract individuals who are passionate about undergraduate and/or graduate medical education, making them fertile grounds for building professional connections and finding like-minded peers. Such academies may also provide an inclusive environment and common purpose to help address the isolation of those from communities historically excluded from medicine.

This theme of building connection can be a meaningful tool to fight against burnout and to navigate professional challenges. Often, educators perceive that they work in silos, that the problems they face are unique and that any solution has been too newly invented and developed in isolation. When educators connect with peers in similar roles, they often find that problems are shared, or at least have common features, and that what worked in one department can often inspire change in another. Colleagues with disparate courses, departments and target trainee audiences can find common ground in challenges such as adapting active learning into the medical student classroom and engaging resident physicians on a busy inpatient service. Indeed, the use of peer support as a development tool has been associated with reduced sense of isolation, increased commitment to teaching and enhanced teaching skills.³⁹ Trainees themselves are often overlooked resources for curriculum development and integration of technology within curricular materials.²⁸

Finally, in addition to school-wide programs like an academy of educators, one can often look close to home for mentors and coaches in education. In any given department or medical school curriculum, there are designated leaders, course directors, residency program directors and clerkship directors who would be reasonable contacts for a new educator. As detailed in the case study, an academy of educators can serve as a go-to resource during increased time of need and can mobilize educators to accelerate/upskill as needed. Traditional mentoring relationships can result from these contacts, where the mentor provides the mentee with knowledge and directs them to opportunities. More recently, a more novel paradigm of coaching has been discussed, where the coach takes on a less directive role (compared to mentoring), but instead helps the coachee gain insight into their own goals, needs and priorities and assists them in reaching certain performance targets.⁴⁰ Coaching honors the idea that educators are autonomous adult learners who have a wealth of prior experience and focuses on problems directly relevant to the coachee's daily life.⁴¹ Coaching promotes each coachee's independence and self-motivated desire to improve as a teacher.

In summary, systems and individuals can promote educator well-being by finding connection, seeking coaching and mentorship, promoting good communication and transparency, and fostering autonomy.



Case Study

Situation

The COVID-19 pandemic struck suddenly in the spring of 2020 impacting medical education and a large residency program, which had heretofore utilized traditional, in-person lectures, recognized that significant changes were necessary and desired in response to the changing world situation. Hospital and school of medicine policies limited in-person gatherings. Physical distancing became widely adopted to diminish the spread of infection.

This new reality compelled a radical transformation of the educational environment. Educators had to learn new virtual platforms to teach, adapt to engaging learners on the computer and keep abreast of all the latest and best practices on using technology to teach. Matters of procedure like when and how to speak in the virtual classroom, turning on one's video camera, utilizing the chat function and creating breakout rooms were all novel experiences. In addition, educators had the challenge to keep their department and school leadership, as well as their learners, updated and coordinated on every change.

Educators had to honor different pressures and needs among learners, administrators and other educators. Each party had their own perspectives, preferences and needs, whether that was honoring department rules (administrators), finding the most convenient avenues to teach (educators) and balancing teaching with new clinical demands (educators and learners).

Approach

In response to the COVID-19 pandemic, this residency program took several steps to analyze its intergenerational culture and climate and to create an action plan that honored the needs of multiple parties. First, consistent, clear communication in the form of virtual town hall meetings and regular emails helped keep educators and learners updated on the latest rules and changes in the department.

Educators sprung to action to learn novel, innovative teaching methods to engage learners in a virtual world, learned the intricacies of virtual platforms and helped to foster proper etiquette and effective ways to communicate online. The school of medicine at large, including the institution's academy of educators, created a free, online depository of resources for teaching in a pandemic. These materials included advice on teaching in various settings (inpatient, outpatient), giving large group presentations and utilizing educational theory in instruction.

As the pandemic waxed and waned, the residency program remained flexible and adaptive. As COVID-19 cases declined and vaccinations increased, there was a gradual return to some in-person teaching. Due to the convenience and the lack of additional cost of the virtual platform, new opportunities to bring in outside, physically remote lecturers to the classroom became available. Some learners found that the virtual platform allowed them to attend sessions that would have been previously unavailable to them—for example, when they were doing an off-site rotation.

Through collaboration, effective communication and innovation, educators in this system were able to maintain the educational enterprise while also promoting their own well-being.

Lessons learned

- Communicating and sharing goals, policies and pressures among different stakeholders in an academic medical center can promote understanding and decrease misperceptions among groups.
- Using the local educational community and system resources can reduce the burden on individual educators and departments in managing education challenges.

Take-Home Points

- The interplay between educators and trainees and the systems in which they work impact well-being.
- Changes in the health care system, increasing use of electronic health records and the implementation of duty hour restrictions are just some of the key systemic changes that lead to frustrations between educators and learners.
- These systemic changes lead to misaligned expectations and the blurring of boundaries between personal lives and work and service obligations.
- Striving for connection and transparency, when appropriate, can mitigate the impact of differing perspectives on well-being.
- Rewarding educators for educational effort through educational value units and promotion and tenure can ameliorate frustrations regarding professional autonomy and academic isolation.
- Mentors and organizations, such as an academy of educators, provide opportunities for both a sense of belonging and professional development, all of which can contribute positively to educators' well-being.

Solutions

Alignment of Mental Models

• Align goals through transparency, communication and collaboration.

Promoting Autonomy

- Integrate medical educators into decision-making bodies.
- Reward educational efforts through awards, promotion, tenure and education value units.

Collaboration

- Create and implement policies to support educators and collaboration.
- Research into well-being and the professional identity of basic science and non-clinical educators.
- Create and support educator communities.

Targeted Training or Educator Support

- Create and implement faculty development and academy of educators.
- Facilitate collaboration with peers and trainees.
- Improve access to mentors and coaches.

References

- 1. Dandar VM, Grigsby RK, Bunton SA. Burnout among U.S. medical school faculty. Analysis in brief, American Association of Medical Colleges 2019; 19(1). https://www.aamc.org/data-reports/analysis-brief/report/burnout-among-us-medical-school-faculty. Accessed December 9, 2022.
- 2. Morris DB, Gruppuso PA, McGee HA, Murillo AL, Grover A, Adashi EY. Diversity of the national medical student body Four decades of inequity. *N Engl J Med.* 2021; 384:1661-1668. doi: 10.1056/NEJMsr2028487.
- 3. Shah DT, Williams VN, Thorndyke LE, et al. Restoring faculty vitality in academic medicine when burnout threatens. *Acad Med.* 2018; 93(7): 979–984. doi: 10.1097/ACM.0000000000002013.
- 4. Yoo J, Matos M, Bota M, Schrewe B, Armstrong L. Bye to burnout: intergenerational narratives break barriers. *Med Educ*. 2019;53(5):518. doi:10.1111/medu.13850.
- 5. Pottle J. Virtual reality and the transformation of medical education. *Future Healthc J.* 2019; 6(3):181-185 doi: 10.7861/fhj.2019-0036.
- 6. Hauer KE, Boscardin C, Brenner JM, van Schaik SM, Papp KK. Twelve tips for assessing medical knowledge with open-ended questions: Designing constructed response examinations in medical education. *Med Teach*. 2020; 42(8):880-885. doi: 10.1080/0142159X.2019.1629404.
- 7. Chen DR, Priest KC, Batten JN, Fragoso LE, Reinfeld BI, Laitman BM. Student perspectives on the "Step 1 Climate" in preclinical medical education. *Acad Med.* 2019 94(3):302-304. doi: 10.1097/ACM.000000000002565.
- 8. Wolpaw JT. It is time to prioritize education and well-being over workforce needs in residency training. *Acad Med.* 2019; 94(11): 1640-1642. doi: 10.1097/ACM.0000000000002949.
- 9. Collins SK, McKinnies RC, Matthews EP, Collins KS. A ministudy of employee turnover in US hospitals. *Health Care Manag (Frederick)*. 2015;34(1):23-27. doi:10.1097/HCM.00000000000038.
- 10. Jeffe DB, Yan Y, Andriole DA. Competing risks analysis of promotion and attrition in academic medicine: A national study of U.S. medical school graduates. *Acad Med.* 2019;94(2):227-236. doi:10.1097/ACM.0000000000002441.
- 11. Lowenstein SR, Fernandez G, Crane LA. Medical school faculty discontent: prevalence and predictors of intent to leave academic careers. *BMC Med Educ*. 2007;7:37. doi:10.1186/1472-6920-7-37.
- 12. Alami H, Lehoux P, Gagnon M-P, Fortin J-P, Fleet R, Ahmed MAA. Rethinking the electronic health record through the quadruple aim: time to align its value with the health system. *BMC Med Inform Decis Mak*. 2020; 20:32. doi: 10.1186/s12911-020-1048-9.
- 13. Brenner AM, Beresin EV, Coverdale JH, et al. Time to teach: Addressing the pressure on faculty time for education. *Acad. Psychiatry.* 2018; 42:5–10. doi: 10.1007/s40596-017-0851-9.
- 14. Regehr G. Kids these days: Reconsidering our conversations about millennial learners. *Med Educ.* 2020;54(1):10-12. doi:10.1111/medu.14002.
- 15. Roberts DH, Newman LR, Schwartzstein RM. Twelve tips for facilitating millennials' learning. *Med Teach*. 2012;34(4):274-278. doi:10.3109/0142159X.2011.613498
- 16. American Medical Association. Organizational Steps to Support Medical Educators. 2022. https://www.ama-assn.org/system/files/protecting-education-mission.pdf. Accessed December 9, 2022.
- 17. Kusner JJ, Chen JJ, Saldana F, Potter J. Aligning student-faculty mentorship expectations and needs to promote professional identity formation in undergraduate medical education. *J Med Educ Curric Dev.* 2022; 9. https://doi.org/10.1177/23821205221096307.

- 18. Borges NJ, Manuel RS, Elam CL, Jones BJ. Comparing millennial and Generation X medical students at one medical school. *Acad. Med.* 2006; 81(6): 571-576. doi: 10.1097/01.ACM.0000225222.38078.47.
- 19. Plochocki JH. Several ways Generation Z may shape the medical school landscape. *J Med Educ Curric Dev.* 2019; 6:1-4. doi: 10.1177/2382120519884325.
- 20. Armstrong MJ. Improving email strategies to target stress and productivity in clinical practice. *Neurol Clin Pract.* 2017; 7(6):512-517. doi: 10.1212/CPJ.000000000000395.
- 21. Auerbach L, Santen SA, Cutrer WB, Daniel M, Wilson-Defosse AL, Roberts NK. The educators' experience: Learning environments that support the master adaptive learner. *Med Teach*. 2020; 42(11): 1270-1274.
- 22. Elster MJ, O'Sullivan PS, Muller-Juge V, Sheu L, Kaiser SV, Hauer KE. Does being a coach benefit clinician-educators? A mixed methods study of faculty self-efficacy, job satisfaction and burnout. *Perspect Med Educ.* 2022; 11: 45-52.
- 23. Gruppen LD, ten Cate O, Lingard LA, Teunissen PW, Logan JR. Enhanced requirements for assessment in a competency-based, time-variable medical education system. *Acad Med.* 2018; 93(3): S17-S21.
- 24. Grande JP. Training of physicians for the twenty-first century: role of the basic sciences. *Med Teach*. 2009;31(9):802-806. doi:10.1080/01421590903137049.
- 25. Woolliscroft JO. Innovation in response to the COVID-19 pandemic crisis. *Acad Med.* 2020;95(8):1140-1142. doi:10.1097/ACM.000000000003402.
- 26. McCoy L, Lewis JH, Dalton D. Gamification and multimedia for medical education: A landscape review. *J Am Osteopath Assoc.* 2016;116(1):22-34. doi:10.7556/jaoa.2016.003.
- 27. Riddell J, Jhun P, Fung CC, et al. Does the flipped classroom improve learning in graduate medical education? *J Grad Med Educ*. 2017;9(4):491-496. doi:10.4300/JGME-D-16-00817.1.
- 28. Peters H, Zdravkovic M, Costa MJ, et al. Twelve tips for enhancing student engagement. *Med Teach*. 2019; 41(6):632-637. doi: 10.1080/0142159X.2018.1459530.
- 29. Elhassan M. The hospitalist huddle: A 1-year experience of teaching hospital medicine utilizing the concept of peer teaching in medical education. *Adv Med Educ Pract.* 2017; 8:785-789. doi: 10.2147/AMEP.S149450.
- 30. Desy JR, Reed DA, Wolanskyj AP. Milestones and millennials: A perfect pairing-competency-based medical education and the learning preferences of Generation Y. *Mayo Clin Proc.* 2017;92(2):243-250. doi:10.1016/j.mayocp.2016.10.026.
- 31. Griffiths J, Dalgarno N, Schultz K, Han H, van Melle E. Competency-based medical education implementation: Are we transforming the culture of assessment? *Med Teach*. 2019; 41(7): 811-818.
- 32. Dhaliwal G, Hauer KE. Excellence in medical training: developing talent-not sorting it. *Perspect Med Educ.* 2021;10(6):356-361. doi:10.1007/s40037-021-00678-5.
- 33. Bellini LM, Kaplan B, Fischel JE, Meltzer C, Peterson P, Sonnino RE. The definition of faculty must evolve: A call to action. *Acad. Med.* 2020; 95: 1515-1520. doi: 10.1097/ACM.000000000003158.
- 34. Ryan MS, Tucker C, DiazGranados D, Chandran L. How are clinician-educators evaluated for educational excellence? A survey of promotion and tenure committee members in the United States. *Med Teach*. 2019;41(8):927-933. doi:10.1080/0142159X.2019.1596237.
- 35. Beck Dallaghan GL, Coe CL, Wright ST, Jordan SG. Mentoring medical education research: Guidelines from a narrative review. *Med Sci Educ*. May 2022. doi:10.1007/s40670-022-01565-2.
- 36. Choi AMK, Moon JE, Steinecke A, Prescott JE. Developing a culture of mentorship to strengthen academic medical centers. *Acad Med.* 2019;94(5):630-633. doi:10.1097/ACM.0000000000002498.

- 37. Cantwell D, Kirkland S, Lebovits H, et al. No Rapunzel in this ivory tower: finding your collective and overcoming academic isolation. February 2022. doi:10.33774/apsa-2022-jmccs.
- 38. Searle NS, Thompson BM, Friedland JA, et al. The prevalence and practice of academies of medical educators: a survey of U.S. medical schools. *Acad Med.* 2010;85(1):48-56. doi:10.1097/ACM.0b013e3181c4846b.
- 39. Campbell N, Wozniak H, Philip RL, Damarell RA. Peer-supported faculty development and workplace teaching: an integrative review. *Med Educ*. 2019;53(10):978-988. doi:10.1111/medu.13896.
- 40. Deiorio NM, Carney PA, Kahl LE, Bonura EM, Juve AM. Coaching: a new model for academic and career achievement. *Med Educ Online*. 2016;21:33480. Published 2016 Dec 1. doi:10.3402/meo.v21.33480.
- 41. Orr CJ, Sonnadara RR. Coaching by design: exploring a new approach to faculty development in a competency-based medical education curriculum. *Adv Med Educ Pract*. 2019;10:229-244. Published 2019 May 1. doi:10.2147/AMEP.S191470.

Chapter 6: Measuring educator well-being

Andrea N. Leep Hunderfund, MD, MHPE; Yasmin Magdaleno, MD; and Elizabeth C. Lawrence, MD

Chapter Summary

Medical educators play an important role within health care organizations and are critical to recruiting, training and retaining the future health care workforce. Measuring the well-being of educators signals the importance of these professionals to the mission of a school or organization and enables system improvements informed by their experiences and perspectives. In this chapter, we discuss why measuring educator well-being is important, if and when to measure, who should lead the measurement effort and who will be invited to participate. We then review some of the most widely used assessment tools and discuss how to choose an instrument considering an institutions' needs. We conclude with a brief overview of what to do with the results and common pitfalls to avoid.

Why Measure?

Educators have high rates of burnout and work-related distress, as noted in earlier chapters. Burnout not only threatens educator well-being, but also risks adverse downstream consequences for the people, organizations and communities educators serve. Factors contributing to burnout vary based on role and work area, and educators have unique challenges and needs that may not be recognized without dedicated assessment. When done well, measurement can illuminate these challenges and help organizations direct and prioritize their improvement efforts.¹

Measuring educator well-being enables continuous improvement by revealing how a system is currently performing, uncovering variation across groups or sites and exploring how these variations relate to barriers and facilitators in the work environment. Each work unit has its own culture and needs, and organizations can use well-being data to understand these differences, identify best practices, provide feedback to local leaders and shed light on equity and fairness. Measurement also helps organizations steward limited resources by directing interventions where they are most needed or most likely to lead to improvement (at both local and institutional levels). Tracking educator well-being over time provides accountability and enables evaluation of well-being interventions to determine if they are making a difference.

Measuring and working to improve educator well-being can also help organizations recruit and retain a diverse workforce and promote equitable and inclusive work environments. Achieving these goals requires organizations to recognize that educators from groups that have historically been excluded from the health professions are underrepresented in academic medicine (particularly at higher academic ranks)² and face disproportionate challenges in the work environment, such as bias and discrimination, limited access to mentors and sponsors, expectations for administrative and diversity work (the "diversity tax") and inequitable system policies.³ Supporting others (or being asked to support others) while simultaneously processing one's own lived experience also takes a physical and emotional toll. Understanding and addressing these factors, as explored in detail in Chapter 4, is essential to building a diverse health care workforce and cocreating organizational systems and cultures where all medical educators can thrive.

Finally, measuring educator well-being can help organizations ensure that they are meeting accreditation requirements. The Accreditation Council for Graduate Medical Education (ACGME), for example, specifies common core requirements related to well-being that residency programs must meet, including faculty education on mental health, fatigue and burnout.⁴ Other accrediting bodies such as the Liaison Committee on Medical Education (LCME)⁵ and the Commission on Osteopathic College Accreditation (COCA)⁶ mention the need for students to have access to well-being education and ready access to mental health providers, but do not spell out specific requirements for educators.

If to Measure?

When considering if the time is right to measure educator well-being, organizational leaders should consider contextual factors. Chaotic situations, such as the early weeks of the COVID-19 pandemic, require leadership to take immediate action to address ever-changing situations. Acting in the face of high uncertainty and urgency requires leaders to look for "what works" with clear communication directed toward people's needs to feel heard, protected, prepared, supported and cared for. Formal measurement can be revisited once the situation has stabilized, even if it remains complicated or complex.

When to Measure?

Having decided to proceed, the timing of measurement requires careful consideration. Even brief surveys can be problematic if they coincide with other pressing "asks" of educator time and attention. Engaging institutional stakeholders associated with each participating educator group can help identify optimal time windows for measurement. Efforts to measure educator well-being should also avoid competing with other surveys requiring educator participation (e.g., ACGME faculty surveys, which are currently conducted between January and April of each academic year and require a minimum 70% completion rate to avoid being flagged as noncompliant with accreditation requirements). Frequency of measurement is important to consider as well, as frequent surveys without time to review and respond to survey results with meaningful improvement efforts can leave educators feeling that nothing is being done with their survey responses and disincentivize future participation.

Who to Measure?

The "who to measure" question has two facets. The first relates to who leads the measurement effort, and the second relates to who is invited to participate.

Who will lead the measurement effort?

Any number of institutional leaders could initiate and administer a survey or other strategy for assessing educator well-being, including those working in human resources or faculty affairs, department chairs, or deans. If an institution has an office for educator well-being or has a designated wellness leader such as a dean for well-being or a chief wellness officer, that office or leader can lead the measurement effort. A local leader is needed even if an outside vendor conducts the survey.

An institution can signal how highly leadership values educator well-being by whom they choose to lead the survey effort. Ideally, this choice would be a highly respected person with seniority in the institution who has access to the necessary resources. Finally, the designated leader should be someone who can motivate high

rates of survey participation and inspire trust that the data will be collected, analyzed and used in ways that honor and protect the identity of participants.

Who will be invited to participate?

Faculty can be defined in relatively narrow or more expansive ways. Academic health centers often classify educators as clinical or non-clinical, clinical or research, core or non-core, or full-time or part-time. Decisions about the target population for educator well-being assessment should be guided by the purpose of measurement, available resources and attention to equity and inclusion. For example, if the purpose of measurement relates to only a subset of educators (e.g., based on role, full- or part-time status, location, sociodemographic factors), then the target population can be limited accordingly. More targeted assessments can also be used to test and refine questions before pursuing a large-scale survey. If resources are constrained, limiting the study population can help ensure there are adequate resources to not only conduct the survey, but also communicate and act on the results.

While limiting the survey population may be justifiable and necessary in some circumstances, it can also pose challenges. First, including only certain groups of educators limits internal benchmarking and can create a perception of differential importance or preferential concern. Second, it risks excluding or obscuring the experiences of educators from groups historically excluded in medicine. Finally, when the measured sample is small, educators may be reluctant to respond (or respond honestly) out of concern that their responses could be identified.

Another question that arises is whether educators should be surveyed at the same time as staff clinicians, fellows, residents, students and administrative staff. This decision depends on whether the instrument being used to measure well-being has validity evidence supporting its use in all the populations mentioned. In some cases, these populations can be surveyed at the same cadence, while in other situations certain groups may need to be surveyed more or less often than others. If an organization has an existing all-employee survey, it may be most feasible to measure educator well-being as part of this survey, provided educator responses can be identified and analyzed separately. It is important to ensure that survey questions are relevant to the educator experience to avoid disenfranchising this group of respondents.

What to Measure?

Educator well-being can be assessed directly using instruments designed for this purpose. When used alone, however, results offer little guidance regarding how to respond or where to direct improvement efforts. Measuring potential drivers of educator well-being (or lack thereof) in the work environment as well can provide actionable insights.

Well-being

Well-being can be conceptualized and measured in different ways. To date, most measures of well-being among health professionals have conceptualized well-being in terms of positive or negative affective states with a disproportionate emphasis on negative states (e.g., emotional exhaustion, depersonalization, depression, anxiety) over positive ones.⁸ Burnout is a construct many organizations choose to measure, for example, because of its relationship to work experiences, prevalence and serious consequences.⁹ Positive states have

received greater attention recently, however, as health care organizations and professional societies have rallied around efforts to promote work engagement, professional fulfillment and joy in work.^{10,11}

Drivers of well-being

Existing studies offer guidance regarding categories of work-related factors relevant to well-being among health professionals that could be explored as part of a measurement effort. Examples include workload (e.g., work demands and support, work efficiency), leadership behaviors, autonomy and flexibility, social support, work-life integration and meaning in work.¹² Workplace accessibility, fairness and exposure to mistreatment or discrimination have also been linked to well-being among health professionals.¹³ Such factors contribute to the diversity, inclusion and equity climate within an organization, which can be measured using survey items or other assessment methods.^{14,15}

Recent research highlights the potential for identity-related tensions associated with evolving expectations of educators and new educator roles. Alignment between individual and organizational values is another important consideration, which may be particularly relevant to educators given tensions between practice, research and education missions within many organizations. To

How to Measure?

Efforts to assess educator well-being benefit from both local and system-level measures. Local measures occur frequently and can be initiated and tracked by particular teams, courses, programs, schools, clinics/units or departments/divisions that use results to inform ongoing improvement efforts. Examples recommended by the Institute for Healthcare Improvement include monthly or quarterly pulse surveys and visual measures such as a glass jar placed near an elevator where staff drop in one marble each day—a blue marble for a good day or a tan marble for a bad day. Touchpoints, check-ins, interviews, focus groups and listening sessions/tours can also be effective ways to monitor educator well-being. These sorts of approaches are often most feasible for local measurement.

Systems-level measures occur at less frequent intervals (e.g., annually) to identify improvement opportunities, identify groups that are struggling or thriving, and track progress over time. Most conceptions of well-being can (or must) be measured via self-report and hence lend themselves to survey items. Table 6.1 provides examples of select well-being measures, organized according to whether the measure assesses a specific construct (e.g., burnout, work engagement), combines multiple constructs into a single instrument (composite measures), assesses well-being across multiple dimensions or uses holistic measurements that avoid imposing a particular conception of well-being and instead allow educators to respond based on their personal interpretations, values and beliefs.

While this chapter focuses primarily on survey-based measures, other system-level indicators could potentially serve as useful (though indirect) indicators of educator well-being. Examples include engagement with well-being programming, use of institutional support resources, leave time utilization, absenteeism, turnover and retention, "soft attrition" (decreasing work effort) and exit interviews (among others).

Selecting an approach to well-being measurement involves numerous practical considerations, including the burden of measurement, applicability to educators and across educator groups, trustworthiness of results, degree to which results will be actionable and sensitivity to change. Attention to diversity, equity and inclusion is also critical throughout the measurement process. The following sections address each of these considerations.

Burden of measurement

Those leading an educator well-being assessment should consider the burden of measurement from the perspective of both educators and the organization.

For educators, time is a leading concern, and multiple, lengthy or frequent surveys can lead to poor response rates. Strategies to address this concern include carefully considering the timing of measurement (as previously discussed), leveraging existing data (if available), limiting the number of survey items and adding measures to existing surveys (e.g., an organizational all-staff survey). Educators consider return on investment and will be more motivated to respond when there is a commitment to act on survey results. Importantly, educators must be aware of this commitment and believe it is sincere. Creating a communication strategy to accompany the launch of the survey and articulating when and how results will be shared and used can increase survey participation. The potential for survey items to trigger distressing thoughts and emotions should also be considered, with careful attention to the wording of questions and inclusion of information about how to access support.

For the organization, the burden of measurement includes costs associated with the use of an instrument and analysis of results. Strategies to reduce these costs include using existing well-being data (e.g., ACGME faculty well-being surveys, other direct or indirect indicators of educator well-being as mentioned previously), incorporating well-being measures into existing data streams (e.g., school faculty surveys, institutional well-being surveys) and choosing inexpensive or cost-free instruments that are relatively simple to analyze. External organizations (survey vendors) are available to administer and analyze surveys if a school or health system does not have the resources to do so internally. Use of a vendor may also bolster confidence in the anonymity of the survey and signal to educators that the institution is willing to invest in well-being. The burden of measurement for organizations also includes costs associated with advertising and socializing the survey, analyzing quantitative and qualitative data if the survey is conducted "in house," disseminating survey results and responding to survey findings.

Applicability

Medical educators have diverse roles, backgrounds, credentials and activities. Some may engage learners directly, while others serve learners more indirectly. Educators also interact with learners in different settings (e.g., physical classrooms, virtual platforms, clinical environments) and can simultaneously be learners themselves (e.g., residents as teachers). Selection of survey items requires careful attention to these factors to ensure the content and wording is relevant to all intended participants. Failure to do so risks sending an implicit message that the experiences of some educator groups matter more than others.

The wording of common well-being measures can also affect their applicability to educators at academic health institutions. Some surveys use the term "client" or "recipient" instead of "patient," while others use the term

"students" instead of the broader term "learners." Table 6.1 provides references to commonly used surveys as well as links for more information on these surveys.

Reliability and validity of results

Using well-being measures with validity evidence supporting the use of resulting scores helps ensure the trustworthiness of results, which is critical to avoid missed or misunderstood problems and misdirected resources. Table 6.1 includes select examples of validity evidence for scores generated by various well-being measures. Interested readers are encouraged to review these resources for more information, along with other articles that provide helpful overviews of validity evidence supporting the use of scores derived from common well-being measures. ^{1,18,19}

Actionability

A common goal of measurement is to not only identify improvement opportunities, but also motivate improvement efforts. If measurement is undertaken in a context where there is already high commitment to educator well-being, this may be sufficient to motivate action based on survey results. Improvement efforts, however, often involve contending for limited resources, and those responsible for leading the survey effort must convince education and system leaders that such investments are justified.

Survey results are more likely to lead to action if they are easy to interpret, meaningful and important to leaders, and reflect input from key stakeholders. Strategies to increase the actionability of survey results include measuring potential work-related drivers of educator well-being, linking well-being data to other valued outcomes, conducting subgroup analyses, providing benchmarking to help contextualize results and offering respondents with opportunities to complete free-text comment boxes.

- Measuring potential work-related drivers of educator well-being can yield valuable information regarding how to direct and prioritize subsequent improvement efforts. This can be done by adding survey items or merging survey results with work environment data from other sources.
- Linking well-being data to other valued outcomes such as educator retention or learner experience can also help motivate investments in improvement efforts. While local data are most compelling, referencing prior research (e.g., showing costs and consequences of burnout)⁹ can sometimes serve a similar purpose.
- Conducting subgroup analyses of survey results can increase the actionability of data by revealing
 disparities (i.e., groups of educators who face disparate challenges in the work environment), identifying
 groups that may benefit from targeted interventions and uncovering best practices that can be shared
 with others or implemented in other contexts.
- **Providing benchmarking data** also helps leaders interpret and act upon survey results. Benchmarking data can be internal (comparing to other staff within the organization), external (comparing to faculty at other schools²⁰ or national surveys of health professionals²¹) or longitudinal (comparing to historical institutional data to assess the impact of interventions and direct ongoing improvement efforts). When benchmarking, it is important to consider the populations from which norms or benchmarks are derived, as they may not reflect the diversity of participating educators.

• Including free-text survey comments is another powerful way to motivate action based on survey results. While potentially time-consuming to analyze, comments add emotion, energy and humanity to otherwise numeric survey results and can include helpful suggestions for improvement.

Sensitivity to change

Ideally, well-being measures would be capable of accurately detecting meaningful effect sizes from interventions. As shown in Table 6.1, some well-being measures specify a time frame (e.g., past two weeks, past month) while others do not, which may reflect their ability to detect change over short time intervals. Because well-being strategies often take time to develop and implement, sensitivity to change over longer time frames (i.e., one year or more) is often adequate. Data supporting sensitivity to change (also known as responsiveness) is most robust for the Maslach Burnout Inventory, which appears capable of detecting moderate effect sizes from various interventions intended to improve physician burnout.

Attention to equity

Measuring educator well-being is often motivated by efforts to ensure the needs of educators receive equitable attention, consideration and support within a school, program or organization. Organizations should, therefore, avoid inadvertent exclusion of relevant educators or educator groups, select survey items that are applicable to all participating educators and ensure mechanisms are in place to make survey results accessible and meaningful to various stakeholders.

Attention to equity within and among educators is also critical throughout the measurement process. Equity among educators (with respect to well-being, ratings of the work climate or experiences in the work environment) can be explored via subgroup analyses based on sociodemographic factors such as age, gender, race/ethnicity, academic rank, faculty role, school/program, site, etc. Such efforts often require balancing the desire to collect demographic information with the need to protect the identity of respondents. Strategies to address this concern include using an external organization to administer the survey and analyze results, using broader demographic categories with a "choose not to disclose" option and redacting data unless a prespecified minimum threshold of respondents is achieved.

The selection and wording of demographics items requires careful consideration. The use of broad categories (such as "Asian," "disability," or "lesbian-gay-bisexual-transgender-queer+ identification") can help protect anonymity but can also obscure or conflate identities that are relevant to the educator experience. A potential way to mitigate this concern is to provide space for respondents to enter more granular information if desired. Demographic questions also typically address only certain aspects of identity. It may thus be helpful to include an open-ended question inviting educators to share other identities that are relevant to their work experience.

Importantly, while subgroup analyses based on demographic factors can help uncover inequities, this approach also risks promoting a deficit framework that blames individuals or groups for disparities rather than recognizing the role of systems and structures in advantaging certain groups and disadvantaging others. Subgroup analyses can also promote eugenic thinking and reinforce implicit or explicit assumptions that certain groups (e.g., white, male, cisgender, heterosexual, able-bodied, wealthy) are innately superior rather than highlighting systemic and structural factors that underlie and perpetuate group-level differences. Those leading the measurement effort

should weigh the potential for unintended consequences of this sort when considering whether to do subgroup analyses and when communicating results.

Finally, subgroup analyses based on a single aspect of identity (e.g., gender or race/ethnicity) neglect the complex interdependence of social identity structures within broader power hierarchies (i.e., intersectionality).²³ Combining responses to demographic items can illuminate disproportionate burdens affecting individuals with combinations of identities. This approach requires large sample sizes to avoid compromising anonymity, however, and increases the complexity of analysis. It also neglects the potential for privilege and oppression to expand and contract over time and across settings.²³ Qualitative approaches are often better suited to understanding how these dynamic complexities manifest within a particular organizational context. These issues are explored in greater depth in Chapter 4.

What To Do With the Results?

Measuring educator well-being is a significant undertaking and accomplishment and may directly promote well-being by giving educators a voice and demonstrating that the institution cares about their experiences and perspectives. Measurement is nevertheless only the first step. Once survey results are in, a communication plan and process can help ensure that key findings are summarized in ways that facilitate effective and efficient communication with various stakeholder groups (e.g., one-page of key findings, slide presentations with additional details, dashboards for internal benchmarking, video recording of a local well-being champion verbally summarizing survey results).

Potential pitfalls to avoid in this process include:

- Asking without acting: Asking about problems that the organization is unwilling or unable to address
 can foster resentment and ill will among educators. A genuine commitment to educator well-being
 involves responsiveness and a sense of responsibility to act on the results. This requires clarifying who
 will be responsible for implementing improvement efforts and accountable for resulting outcomes.
 Importantly, the team that collects and distributes survey data may differ from those responsible for
 improvements and accountable for outcomes.²⁴
- Action without communication: Improving work environments takes time, and educators can become frustrated by the lack of progress, especially if they are not involved in or informed of improvement efforts. Repeated communications via multiple modalities are often needed given the numerous issues and challenges vying for educator attention.
- Analysis paralysis: Because data often begets new questions, a common pitfall is to succumb to
 "analysis paralysis," channeling energy and resources into additional analyses or repeated
 measurements rather than improvement efforts.
- Predetermined solutions: Organizational leaders often have ideas for school- or system-wide
 improvements, but local leaders must own their survey results considering context-specific challenges
 and opportunities. Improvement efforts should also engage educators by using a Listen-Act-Develop
 model,²⁵ conducting follow-up surveys or collecting qualitative input, all while taking care to ensure
 educators are both empowered and supported in the process.
- Blame game: Existing data suggest systems factors are the primary driver of burnout and root cause of inequities. Blaming individuals or groups for systems problems can foster resentment and ill will among

- educators, perpetuate harmful stereotypes or assumptions or lead educators to pursue solutions that are personally beneficial but harmful to the organization and society. Blaming the system without willingness to improve the system is also problematic, as improvement efforts are more likely to be successful when organizations and educators are mutually accountable for system outcomes.
- Deficit thinking: Improvement efforts often focus on fixing problems (i.e., deficit thinking) by identifying risk factors for a problem (e.g., educator burnout, disparate outcomes among social groups) and developing prevention and intervention strategies directed toward these factors—often using prescribed resources, assumed solutions and negative labels. Strengths-based approaches, in contrast, focus on affirmation, appreciation and positive dialogue by locating problems outside of individuals and empowering groups to work together to discover what inspires them and bring about their goals. Appreciative inquiry exemplifies this approach.²⁶
- Disregarded distress: Including information about local employee assistance programs, mental health
 resources and other sources of support can ensure educator respondents know how to seek help if
 they are experiencing distress. Awareness of emotions is also important when sharing survey results.
 For example, unfavorable results could be particularly discouraging to faculty affairs leaders who have
 worked hard to advocate for and support educators or elicit defensive reactions from education or
 system leaders. Careful listening, empathic responses and acknowledgement of uncertainty are critical
 for navigating the communication of survey results.

Conclusion

Medical educators are a unique population with unique experiences. They play a pivotal role in training the next generation of scientists and physicians and in advancing human understanding in the sciences and medicine. Measurement can illuminate the unique and overlapping factors relevant to the well-being of educators and help organizations understand what is working well for educators and what pressures and challenges are keeping them from being their best professional selves. Measurement of well-being, if done well, can ideally inspire and inform improvement initiatives that strengthen an organization's ability to provide the highest-quality teaching, patient care and scientific innovation.



Case Study

Situation

The chair of the family medicine department at a medium sized academic health center oversees 70 faculty members. Her faculty members are mostly physician educators who have clinical duties but also, to a lesser degree, are basic scientists and psychologists. She has been concerned about recent reports of low morale among the faculty and has seen many faculty either leave or retire early in the last year. She has also talked to other faculty, many of whom are her contemporaries, who seem to be thriving. She was trained in an era where well-being was not a priority and is aware of a cultural shift in medicine where many of the newly trained health care professionals are prioritizing self-care and work-life integration. She is fully invested in improving the well-being of her colleagues as she has seen the devastating effects of burnout in her long career.

Approach

This person knows that to better understand the well-being of her faculty, she must conduct a survey. She has only a small budget to support this effort but does have a well-being champion with dedicated salary support in her department to head this survey. Her well-being champion is a junior faculty member who is looking to take on more quality improvement projects, so she has approved this survey as part of a quality improvement initiative. The department chair and her team selected a brief, composite survey that included questions relevant to all roles to measure educator well-being. This department chair decided to administer the survey during an all-faculty meeting where there was 75% participation. A total of 67% of faculty fully completed the survey. Results were collated by the well-being champion and her quality improvement team and shared with each division and their constituents via PowerPoint presentations. During these presentations, questions were often raised about how other divisions were performing and what should be done about the results.

Lessons learned

- The department chair chose a survey instrument that was free for use by nonprofit organizations for quality improvement purposes (and therefore low-cost) and short in length (and therefore easy to administer and complete). Other surveys may have been more expansive in well-being domains, however also, more time-consuming and/or more expensive. Non-survey options such as focus groups could have been an alternative but would have required skilled facilitators and more time.
- The department chair picked a time when faculty were already gathering to administer this short survey to maximize participation.
- A dedicated well-being champion with chair support was able to lead the survey effort and ensure that the survey results were quickly and accurately relayed back to the faculty.
- Divisions struggled to interpret and act on the survey results, however, due to lack of benchmarking and lack of information about potential causes of faculty distress.

Take-Home Points

- Surveying the well-being of educators is important to inform institutional policies and initiatives and to identify and address diversity and equity issues.
- Choosing a survey leader or leadership team that conveys the importance of the survey to educators, engenders trust and inspires participation is critical.
- Engaging stakeholders and other institutional leadership in the survey planning and implementation process and ensuring there is a plan for communicating, and acting upon, results contributes to success in measurement.
- There is no one-size-fits-all survey instrument or survey plan. Both the survey itself and its implementation need to be tailored to the institution. Asking why, when, who, what, how and what is next can help develop the most effective survey plan.

Solutions

Why survey?

- Understand the experience and well-being of an institution's educator workforce
- Uncover variation across groups or sites, and explore how these variations relate to barriers and facilitators in the work environment
- Assess the impact of well-being initiatives
- Recruit and retain a diverse body of educators
- Meet accreditation requirements
- Identify areas of success and areas of opportunity
- Help organizations direct and prioritize their improvement efforts

If?

Assess overall climate/context

When?

- Avoid competition with other surveys, major initiatives or pressing asks
- Allow time between surveys for institutions to review and respond to survey results with meaningful improvement efforts

Who?

- Choose a survey leader who is highly respected person with seniority in the institution, who has
 access to the necessary resources and who can inspire trust and motivate high rates of survey
 participation
- Consider educator type (clinical or non-clinical), schedule (part-time or full-time), location, role, and rank

What?

- Consider conceptions of well-being
- Consider drivers of well-being

How?

- Take into account:
 - o Burden of measurement (for educators and the organization)
 - o Applicability (to target populations of educators)

- o Reliability and validity of results
- o Actionability of results
- o Sensitivity to change
- o Attention to equity

What happens next?

- Analyze results (or use outside consultants)
- Share with institutional leadership
- Share at the level at which initiative can be developed and implemented (department, unit, program, specific location)
- Determine who is accountable for improvement efforts and tracking outcomes
- Resurvey to assess impact of initiatives—may use limited survey with limited population at this time, depending on initiative being evaluated

Pitfalls to avoid

- Asking without acting
- Action without communication
- Analysis paralysis
- Predetermined solutions
- Blame game
- Deficit thinking
- Disregarding distress

Table 6.1: Well-Being Measures (Select Examples)

-specific s to nts'
s to
s to

	Academy of Medicine,				
	Washington, DC.				
Maslach	Maslach C, Jackson SE,	22 items	Exhaustion	Rating	Work-specific
Burnout	Leiter MP. <i>Maslach</i>	22 1161113	(emotional)	scale	Refers to
Inventory:	Burnout Inventory		Depersonalization	ranges	'students'
Educators	Manual. 4th Edition.		Personal	from	Cost
Survey (MBI-	1996-2018.		accomplishment	'never'	COSt
ES)	https://www.mindgarden.		accomplishment	to 'every	
	com/maslach-burnout-			day'	
	inventory-mbi/685-mbi-			day	
	manual.html. Accessed				
	December 9, 2022.				
Maslach	West CP, Dyrbye LN,	1-2 items	Exhaustion	Rating	Work-specific
Burnout	Sloan JA, Shanafelt TD.	1 2 101113	(emotional)	scale	Cost
Inventory	Single item measures of		Depersonalization	ranges	
(MBI) single	emotional exhaustion			from	
item	and depersonalization			'never'	
measures	are useful for assessing			to 'every	
	burnout in medical			day'	
	professionals. <i>J Gen</i>				
	Intern Med.				
	2009;24(12):1318.				
	West CP, Dyrbye LN,				
	Satele DV, Sloan JA,				
	Shanafelt TD. Concurrent				
	validity of single-item				
	measures of emotional				
	exhaustion and				
	depersonalization in				
	burnout assessment. J				
	Gen Intern Med.				
	2012;27(11):1445-1452.				
	Waddimba AC, Scribani				
	M, Nieves MA, Krupa N,				
	May JJ, Jenkins P.				
	Validation of single-item				
	screening measures for				
	provider burnout in a				
	rural health care				

	network. Eval Health				
Oldenburg Burnout Inventory (OBI)	Prof. 2016;39(2):215-225. Halbesleben JRB, Demerouti E. The construct validity of an alternative measure of burnout: investigating the English translation of the Oldenburg burnout inventory. Work Stress. 2005;19(3):208–220. Demerouti E, Mostert K,	16 items	Exhaustion (physical, emotional, mental) Disengagement	None specified	Work-specific
	Bakker AB. Burnout and work engagement: a thorough investigation of the independency of both constructs. <i>J Occup Health Psychol</i> . 2010;15(3):209.				
	Demerouti E, Bakker AB. The Oldenburg Burnout Inventory: A good alternative to measure burnout and engagement. 2007. https://www.researchgat e.net/publication/467041 52 The Oldenburg Burn				
	out Inventory A good al ternative to measure bu rnout and engagement. Accessed December 9, 2022.				
Copenhagen Burnout Inventory (CBI)	Kristensen TS, Borritz M, Villadsen E, Christensen KB. The Copenhagen Burnout Inventory: A new tool for the assessment of burnout.	19 items	Exhaustion (physical and psychological) Personal Work-related Client-related	None specified	Work-specific and non-work specific Refers to 'clients'

	Work Stress. 2005;19(3):192-207.				
Mini Z 1.0, Mini Z 2.0	Mini Z Survey Institute for Professional Worklife. https://www.professional worklife.com/mini-z- survey. Accessed December 9, 2022.	1 item	Burnout as defined by survey respondent	None specified	Work-specific
	Rohland BM, Kruse GR, Rohrer JE. Validation of a single-item measure of burnout against the Maslach Burnout Inventory among physicians. <i>Stress Health</i> . 2004;20(2):75–79.				
	Dolan ED, Mohr D, Lempa M, et al. Using a single item to measure burnout in primary care staff: A psychometric evaluation. <i>J Gen Intern</i> <i>Med</i> . 2015;30(5):582-587.				
	Waddimba AC, Scribani M, Nieves MA, Krupa N, May JJ, Jenkins P. Validation of single-item screening measures for provider burnout in a rural health care network. <i>Eval Health</i>				
Work engagen	<i>Prof.</i> 2016;39(2):215-225.				
Utrecht Work Engagement Scale (UWES) - short version	Schaufeli WB, Bakker AB, Salanova M. The measurement of work engagement with a short questionnaire: a cross-	9 items	Vigor Dedication Absorption	Rating scale ranges from 'never'	Work-specific

	national study.			to 'every	
	•			,	
	Educational and			day'	
	Psychological				
	Measurement.				
	2006;66(4):701–716.				
	Rosenthal J. Critical				
	Synthesis Package:				
	Utrecht Work				
	Engagement Scale				
	(UWES). MedEdPORTAL.				
	Published online August				
	1, 2014:9862.				
	Kulikowski K. Do we all				
	agree on how to				
	measure work				
	engagement? Factorial				
	validity of Utrecht Work				
	Engagement Scale as a				
	standard measurement				
	tool - a literature review.				
	Int J Occup Med Environ				
	Health. 2017;30(2):161-				
	175.				
Utrecht Work	Schaufeli WB, Shimazu A,	3 items	Same as above	Rating	Work-specific
Engagement	Hakanen J, Salanova M,			scale	
Scale (UWES)	De Witte H. An ultra-			ranges	
– ultrashort	short measure for work			from	
version	engagement: the UWES-			'never'	
	3 validation across five			to 'every	
	countries. <i>Eur J Psychol</i>			day'	
	-			uay	
	Assess. 2019;35(4):577-				
	591.				
	Merino-Soto C, Lozano-				
	Huamán M, Lima-				
	Mendoza S, Calderón de				
	la Cruz G, Juárez-García				
	A, Toledano-Toledano F.				
	Ultrashort version of the				
	Utrecht Work				
	OB COME TO THE				

Composite me Well-Being Index (WBI)	Dyrbye LN, Satele D, Sloan J, Shanafelt TD. Utility of a brief screening tool to identify physicians in distress. J Gen Intern Med.	7 items	Fatigue Depression Burnout Anxiety/stress Mental/physical quality of life	Past 1 month	Work-specific and non-work specific Cost
Expanded Well-Being Index (eWBI)	2013;28(3):421-427. Dyrbye LN, Satele D, Shanafelt T. Ability of a 9-Item well-being index to identify distress and stratify quality of life in US workers. <i>J Occup Environ Med</i> . 2016;58(8):810-817.	9 items	Same as above + Meaning in work Satisfaction with work-life integration	Past 1 month	Same as above
Professional Fulfillment Index (PFI)	Trockel M, Bohman B, Lesure E, et al. A brief instrument to assess both burnout and professional fulfillment in physicians: reliability and validity, including correlation with self- reported medical errors, in a sample of resident and practicing physicians. Acad Psychiatry. 2018;42(1):11- 24. Vetter M, Vetter M, Cohn D, Yeager K, Fowler J. Utility of the Stanford Professional Fulfillment Index as a single survey	16 items	Professional fulfillment Work exhaustion Interpersonal disengagement	Past 2 weeks	Work-specific Refers to 'patients' and 'colleagues'

instrument to assess				
· •				
S328.				
Su R, Tay L, Diener E. The	10 items	Life satisfaction	None	Not work-
development and		Positive emotions	specified	specific
validation of the		Support		
Comprehensive		Belonging		
Inventory of Thriving		Meaning		
(CIT) and the Brief		Engagement		
Inventory of Thriving		Self-worth		
(BIT). Appl Psychol Health		Self-efficacy		
Well Being. 2014;6(3):251-		Accomplishment		
279.		Optimism		
Grossi E, Groth N,	6 items	Anxiety	Past 1	Not work-
Mosconi P, Cerutti R,		Vitality	month	specific
Pace F, Compare A,		Depressed mood		
Apolone G. Development		Self-control		
and validation of the		Positive well-being		
short version of the				
Psychological General				
Well-Being Index				
(PGWB-S). Health Qual				
Life Outcomes.				
2006;4:88.				
al measures				
VanderWeele TJ. On the	12 items	Happiness and life	None	Not work-
promotion of human		satisfaction	specified	specific
flourishing. Proc Natl		Physical and mental		
Acad Sci U S A.		health		
2017;114(31):8148-8156.		Meaning and purpose		
		Character and virtue		
VanderWeele TJ,		Close social		
McNeely E, Koh HK.		relationships		
Reimagining health—		Financial stability		
flourishing. JAMA.				
2019;321(17):1667.				
Węziak-Białowolska D,				
McNeely E, VanderWeele				
TJ. Flourish index and				
	holistic physician wellbeing. <i>Gynecol Oncol</i> . 2021;162:S327-S328. Su R, Tay L, Diener E. The development and validation of the Comprehensive Inventory of Thriving (CIT) and the Brief Inventory of Thriving (BIT). <i>Appl Psychol Health Well Being</i> . 2014;6(3):251-279. Grossi E, Groth N, Mosconi P, Cerutti R, Pace F, Compare A, Apolone G. Development and validation of the short version of the Psychological General Well-Being Index (PGWB-S). <i>Health Qual Life Outcomes</i> . 2006;4:88. al measures VanderWeele TJ. On the promotion of human flourishing. <i>Proc Natl Acad Sci U S A</i> . 2017;114(31):8148-8156. VanderWeele TJ, McNeely E, Koh HK. Reimagining health—flourishing. <i>JAMA</i> . 2019;321(17):1667. Węziak-Białowolska D, McNeely E, VanderWeele	holistic physician wellbeing. Gynecol Oncol. 2021;162:S327- S328. Su R, Tay L, Diener E. The development and validation of the Comprehensive Inventory of Thriving (CIT) and the Brief Inventory of Thriving (BIT). Appl Psychol Health Well Being. 2014;6(3):251- 279. Grossi E, Groth N, Mosconi P, Cerutti R, Pace F, Compare A, Apolone G. Development and validation of the short version of the Psychological General Well-Being Index (PGWB-S). Health Qual Life Outcomes. 2006;4:88. al measures VanderWeele TJ. On the promotion of human flourishing. Proc Natl Acad Sci U S A. 2017;114(31):8148-8156. VanderWeele TJ, McNeely E, Koh HK. Reimagining health— flourishing. JAMA. 2019;321(17):1667. Węziak-Białowolska D, McNeely E, VanderWeele	holistic physician wellbeing. Gynecol Oncol. 2021;162:S327- S328. Su R, Tay L, Diener E. The development and validation of the Comprehensive Inventory of Thriving (CIT) and the Brief Inventory of Thriving (BIT). Appl Psychol Health Well Being. 2014;6(3):251- 279. Grossi E, Groth N, Mosconi P, Cerutti R, Pace F, Compare A, Apolone G. Development and validation of the short version of the Psychological General Well-Being Index (PGWB-S). Health Qual Life Outcomes. 2006;4:88. al measures VanderWeele TJ. On the promotion of human flourishing. Proc Natl Acad Sci U S A. 2017;114(31):8148-8156. VanderWeele TJ, McNeely E, Koh HK. Reimagining health— flourishing. JAMA. 2019;321(17):1667. Węziak-Białowolska D, McNeely E, VanderWeele Welziak-Białowolska D, McNeely E, VanderWeele Weziak-Białowolska D, McNeely E, VanderWeele	holistic physician wellbeing. Gynecol Oncol. 2021;162:S327- S328. Su R, Tay L, Diener E. The development and validation of the Comprehensive Inventory of Thriving (CIT) and the Brief Inventory of Thriving (BIT). Appl Psychol Health Well Being. 2014;6(3):251- 279. Grossi E, Groth N, Mosconi P, Cerutti R, Pace F, Compare A, Apolone G. Development and validation of the short version of the Psychological General Well-Being Index (PGWB-S). Health Qual Life Outcomes. 2006;4:88. al measures VanderWeele TJ, On the promotion of human flourishing. Proc Natl Acad Sci U S A. 2017;114(31):8148-8156. VanderWeele TJ, McNeely E, Koh HK. Reimagining health— flourishing. JAMA. 2019;321(17):1667. Węziak-Białowolska D, McNeely E, VanderWeele I0 items Life satisfaction Positive emotions Support Self-sention Support Self-worth Self-worth Self-worth Self-worth Self-worth Self-worth Self-control Positive well-being None specified None specified None specified Physical and mental health Meaning and purpose Character and virtue Close social relationships Financial stability Financial stability

	I a		T	T	<u> </u>
	secure flourish index –				
	validation in workplace				
	settings. Cogent				
	Psychology. 2019;6				
	(1598926):1-10.				
Holistic measu	res				
Single item	Wanous JP, Reichers AE,	1 item	Job satisfaction	None	Work-specific
measure of	Hudy MJ. Overall job			specified	
job	satisfaction: how good			(typically	
satisfaction ^a	are single-item)	
	measures? J Appl				
	Psychol. 1997;82(2):247-				
	252.				
	Dolbier CL, Webster JA,				
	McCalister KT, Mallon				
	MW, Steinhardt MA.				
	Reliability and validity of				
	a single-item measure of				
	job satisfaction. Am J				
	Health Promot.				
	2005;19(3):194-198.				
Single item	Cheung F, Lucas RE.	1 item	Life satisfaction	None	Not work
measure of	Assessing the validity of			specified	specific
life	single-item life			(typically	'
satisfaction	satisfaction measures:)	
	results from three large				
	samples. <i>Qual Life Res</i> .				
	2014;23(10):2809-2818.				
Linear	Norman GR, Sloan JA,	1+ items	Quality of life overall		Not work
analog scale	Wyrwich KW.		or across multiple		specific ^a
assessment	Interpretation of changes		dimensions (e.g.,		0-10 or 0-100
quality of life	in health-related quality		physical, mental,		linear analog
measures ^a	of life: the remarkable		emotional, social,		scale
	universality of half a		financial, spiritual,)		
	standard deviation. <i>Med</i>				
	Care. 2003;41(5):582-92.				
	Singh JA, Satele D,				
	Pattabasavaiah S,				
	Buckner JC, Sloan JA.				
	Normative data and				

1: 11 : 10 : 10 : 1		
clinically significant effect		
sizes for single-item		
numerical linear		
analogue self-		
assessment (LASA)		
scales. Health Qual Life		
Outcomes. 2014;12:187.		
Dyrbye LN, Satele D,		
Shanafelt T. Ability of a		
9-Item Well-Being Index		
to identify distress and		
stratify quality of life in us		
workers. J Occup Environ		
Med. 2016;58(8):810-817.		

^a These approaches could be adapted to assess satisfaction or quality of (work) life across multiple work-related roles or activities by repeating single-item, holistic ratings across each role/activity.

References

- 1. Dyrbye LN, Meyers D, Ripp J, Dalal N, Bird SB, Sen S. A pragmatic approach for organizations to measure health care professional well-being. *NAM Perspect*. Published online October 1, 2018. doi:10.31478/201810b.
- 2. 2021 U.S. medical school faculty. AAMC. https://www.aamc.org/data-reports/faculty-institutions/interactive-data/2021-us-medical-school-faculty. Accessed December 9, 2022.
- 3. Nguemeni Tiako MJ, South EC, Ray V. Medical schools as racialized organizations: A primer. *Ann Intern Med.* 2021;174(8):1143-1144. doi:10.7326/M21-0369.
- 4. Common program requirements. https://www.acgme.org/what-we-do/accreditation/common-program-requirements/. Accessed December 9, 2022.
- 5. Standards, publications, & notification Forms. LCME. https://lcme.org/publications/. Accessed December 9, 2022.
- 6. Accreditation Standards. American Osteopathic Association. https://osteopathic.org/accreditation/standards/. Accessed December 9, 2022.
- 7. Shanafelt T, Ripp J, Trockel M. Understanding and addressing sources of anxiety among health care professionals during the COVID-19 pandemic. *JAMA*. 2020;323(21):2133. doi:10.1001/jama.2020.5893
- 8. Brady KJS, Trockel MT, Khan CT, et al. What do we mean by physician wellness? A systematic review of its definition and measurement. *Acad Psychiatry*. 2018;42(1):94-108. doi:10.1007/s40596-017-0781-6.
- 9. West CP, Dyrbye LN, Shanafelt TD. Physician burnout: contributors, consequences and solutions. *J Intern Med.* 2018;283(6):516-529. doi:10.1111/joim.12752.
- 10. Lall M, Gaeta T, Chung A, et al. Assessment of physician well-being, part two: Beyond burnout. *West J Emerg Med.* 2019;20(2):291-304. doi:10.5811/westjem.2019.1.39666.

^b For all instruments, it is important to determine if permissions/approvals are needed to reproduce survey items.

- 11. Perlo J, Balik B, Swensen S, Kabcenell A, Landsman J, Feeley D. IHI Framework for improving joy in work. IHI White Paper. Published 2017. https://www.ihi.org/resources/Pages/IHIWhitePapers/Framework-Improving-Joy-in-Work.aspx. Accessed December 8, 2022.
- 12. Committee on Systems Approaches to Improve Patient Care by Supporting Clinician Well-Being, National Academy of Medicine, National Academies of Sciences, Engineering, and Medicine. *Taking Action Against Clinician Burnout: A Systems Approach to Professional Well-Being*. National Academies Press; 2019:25521. doi:10.17226/25521.
- 13. Dyrbye LN, West CP, Sinsky CA, et al. Physicians' experiences with mistreatment and discrimination by patients, families, and visitors and association with burnout. *JAMA Netw Open.* 2022;5(5):e2213080. doi:10.1001/jamanetworkopen.2022.13080.
- 14. Person SD, Jordan CG, Allison JJ, et al. Measuring diversity and inclusion in academic medicine: The diversity engagement survey. *Acad Med J Assoc Am Med Coll*. 2015;90(12):1675-1683. doi:10.1097/ACM.0000000000000921.
- 15. Achieving inclusion excellence in academic medicine: Foundational principles of inclusion excellence (FPIE) toolkit. https://store.aamc.org/foundational-principles-of-inclusion-excellence-fpie-toolkit.html. Accessed December 9, 2022.
- 16. Simpson D, Marcdante K, Souza KH, Anderson A, Holmboe E. Job Roles of the 2025 medical educator. *J Grad Med Educ.* 2018;10(3):243-246. doi:10.4300/JGME-D-18-00253.1
- 17. Gonzalo JD, Dekhtyar M, Caverzagie KJ, et al. The triple helix of clinical, research, and education missions in academic health centers: A qualitative study of diverse stakeholder perspectives. *Learn Health Syst*. 2021;5(4):e10250. doi:10.1002/lrh2.10250.
- 18. Linton MJ, Dieppe P, Medina-Lara A. Review of 99 self-report measures for assessing well-being in adults: exploring dimensions of well-being and developments over time. *BMJ Open*. 2016;6(7):e010641. doi:10.1136/bmjopen-2015-010641.
- 19. Valid and reliable survey instruments to measure burnout, well-being, and other work-related dimensions. National Academy of Medicine. https://nam.edu/valid-reliable-survey-instruments-measure-burnout-well-work-related-dimensions/. Accessed December 9, 2022.
- 20. Burnout among US medical school faculty. AAMC. https://www.aamc.org/data-reports/analysis-brief/report/burnout-among-us-medical-school-faculty. Accessed December 9, 2022.
- 21. Shanafelt TD, West CP, Sinsky C, et al. Changes in burnout and satisfaction with work-life integration in physicians and the general US working population between 2011 and 2020. *Mayo Clin Proc.* 2022;97(3):491-506. doi:10.1016/j.mayocp.2021.11.021.
- 22. Panagioti M, Panagopoulou E, Bower P, et al. Controlled interventions to reduce burnout in physicians: A systematic review and meta-analysis. *JAMA Intern Med.* 2017;177(2):195. doi:10.1001/jamainternmed.2016.7674.
- 23. Wyatt TR, Johnson M, Zaidi Z. Intersectionality: a means for centering power and oppression in research. *Adv Health Sci Educ Theory Pract.* 2022;27(3):863-875. doi:10.1007/s10459-022-10110-0.
- 24. Shanafelt T, Trockel M, Ripp J, Murphy ML, Sandborg C, Bohman B. Building a program on well-being: Key design considerations to meet the unique needs of each organization. *Acad Med.* 2019;94(2):156-161. doi:10.1097/ACM.0000000000002415.

- 25. Swensen S, Kabcenell A, Shanafelt T. Physician-organization collaboration reduces physician burnout and promotes engagement: The Mayo Clinic experience: *J Healthc Manag*. 2016;61(2):105-127. doi:10.1097/00115514-201603000-00008.
- 26. Armstrong AJ, Holmes CM, Henning D. A changing world, again. How appreciative inquiry can guide our growth. *Soc Sci Humanit Open*. 2020;2(1):100038. doi:10.1016/j.ssaho.2020.100038.

Appendix

At the end of 2021/beginning of 2022 as the COVID-19 pandemic entered its third year the AMA Accelerating Change in Medical Education Consortium and the AMA Council on Medical Education collaborated to make recommendations that would support educators and help medical education institutions continue their education mission in the face of sustained disruption. This preliminary guide was published on the AMA website in February 2022 and forms the basis for this book. It is reprinted here.



PROTECTING THE EDUCATION MISSION DURING SUSTAINED DISRUPTION

Organizational steps to support medical educators

Longstanding threats to the education mission in medicine continue to undermine educators, organizations and systems. The global COVID-19 pandemic has highlighted these threats and amplified pre-existing strains on the medical educator workforce. The disruptions caused by the pandemic will continue to impact educational systems even as pandemic conditions improve. In addition, the attrition of medical educators has far-reaching implications for health care systems nationwide.

Since educational efforts have traditionally been undervalued by health care organizations, there is risk that individuals under stress will abandon educational roles.

Such response is doubly unfortunate since well-being is highly associated with a sense of purpose in one's work and many derive tremendous satisfaction from educational activities. Concurrently, the disruption of the pandemic propelled important educational innovations—ones that we should consider sustaining or developing further.

It is imperative to preserve the capacity for creativity among educators and avoid reversion to historical practices out of sheer exhaustion and change fatigue. This is not an issue of individual resilience. Health care organizations and educational institutions must take action to avoid mass abandonment of educational duties and loss of educational leaders.

The following list of challenges and associated recommendations emerged from the vision and collaborative work among members of the AMA Accelerating Change in Medical Education Consortium and the AMA Council on Medical Education:

Key threats to the education mission during sustained disruption

Disruption of the clinical learning environment

- Operating in strained clinical systems impairs educators' ability to engage in teaching, direct observation and provision of feedback to learners.
- Physicians who are over-extended and under-supported cannot serve as inspiring role models to students, resident physicians and fellows.
- Resident physicians have experienced clinical service demands that altered their own education and training and impact their pathway to specialty certification and independent practice. Such disruption makes it difficult for residents to fulfill their critical role in educating and serving as role models for medical students.



Repeated redesign of the educational program

- Educators have been called upon to urgently redesign educational programs while simultaneously dealing with constant changes in workflows.
- Conversion to virtual and hybrid settings has required educators to learn new skills and evaluate tools while addressing competing organizational and personal demands.
- Administrative burdens related to educational programming have not been adjusted to offset competing demands, and in some cases administrative support has been reduced due to staffing limitations.

Cognitive and emotional load on educators

- Educators have faced greater cognitive load in all daily routines (disruptions to clinical, classroom and home activities), making it difficult to continue to juggle multiple roles.
- Educators bear responsibility for learner well-being and have experienced an added empathic burden as learner well-being has been compromised.
- Learner anxiety and frustration with alterations to training and career progression have compromised relationships between learners and educators; yet such relationships are important motivation for engaging in educational duties.
- Increased awareness of chronic health inequities highlighted the imperative
 to combat structural racism in educational programs. Educators need time
 and resources for personal reflection, developing individual and organizational
 competency, and redesigning educational systems. This challenge places an
 even greater burden on educators who themselves identify with historically
 marginalized groups.

Organizational gaps

- Historically, key activities to support medical education are not sufficiently valued or protected by organizations.
- The medical education system lacks educational surge capacity.
- Medical educators function at the intersections of systems. Each educational program interacts with associated health system(s), accreditors and licensure systems that are also in flux. This places constraints on the educator's influence and creates tension.
- Career development among educators is languishing as they strive to meet the many challenges presented by recent disruptions.
- Well-being efforts appropriately focus on learners and on health care providers; however specific programming to support the unique needs of educators—and the diversity of needs among educators—is commonly neglected.

Identifying organizational strategies

In developing organizational responses to the many challenges, it is important to recognize that education is not limited to delivery of content. Critical educator roles also include direct observation and assessment of learner progress, coaching and mentoring. The broad continuum of educator roles ranges from informal to more structured duties—from "frontline" clinical supervisors in the workplace (including resident physicians); to educators who contribute with intermittent small group facilitation, lectures or mentoring; to those with formally appointed roles in teaching, assessment and advising. Finally, administrators charged with oversight of curricular and assessment programs require support from the leadership of both the educational institution and their associated health system(s) to implement necessary changes.

Given this breadth of educator roles, the potential actions to support them are equally broad.

Recommendations:

- 1. Monitor and support well-being of educators as well as learners
 - Explicitly assess and monitor educator well-being.
 - Establish organizational well-being efforts that specifically consider the context of educators. Read AMA STEPS Forward: Creating a Resilient Organization.

2. Refine workflows to support educational interactions

- Design clinical workflows and spaces that support inclusion of the education mission.
- Identify systems issues that create constraints on individual educators' efforts.
- Empower educators, administrators and learners to propose adjustments to workflows.
- Solicit input from resident physicians—who are simultaneously learners, educators and role models.
- Foster an environment of shared learning, facilitate positive communications and promote co-production of solutions between educators and learners.

3. Protect time for educational activities

- Provide appropriate protected time and remuneration for educational roles.
- Alleviate administrative burdens on educators and eliminate non-essential tasks.
- Provide adequate educational administrative staffing.
- Leverage shared online or commercial resources as appropriate rather than generating new institutionally based materials or solutions.

4. Provide ongoing training

- Provide frontline clinical supervisors tips for efficiency in workplace-based education practices:
 - Reduce focus on delivery of content and encourage more direct observation of encounters with patients, discussion of clinical reasoning and provision of feedback.
 - Encourage educational huddles to set priorities with learners.

- Provide faculty development in efficient workplace-based tools, such as one minute preceptor. View the <u>AMEE Guide no. 34: Teaching in the clinical</u> environment.
- Reduce educator burden related to evaluations of learner performance by soliciting focused feedback in a timely manner and monitoring total requests on the individual educator.
- Provide training for educators in tools for remote education, simulation and for engaging learners in remote care settings. Read the <u>AMA Telehealth Clinical</u> <u>Education Playbook</u>.
- Support educator involvement in professional development programs.

5. Nurture educator career advancement

- Remunerate for educational roles (e.g., protected time, merit pay, inclusion in annual evaluations)
- Provide reports of educators' activities to department chairs or other relevant supervisors
- Provide regular reviews of each educator's progress regarding promotion and tenure and provide appropriate mentoring.
- Create mechanisms to acknowledge the work of educational design (and redesign) in promotions and tenure evaluations as scholarly teaching and/or institutional service.
- Provide centralized support for scholarly work in education, such as amplifying opportunities to submit for presentations and grants, hosting writing huddles, and supporting travel to educational conferences.
- Amplify a view of the education mission as a critical factor in organizational success and showcase educational contributions across the organizational community.

6. Build educational surge capacity

- Inventory individual educator's tasks to create flexibility and consider re-distribution as indicated.
- Engage more educators across the program.
- Create a deliberate succession plan for key educational roles.
- Consider opportunities for near-peer teaching and assessment and for multi-media approaches that support self-directed learning and assessment.
- Offset the burden of developing educational materials and tools by using open source or vendor solutions as appropriate.
- Actively leverage technologic solutions to administrative and educational tasks.

7. Advocate for systems change, locally and nationally

- Encourage all stakeholders to apply <u>systems thinking</u> to the educational process and engage learners in co-production of educational redesign.
- Strengthen relationships between the educational institution and associated health system(s) to facilitate shared problem solving and align communications.
- Implement value-added roles for learners.
- Advocate for appropriate adjustments in accreditation processes and assist leaders of educational programs to follow evolving recommendations and requirements.

